

C. M. Jewell & Associates Pty Ltd Water and Environmental Management A.C.N. 056 283 295

SITE AUDIT STATEMENT

Schedule 1, Form 2 (Contaminated Land Management Regulation 1998)

Site Auditor (accredited under Contaminated Land Management Act 1997):

Name:	Christopher M. Jewell	Phone:	02) 4759 3251
Company:	C. M. Jewell & Associates Pty Ltd	Fax:	02) 4759 3257
Address:	1/13 Kalinda Road,	Accred. No:	9810
	Bullaburra NSW 2784		

Site Audit Statement No.: SA183

Site Details:

Address:100-120 King Street
RANDWICK NSWPostcode:2031Lot and DP Number:Lot 202 in DP879576Local Government Area:Randwick City Council

Site audit requested by:

Name:	Mr David Freeman		
Company:	Sir Moses Montefiore Jewish Home		
Address:	120 High Street		
	HUNTERS HILL NSW	Postcode:	2110
Phone:	02 9879 2715	Fax:	02 9871 2700
Name of cont	act person (if different from above):		

Consultancy who conducted the site investigations and/or remediation:

URS Australia Pty Ltd

Titles of reports reviewed:

Supplementary Sampling and Analysis Plan – Lot 202, King Street, Randwick. Letter/Report dated 12 June 2002, prepared by URS Australia Pty Ltd.

Data Assessment Report - Lot 202, King St, Randwick. Report No. 51072-001-R001G dated 20 September 2002, prepared by URS Australia Pty Ltd.

Other information reviewed:

Regional topographical and geological mapping.



Site Audit Statement No.: SA183

ł

ł

Summary Site Audit Report

Title: Site Audit – 110-120 King Street, Randwick (Sir Moses Montefiore Jewish Home)

Date: 30 September 2002

I have completed a site audit (as defined in the Contaminated Land Management Act 1997) and reviewed the reports and information referred to above with due regard to relevant laws and guidelines. I certify that the site (tick **all** appropriate boxes):

(a) is suitable for the following use(s):

- residential, including substantial-vegetable garden and poultry;
- ----- residential, including substantial vegetable-garden excluding poultry;
- residential with accessible soil, including garden (minimal home-grown produce contributing less than-10% fruit and vegetable intake) excluding poultry;
- residential with minimal opportunity for soil-access, including units;
- □ secondary school;
- □ _____park, recreational open space, playing field;
- □ commercial/industrial-use;
- □-----other (please specify):

(b) is not suitable for any beneficial use due to risk of harm from contamination:

☑ (comments):

This audit was carried out for the purpose of determining:

- (1) the nature and extend of any contamination of the land,
- (2) the nature and extent of the investigation or remediation,
- (3) what investigation or remediation remains necessary before the land is suitable for residential use with minimal opportunity for soil access.

It has been determined that parts of the site were potentially contaminated by:

- heavy metals,
- total petroleum hydrocarbons (TPH),
- benzene, toluene, ethyl benzene and xylenes (BTEX),
- polycyclic aromatic hydrocarbons (PAH), and
- asbestos-containing materials (ACM)

as a result of its previous use as a bus depot and subsequent demolition work.

Document Qa-O21 Rev 12/01/2000:CMJ



C. M. Jewell & Associates Pty Ltd Water and Environmental Management A.C.N. 056 283 295

Site Audit Statement No.: SA183

Remediation and validation of the site has previously been carried out, but it has been determined that this work was not performed and documented in a manner that meets current standards.

Investigations carried out by URS in 2002 were adequate and appropriate. These investigations have established that some residual contamination (or recontamination) by petroleum hydrocarbons is present in part of the site and that both soil and groundwater are impacted, to the extent that remediation is required before the site will be suitable for residential use.

It is considered that the remedial strategy proposed by URS is appropriate for the site. It is therefore recommended that:

- (1) A formal remedial action plan (RAP) be prepared.
- (2) The RAP should include procedures for:
 - Excavation of hydrocarbon-contaminated soil
 - On-site treatment by land farming
 - Validation sampling and analysis
 - Groundwater monitoring
 - Contingency groundwater remediation and validation.
- 3) The RAP be reviewed and approved by an Auditor. That approval should be given in letter form.
- 4) A validation report be prepared.
- 5) A further site audit be carried out and a new site audit statement issued.

Subject to successful implementation of these recommendations, it is considered that the site can be rendered suitable for residential use with minimal opportunity for soil access.

This Site Audit Statement should be used in conjunction with the attached Summary Site Audit Report which contains important supporting information.

I am accredited by the NSW Environment Protection Authority under the *Contaminated Land* Management Act 1997 as a Site Auditor

Accreditation Number: 9810

zie.



Site Audit Statement No.: SA183

I certify that:

- (a) I have personally examined and am familiar with the information contained in this statement, including the reports and information referred to in this statement, and
- (b) this statement is, to the best of my knowledge, true, accurate and complete, and
- (c) on the basis of my inquiries made to those individuals immediately responsible for making the reports, and obtaining the information referred to in this statement, those reports and that information are, to the best of my knowledge, true, accurate and complete.

I am aware that there are penalties for wilfully submitting false, inaccurate or incomplete information.

Signed:

Date: <u>30 September 2002</u>



Report No. J0807.3

SITE AUDIT 110-120 KING STREET, RANDWICK (SIR MOSES MONTEFIORE JEWISH HOME)

for

SIR MOSES MONTEFIORE JEWISH HOME

September 2002

Controlled Copy No. 4



C.M. Jewell & Associates Pty Ltd A.C.N. 056 283 295 A.B.N. 54 056 283 295

Water and Environmental Management

My Ref: J0807.3 30 September 2002 CMJ:mp

1/13 Kalinda Road, Bullaburra, NSW 2784, Australia P.O. Box 10, Wentworth Falls, NSW 2782 Phone: (02) 4759 3251 Fax: (02) 4759 3257 Email: postle@cm-jewell.com.au

Sir Moses Montefiore Jewish Home 120 High Street HUNTERS HILL NSW 2110

Attention: Mr David Freeman

Dear Mr Freeman,

RE: Site Audit - 100-120 King Street, Randwick (Sir Moses Montefiore Jewish Home)

As requested, I have carried out a Statutory Site Audit of the above land; my Site Audit Report is attached to this letter.

Further remediation and validation of the site are required, as set out in the attached report and in the conditions on the Site Audit Statement.

As required by the Contaminated Land Management Act 1997, I have sent copies of the Site Audit Statement to Randwick City Council and the EPA. As a courtesy, copies of the report have also been provided to Randwick City Council and URS Australia Limited.

If you have any questions please do not hesitate to contact me on (02) 4759 3251.

For and on behalf of <u>C. M. JEWELL & ASSOCIATES PTY LTD</u>

CHRIS JEWELL

Distribution Sir Moses Montefiore Jewish Home (Controlled Copy No. 1) URS Australia Limited (Controlled Copy No. 2) Randwick City Council (Controlled Copy No. 3) CMJA Library (Controlled Copy No. 4) File (Original)

TABLE OF CONTENTS

		Ň
1.0	INTRODUCTION	
	1.1 Background	I
	1.2 Involvement of Auditor	
	1.3 Scope and Structure of Review Report	
	1.4 Limitations and Intellectual Property Matters	
2.0	SITE INFORMATION	
	2.1 Site Identification and Location	
	2.2 Site Setting	
	2.3 Topography and Surface Hydrology	
	2.4 Geology	4
	2.5 Hydrogeology	4
	2.6 Site History	
	2.7 Contaminants of Concern	
	2.8 Proposed Development	
	2.9 Assessment Criteria	б
	2.9.1 Adopted Criteria	
	2.9.2 Auditor's Comments	8
3.0	SUMMARY OF PREVIOUS INVESTIGATIONS	
2.0	3.1 Auditor's Comments	
4.0		
4.0	ADDITIONAL SITE ASSESSMENT AND PROPOSED REMEDIAL WO	
	4.1 Initial Supplementary Works, July 2002	
	4.1.1 Soil Sampling and Analysis	
	4.1.2 Stockpile Assessment	
	4.1.3 Asbestos Survey	
	4.2 Soil and Groundwater Sampling, August 2002	
	4.2.1 Soil Sampling	
	4.2.2 Groundwater Sampling	
	4.3 Soil Sampling, September 2002	
	4.4 Discussion of Results	
	4.5 Remedial Action Strategy	
	4.5.1 Remediation Objectives	
	4.5.2 Remediation Area	
	4.5.3 Proposed Soil Remediation Program	
	4.5.4 Proposed Groundwater Remediation Program	
	4.6 URS's Conclusions	
	4.6.1 Auditor's Comments	
5.0	COMPLETENESS AND ADEQUACY OF INVESTIGATION	
	5.1 Initial Sampling Strategy	
	5.2 Sampling Procedures	
	5.3 Quality Assurance/Quality Control	
	5.4 Groundwater Issues	
	5.5 Aesthetic Issues	
	5.6 Chemical Mixtures.	
	5.7 Reporting Standards	
6.0	CONCLUSIONS	10
0.0		

J0807.3 30 September 2002	Page iii
REFERENCES	
Important Information about your Environmental Site Assessment	X ⁱ

TABLES

Table 1	Assessment Criteria - Soils
Table 2	Soil Sampling and Analysis Schedule – July 2002
Table 3	Analysis Schedule – August 2002
Table 4	Summary of Statistics of Assessment Data

FIGURES

Figure 1	Site Location and Setting
Figure 2	Site Layout and Sampling Locations
Figure 3	Asbestos Assessment Zones

APPENDICES

;]

l

Appendix A Contaminant Groups

Appendix B Communications with the Auditor

Appendix C Information Relied Upon by the Auditor

Appendix D QA/QC Information

ASSOCIATED REPORTS

Supplementary Sampling and Analysis Plan – Lot 202, King Street, Randwick. Letter/Report dated 12 June 2002, prepared by URS Australia Pty Ltd.

Data Assessment Report - Lot 202, King St, Randwick. Report No. 51072-001-R001G dated 20 September 2002, prepared by URS Australia Pty Ltd.

LIST OF ABBREVIATIONS

<u>Measures</u>	
μg/L	micrograms per litre
km	kilometre
L	litre
m	metre
m^2	square metre
mg/kg	milligrams per kilogram
mm	millimetre
<u>General</u>	
AHD	Australian Height Datum
ANZECC	Australian and New Zealand Environment and Conservation Council
AST	above-ground storage tank
CLM Act	Contaminated Land Management Act
CMJA	C. M. Jewell & Associates Pty Ltd
DA	development application
DLWC	Department of Land and Water Conservation
DP	deposited plan
DQO	data quality objectives
EPA	Environment Protection Authority
ESA	Environmental Site Assessment
HDPE	high-density polyethylene
NATA	
	National Association of Testing Authorities
NEPM	National Environment Protection Measure
PID	photoionisation detector
PQL	practical quantitation limit
PSH	phase-separated hydrocarbons
QA	quality assurance
QC	quality control
RAP	remediation action plan
RL	relative level
RPD	relative percentage difference
TCLP	Toxicity Characteristic Leaching Procedure
UCL	upper confidence limit
UST	underground storage tank
<u>Analytes - O</u>	
BaP	benzo(a)pyrene
BTEX	benzene, toluene, ethyl benzene, xylene
OCP	organochlorine pesticides
OPP	organophosphorus pesticides
PAH	polycyclic aromatic hydrocarbons
PCB	polychlorinated biphenyls
SVOC	semivolatile organic compounds
TPH	total petroleum hydrocarbons
VHC	volatile halogenated compounds
VIIC	
	volatile organic compounds
<u>Analytes - In</u>	<u>organic</u>
As	arsenic
Cd	cadmium
Cr	chromium
Cu	copper
Fe	iron
Hg	mercury
Mn	-
	manganese
Ni	nickel
Pb	lead
Zn	zinc

1.0 INTRODUCTION

1.1 Background

This Site Audit Report relates to the land located at 100-120 King Street, Randwick, in New South Wales.

Specifically, the Site Audit relates to the validation work completed by URS Australia Limited (URS) on behalf of Sir Moses Montefiore Jewish Home (and as described in URS's report dated 20 September 2002).

The Site Audit which this report describes was requested by Mr David Freeman of Sir Moses Montefiore Jewish Home on 27 May 2002. The Audit was required by Randwick City Council to assist in fulfilling Council's obligations under Clause 7 of the State Environmental Planning Policy No. 55 (SEPP55) prior to considering a development application, submitted to Council on 3 June 2002 in respect of the site. The Site Audit is thus a Statutory Site Audit under the provisions of Part 4 of the Contaminated Land Management Act 1997.

The audit was conducted for the purpose of determining:

- (i) the nature and extent of any contamination of the land,
- (ii) the nature and extent of the investigation or remediation,
- (iii) what investigation or remediation remains necessary before the land is suitable for any specified use or range of uses.

The Site Audit Report has been prepared in accordance with the guidelines issued by the NSW Environment Protection Authority (EPA) (*Guidelines for the NSW Site Auditor Scheme*, 1998). It has been prepared by Christopher Jewell, who is a Site Auditor accredited under the NSW Contaminated Land Management Act 1997.

1.2 Involvement of Auditor

The Auditor became involved in this project at the start of URS's current involvement, and has had appropriate input into the scope and planning of that phase of assessment. He had no involvement in previous work.

The Auditor has visited the site on two occasions. A compliance checklist has been completed and is held on file.

1.3 Scope and Structure of Review Report

Section 2 of this report sets out basic identification and location information concerning the site, and briefly describes the site's topography, geology and hydrogeological setting. An indication of the site's history and an outline of the proposed future use and the associated assessment criteria are also provided.

Section 3 sets out a summary of those data from previous environmental assessments that have been assessed as useable in the current assessment.

Section 4 provides a summary of the supplementary investigations undertaken by URS. The completed proposed remedial works are also discussed within this section.

Section 5 of this report presents an audit of the completeness and adequacy of the environmental assessments which have been completed. The audit was carried out against the criteria established by the NSW EPA publication, Guidelines for the NSW Site Auditor Scheme (1998), but also incorporates the reviewer's own judgement; reference has been made to other guideline publications issued or endorsed by the NSW EPA, including Guidelines for Consultants Reporting on Contaminated Sites (1997), Guidelines for Assessing Service Station Sites (1994), Sampling Design Guidelines (1995) and the National Environmental Protection (Assessment of Site Contamination) Measure (1999), as appropriate.

Section 6 of this report sets out the Auditor's conclusions with regard to the current condition of the land at the site and its suitability for the intended use.

Throughout this report, extensive use has been made of the site assessment report prepared by URS; sections of those reports have been adopted for use in this report.

1.4 **Limitations and Intellectual Property Matters**

This report has been prepared by C. M. Jewell & Associates Pty Ltd for the use of the client and local government agency identified in Section 1.1, for the specific purpose described in that section.

The work has been carried out, and this report prepared, utilising the standards of skill and care normally expected of a site auditor practising in New South Wales under the requirements of the Contaminated Land Management Act 1997. The level of confidence of the conclusions reached is governed, as in all such work, by the scope of the investigation carried out and by the availability and quality of the data. The Auditor has satisfied himself that the available data are adequate to support the conclusions he has reached, and comply with the minimum requirements indicated in the guideline documents specified for the NSW Site Auditor Scheme. Where limitations or uncertainties in conclusions are known, they are identified in this report. However, no liability can be accepted for failure to identify conditions or issues which arise in the future and which could not reasonably have been assessed or predicted using the site information and analytical data available for review.

Data collected by others have, of necessity, been used to support the conclusions of this report. Those data have been subjected to reasonable scrutiny but have essentially, and necessarily, been used in good faith. Liability cannot be accepted for errors in data collected by others where such errors could not have been detected by reasonable scrutiny of the data and supporting information supplied to or requested by the Auditor.

This report, any original data contained in the report, and its findings and conclusions remain the intellectual property of C. M. Jewell & Associates Pty Ltd. A licence to use the report for the specific purpose identified in Section 1.1 is granted to the persons identified in that section on the condition of receipt of full payment for the services involved in the preparation of the report.

It is recommended that this report should not be used by other persons or for other purposes than those identified in Section 1.1 without prior reference to the Auditor. The report should not be reproduced except in full and with the permission of C. M. Jewell & Associates Pty Ltd.

Page 2

2.0 SITE INFORMATION

2.1 Site Identification and Location

The site is located at 100-120 King Street, Randwick, New South Wales. The site location is shown on Figure 1. Australian Map Grid Zone 56H co-ordinates of the centre of the site are approximately 33700E 6246650N. At the date of this report, the site was owned by the Honorary Board of Management of the Sir Moses Montefiore Jewish Home.

The site lies within Randwick City Council local government area and is currently zoned 2B and 2C. The site is identified as Lot 202 in DP879576, in the Parish of Alexandria, County of Cumberland.

The site is rectangular in shape, with dimensions of approximately 165 metres north-south by 178 metres east-west, and has an area of approximately 29,350 m^2 .

An indication of the site's layout is provided on Figure 2.

2.2 Site Setting

The site is located in a predominantly residential area, bounded as outlined below.

To the north

The site is bounded to the north by Govett Lane. Residential properties are located on the northern side of Govett Lane

To the east

The site is bounded to the east by Dangar Street. Residential properties are located to the east of Dangar Street.

To the south

King Street bounds the site to the south. The Moriah Daycare facility adjoins the south-west portion of the site. Residential properties are located across King Street.

To the west

Lot 201, which together with Lot 202 was the former location of the Randwick Bus Depot, adjoins the western boundary of the site. A NSW State Transit Authority bus depot is located to the west of Lot 201.

2.3 Topography and Surface Hydrology

URS describes the topography as follows.

The ground gently slopes in a general westerly direction across the site, except at a point in the eastern third of the site where the ground drops more sharply at a constructed cutting. This cutting was constructed historically to improve the access for the bus depot operations. In places this cutting has been reinforced by the placement of a concrete retaining wall. This effectively divides the site into a higher eastern section and a lower western section.

...a number of stockpiles exist on site, some of which are up to 4 m high. The estimated volume of the stockpiled material is approximately 9,000 m³. The largest of these stockpiles extends from half way along the King Street boundary towards the centre of the site and is comprised primarily of building rubble and sandy soil, with some areas of sandstone rubble.

In the northwestern section of the site there is a large excavated area, which was undertaken as part of the previous investigations and partial remediation of the site. The excavation is approximately 100 m in length running east to west and up to 40 m in width. The elevation of the base of the excavation is between 37 and 38 mAHD. Three brick buildings remain on the site and are located in the northeast corner of the site.

Concrete slabs cover some areas of the site. These are predominantly in the area to the north and west of the excavation, and in the eastern section of the site.

Given that most the site has been cleared of pavement and the surface soils are typically sandy, it is expected that rainfall will infiltrate the surface soils directly or via runoff from the pavement. Prolonged heavy rainfall may cause saturation of the surface soils and percolation of rainwater into the groundwater system.

2.4 Geology

Reference to the 1:100,000-scale Sydney Geological Sheet (9130, Edition 1, 1983) indicates that the site is situated on the Holocene sediments of the Botany Basin. The sediments are variable in thickness, ranging up to 35 metres, although thickness is commonly of the order of 15 metres in the north of the basin. The sediments predominantly consist of well-sorted quartz sands interbedded with minor clay, peat and ironstone lenses. Most of the formation overlies the eroded bedrock surface of the Triassic Hawkesbury Sandstone, although moderately to highly weathered horizons of Ashfield Shale have been observed to underlie the Botany Sands in the north and west of the basin.

Reference to the Sydney 1:100,000 Soils Landscape Series Sheet (9130) indicates that the site is underlain by aeolian soil of the Newport soil landscape. The typical characteristics are shallow (less than 0.5 metre) well sorted siliceous sands overlying moderately deep (less than 1.5 metres) buried soils including yellow podzolic soils, with sandy topsoils on crests and gentle slopes. Deep podzols are found on steep slopes, lower slopes, and in depressions. Limitations of this soil landscape group include very high soil erosion hazard, localised steep slopes, non-cohesive topsoil, and very low soil fertility.

Intrusive investigations indicated that fill from 0.1 metre up to several metres thick overlies most the site. URS described the fill as being of a similar type to the underlying natural material.

2.5 Hydrogeology

The Quaternary alluvial, estuarine and, in places, aeolian sediments are known locally as the Botany Sands. The Botany Sands host an unconfined aquifer that has in the past been used extensively for water supply purposes. The aquifer is still used for industrial and irrigation purposes.

Groundwater movement within the Botany Sands occurs via primary porosity (i.e. intergranular flow). The hydraulic conductivity of the cleaner sands ranges up to 30 metres per day (m/d), with 10-15 m/d more typical for the clayey and peaty sands in the area. Yields obtained from the Botany Sands are generally moderate to high, usually of the order of 10 litres per second (L/s), although yields of up to 35 L/s have been obtained from the aquifer for industrial purposes.

The groundwater in the Botany Basin is considered to be of good and generally potable quality. It has a low salinity, typically less than 200 milligrams per litre (mg/L), although higher concentrations of up to 4000 mg/L occur. The pH of the water is generally low, usually of the order of 5.0 to 6.0 pH units, allowing for the enhanced solubility of most metals, including iron, in the aquifer.

Intrusive investigations at the site indicated that groundwater was at depths of less than 3 metres. It was assessed that the hydraulic gradient at the site was towards the west although in the Botany Aquifer the hydraulic gradient is generally towards the south-west.

The nearest receptors of the groundwater flowing from the site are likely to be irrigation wells located at Randwick Racecourse. At its closest point Randwick Racecourse is located approximately 330 metres south-west of the site

2.6 Site History

URS provided the following site history.

The former Randwick Bus Depot has been the subject of several site contamination investigations conducted between 1991 and 1998 by both Sinclair Knight and Dames & Moore (D&M).

A detailed site historical study was conducted by Doring (1990)....The historical data provided in the Doring study indicates that the eastern portion of the original depot, including both Lots 201 and 202, was used for tram and bus services mechanical (workshops, depot) throughout its life. The study indicated that the area initially commenced operations as a tramway workshop and depot in 1881.

Critical uses of the site not known to have occurred on either Lot 201 or Lot 202, which have occurred in other areas of the depot (the western portion of original depot property where the present day STA bus depot exists), as reported by Doring, included:

- A foundry;
- Munitions and arms manufacturing during WWII; and
- Electrical substations.

2.7 Contaminants of Concern

On the basis of the site history and previous environmental investigations undertaken at the site, potential contaminant groups of concern are identified as:

- heavy metals;
- monocyclic aromatic hydrocarbons, particularly benzene, toluene, ethyl benzene, and xylenes (BTEX);
- total petroleum hydrocarbons (TPH); and
- polycyclic aromatic hydrocarbons (PAH).

The individual compounds that make up these contaminant groups are listed in Appendix A.

Given the age of the buildings that had been located on the site, it was considered that asbestos is also a contaminant of concern.

2.8 Proposed Development

It is proposed that the site be redeveloped for an aged care facility.

2.9 Assessment Criteria

2.9.1 Adopted Criteria

The investigation criteria proposed by URS are the lower of those guideline levels set out in Columns 2 and 5 of the table: 'Soil Investigation Levels for Urban Redevelopment Sites in NSW', in the NSW EPA's *Guidelines for the NSW Site Auditor Scheme* (1998).

Criteria derived from Column 2 are health-based soil investigation levels for residential settings where there is minimal opportunity for soil access, originally developed by Imray and Langley in 1996, and currently reissued as Imray and Langley (1999): *Health-Based Soil Investigation Levels*, National Environment Protection (Assessment of Site Contamination) Measure (the NEPM), Schedule B, Guideline 7A. These soil investigation levels are also listed in Column D of Table 5-A, Schedule B(1) of the NEPM.

Criteria derived from Column 5 are provisional phytotoxicity-based investigation levels, and are identical to the Interim Urban Ecological Investigation Levels listed in Table 5-A, Schedule B(1) of the NEPM.

Criteria for TPH and BTEX are those published in the NSW EPA's *Guidelines for Assessing* Service Station Sites (1994) and listed in its Table 3 – 'Threshold Concentrations for Sensitive Land Use – Soils'.

These criteria are listed in Table 1.

٩.

TABLE 1 Assessment Criteria – Soils (mg/kg)					
Analyte	<i>EPA Column 2*</i> (NEPM Column D) Residential with Minimal Access to Soil	EPA Column 5* (NEPM Interim Urban Ecological) Phytotoxicity- based Criteria ^{††}	Sensitive Land Use – Soils [†]		
Arsenic	400	20	-		
Cadmium	80	3			
Chromium (III)	48%	400	-		
Copper	4000	100	-		
Lead	1200	600	-		
Mercury (inorganic)	60	15	-		
Nickel	2400	60	-		
Zinc	28,000	200	-		
Aldrin + Dieldrin	40	-			
Chlordane	200	-	-		
DDT + DDD + DDE	800	-	·		
Heptachlor	40	-	-		
РАН	80	-	_		
Benzo(a)pyrene	4	-	-		
PCB	40	-			
TPH C ₆ -C ₉	-	-	65		
TPH C ₁₀ -C ₄₀	-	-	1000		
TPH C ₁₆ -C ₃₅ aromatics ^{‡‡}	360	-	-		
TPH C ₁₆ -C ₃₅ aliphatics	22,400	-	-		
TPH >C ₃₅ aliphatics	224,000		-		
Benzene			1		
Toluene	-	-	1.4 [‡] /130 ^{\$}		
Ethyl benzene	-	-	3.1 /50		
Total xylenes	-		14**/25		

Notes: * NSW EPA (1998)

¹¹ Interim EILs for the urban setting are based on considerations of phytotoxicity, ANZECC B levels, and soil survey data from urban residential properties in four Australian capital cities.

¹¹ The carbon number is an 'equivalent carbon number' based on a method that standardises according to boiling point. It is a method used by some analytical laboratories to report carbon numbers for chemicals evaluated on a boiling point GC column.

[†] and notes below: NSW EPA (1994)

[†] The toluene threshold concentration is the Netherlands MPC to protect terrestrial organisms in soil. This value is obtained by applying a US EPA assessment factor to terrestrial chronic No Observed Effect Concentration (NOEC) data. The MPC is an 'indicative' value (Van de Plassche et al. 1993; Van de Plassche & Bockting 1993).

⁵ Human health and ecologically based protection level for toluene. The threshold concentration presented here is the Netherlands intervention value for the protection of terrestrial organisms. Other considerations such as odours and the protection of groundwater may require a lower remediation criterion.

The ethyl benzene threshold concentration is the Netherlands MPC for the protection of terrestrial organisms in soil. No terrestrial ecotoxicological data could be found for use in the Netherlands criteria derivation. Therefore, equilibrium partitioning has been applied to the MPC for water to obtain estimates of the MPC for soil. The MPC for water has been derived from aquatic ecotoxicological data (Van de Plassche et al. 1993; Van de Plassche & Bockting 1993).

¶ Human health based protection level for ethyl benzene or total xylenes as shown. The threshold concentration presented here is the Netherlands intervention value. Other considerations such as odours and the protection of groundwater may require a lower remediation criterion.

The xylene threshold concentration is the Netherlands MPC for the protection of terrestrial organisms in soil. No terrestrial ecotoxicological data could be found for use in the Netherlands criteria derivation. Therefore, equilibrium partitioning has been applied to the MPC for water to obtain an estimate of the MPC for soil. The MPC for water has been derived from aquatic ecotoxicological data. The concentration shown applies to total xylenes and is based on the arithmetic average of the individual xylene MPCs (Van de Plassche et al. 1993; Van de Plassche & Bockting 1993).

2.9.2 Auditor's Comments

The suggested criteria are considered appropriate for the site and the proposed use.

3.0 SUMMARY OF PREVIOUS INVESTIGATIONS

The site forms part of a property which operated as a tram and bus depot from 1881 until the 1980s. The western portion of this property houses the current Randwick NSW State Transit Authority Bus Depot. Excerpts from URS's Data Assessment Report for the site are provided below, and include a brief summary of previous investigations undertaken on the site as part of investigations of the property. As indicated below, some data from previous investigations have been used to complement data recently obtained by URS to complete an environmental assessment of the site.

In 1997, STA required subdivision of the greater property area into Lots 201 and 202 in preparation for future divestment. The subdivision required the demolition of buildings which straddled the boundary of Lot 201 and Lot 202. As the demolition works appeared to have been conducted in an uncontrolled manner Lot 202 required Post Demolition Validation in 1998 to assess if the demolition works had not adversely impacted the environmental condition of the site. The 1998 assessment was completed prior to the enactments of various sections of the *Contaminated Land Management Act, 1997*. In particular the works did not result in a Site Audit Statement (SAS) being prepared by an Independent Site Auditor which is a key requirement in the DA approvals process for Randwick City Council.

The objective of the assessment is to combine the results of recent investigations (July, August and September 2002) and historic investigations (1995 and 1998) in order to characterise the site condition for review by the site Auditor and preparation of a SAS.

The former Randwick Bus Depot has been the subject of several site contamination investigations conducted between 1991 and 1998 by both Sinclair Knight (SKM) and Dames & Moore (D&M) as outlined below:

- Sinclair Knight Partners (SKP, 1991) Preliminary Geotechnical and Environmental Investigation of the State Transit Authority's Randwick Bus Depot;
- Dames & Moore (D&M, 1991) Feasibility/Design Study for a Site Remediation Program at the Randwick Bus Depot;
- D&M (1992) EPA Compliance Report, Site Remediation Program: Phase I, Randwick Bus Depot;
- D&M (1994) EPA Compliance Report, Initial Site Validation Program, Randwick Bus Depot;
- D&M (1995) EPA Compliance Report, Final Site Validation Program, Randwick Bus Depot; and
- D&M (1998) EPA Compliance Report Post Demolition Site Validation Program Lot 202 King and Dangar Streets Randwick.

perspective.

Appropriate data, relevant to the current site conditions, was extracted from each of these reports in order to undertake this assessment....Other data has been rejected as not being suitable for site validation reporting.

uncontrolled demolition rendered some of the data unusable from a materials tracking

Acceptable data from the sampling at the site (1991-1998) have been reviewed as part of the assessment process in April 2002. URS assessed the data for its suitability for inclusion in the final site validation of Lot 202 with respect to the current Auditor Guidelines. Several of the data were found to be invalid for reasons such as lack of laboratory certificates and laboratory quality assurance/quality control (QA/QC). This resulted in all data prior to and including 1994 as being identified as invalid for validation and issue of a SAS.

The summary table below outlines the analytes and numbers of acceptable historical data (1991 - 1998) considered suitable by URS for use in the final site validation.

Analytes	Number of Retained Historic Samples
Heavy Metals	30
TPH/BTEX	54
TPH only	7
PAHs	10

Summary tables for data collected during the 1995 and 1998 works and considered suitable for validation purposes are provided in Appendix C.

URS's sampling strategy is outlined below.

The NSW EPA Sampling Design Guidelines (1994) state that for a site of this size (2.93 hectares), approximately 40 sample locations is optimum for analysing and defining the site for a specific landuse and reducing the possibility of the occurrence of contamination 'hotspots'. Based on the selected contaminants of concern and the locations of accepted historic samples, URS recommended that supplementary site assessment be undertaken for asbestos, metals, TPH/BTEX and PAHs. Although over 54 historical samples for TPH/BTEX could be retained, the samples are concentrated in localised areas associated with excavations. The intention of the supplementary works was to enable a spread of the sample locations across the site...

URS devised a sampling and analysis plan (URS 12 June 2002) for supplementary works at Lot 202 based on the data gaps listed above and submitted the plan to the Site Auditor for approval commensurate with current NSW EPA Auditor Guidelines.

The number of sampling locations for the supplementary works at Lot 202 has been devised using an approximate 40 m grid pattern with a total of an additional 25 sample locations as outlined in the Sampling and Analysis Plan (12 June 2002). Samples collected at each location were analysed for different sets of contaminants of concern. For example, only 15 of the locations required samples for PAHs and 10 locations for heavy metals. The background to this is that some of the historical data is considered valid in particular areas (for particular analytes) and the supplementary program fulfilled the existing data gaps for each contaminant of concern. The

locations were chosen on the basis of where sampling and analysis for particular analytes had not been completed historically. The extent of supplementary sampling was minimised to a level required to meet the Site Auditor's Guidelines and submitted to the Site Auditor for 'comment prior to commencing the fieldwork.

3.1 Auditor's Comments

The Auditor reviewed the Supplementary Sampling and Analytical Plan developed by URS and suggested some minor modifications to URS in a facsimile dated 26 June 2002. A copy of this facsimile is provided in Appendix B.

4.0 ADDITIONAL SITE ASSESSMENT AND PROPOSED REMEDIAL WORKS

4.1 Initial Supplementary Works, July 2002

4.1.1 Soil Sampling and Analysis

The initial supplementary works carried out by URS included:

- the excavation of twenty-eight test pits (identified as TP1 to TP28) across the site on a grid pattern with an approximate spacing of 40 metres;
- analysis for PAHs for samples collected from seventeen sample locations;
- analysis for heavy metals for samples collected from fourteen sample locations;
- analysis for TPH and BTEX for samples collected from fourteen sample locations; and
- analysis for asbestos for samples collected from twelve sample locations.

The sampling and analysis schedule for initial supplementary sampling is summarised in Table 2, whilst the sampling locations are shown on Figure 2.

TABLE 2							
Soil Sampling and Analysis Schedule – July 2002							
Sample	Heavy Metals	ТРН	BTEX	РАН	Asbestos		
TP01/0.3-0.5		100-1408-81W-4-			✓		
TP01/1.3-1.5		\checkmark	 ✓ 				
TP02/0.2-0.4	✓			\checkmark			
TP03/0.3-0.5		4	\checkmark		✓		
TP04/0.3-0.5	\checkmark			\checkmark			
TP04/1.3-1.5	✓			\checkmark			
TP5/0.3-0.5					✓		
TP5/1.3-1.5	✓			✓			
TP6/1.3-1.5		✓	 ✓ 	\checkmark			
TP07/0.3-0.5							
TP8/0.3-0.5	\checkmark			✓			
TP8/1.3-1.5	\checkmark	\checkmark	✓				
TP09/0.3-0.5							
TP9/1.3-1.5		✓	 ✓ 				
TP10/0.3-0.5				1			
TP11/0.3-0.5					✓		
TP12/0.3-0.5	 ✓ 			✓			
TP13/0.2-0.4					1		
TP14/0.3-0.5	✓	\checkmark	 ✓ 				
TP14/1.3-1.5	\checkmark	\checkmark	 ✓ 	✓			
TP14/2.4-2.6	1	✓	\checkmark				
TP15/0.3-0.5	\checkmark	\checkmark	\checkmark	✓			
TP15/1.3-1.5	✓	1	 ✓ 	✓			
TP16/0.2-0.4	\checkmark			✓			
TP17/0.3-0.5					 ✓ 		
TP18/0.3-0.5	~	~	✓	✓			
TP19/0.3-0,5					 ✓ 		
TP20/0.3-0.5	✓			1	 ✓ 		
TP21/0.3-0.5		\checkmark	\checkmark	~	\checkmark		
TP22/0.3-0.5	\checkmark	\checkmark	\checkmark		\checkmark		
TP22/1.3-1.5	✓	\checkmark	✓		✓		
TP23/0.3-0.5	1			··· ··· · · · · · ·····	√		
TP23/1.3-1.5	✓	\checkmark	✓	✓			
TP24/0.3-0.5		\checkmark	✓				
TP25/0.5-0.7	✓	✓	✓	~			
TP25/1.3-1.5	✓	\checkmark	\checkmark	✓	-		
TP26/0.3-0.5	✓	\checkmark	✓	~~~~			
TP26/1.3-1.5	✓	~	 ✓ 	✓			
TP26/2.9-3.0		✓	1				
TP27/0.3-0.5					✓		
TP28/0.3-0.5							

The results of laboratory analysis indicated that the concentrations of contaminants of concern were within the assessment criteria except for the following:

• lead (1880 mg/kg) in sample TP02/0.2-0.4; and

1

• TPH C₁₀-C₃₆ (1716 mg/kg) in sample TP25/1.3-1.5.

t

s,

٩,

Furthermore, asbestos was not detected in any of the samples.

4.1.2 Stockpile Assessment

URS's summary of the stockpile assessment follows.

In addition to the assessment of insitu materials, investigation of the stockpiles on the site was also required to determine their suitability to remain on site for use as fill. Survey results have indicated that the total volume of the stockpiles is approximately 9000 m³. To achieve the stockpile assessment URS undertook the following scope of work:

- Site inspection to establish those stockpiles requiring sampling. Stockpiles which were comprised of concrete and or brick rubble are not considered to require assessment. Stockpiles comprising soils were identified for assessment.
- Division of those stockpiles requiring assessment by an excavator into zones of approximately 300 m³ for individual assessment.

Ten stockpiles (identified as SP1 to SP7, and SP9 to SP11) were identified as comprising predominantly sandy soil. SP2 was identified as a concrete rubble stockpile but was also sampled. Samples were collected from these stockpiles as described below and analysed for heavy metals, TPH, BTEX, and PAH.

Samples were collected from stockpiles containing predominantly soil. One composite sample (3 point) from each approximate 300 m³ was collected. Stockpiles SP2, SP8, SP12, SP13, SP14, SP15, SP16, SP18 and SP19 consist of concrete slabs or brick (SP15 only) will be recycled and either retained on site or taken off site for recycling.

The results of laboratory analysis indicated that the concentrations of contaminants of concern were within the assessment criteria.

The stockpile samples were also submitted to a laboratory to determine whether any asbestos fibres were present in the samples. Asbestos (chrysotile) was only found in the sample from stockpile SP1. URS stated:

The stockpile was resampled and 4 representative samples were collected across the 300 m^3 . One sample was collected from each quadrant of the 300 m^3 . Each of these samples returned a non-detect result for asbestos. No asbestos cement fragments were identified in SP1.

4.1.3 Asbestos Survey

URS described the asbestos survey as follows:

The asbestos assessment followed the approach outlined in the ACLCA Code of Practice (draft February 2002). Sampling was undertaken by test pits such that observations of any suspect potential AC materials can be made. The sampling program was extended by a walk-over inspection of the whole site by an asbestos specialist. An outcome of the walk-over inspection was the inclusion of a further 2 test pits (TP27 and TP28) in a location where some demolition material had been spread, adjacent to the eastern boundary of the site.

An asbestos survey was conducted by Hibbs & Associates Pty Ltd (Hibbs) on 19 and 20 July 2002. To assist with the survey Hibbs divided the site into assessment zones. The survey identified the following issues:

- Fragments of bonded asbestos cement sheeting were noted at various locations around the site including within the assessment zones No. 1, 4, 5
- Some demolition rubble mixed with soil was spread across the eastern area of the site (grid location A3 A5) adjacent to the site boundary. Cement bonded asbestos fragment were observed in this material.

Figure 3 shows the asbestos assessment zones, together with the areas where positive identification was made of asbestos-containing materials. A copy of the summarised results of the asbestos survey is included in Appendix C.

4.2 Soil and Groundwater Sampling, August 2002 4.2.1 Soil Sampling

Twenty-four boreholes (identified as BH01 to BH24) were drilled and sampled to delineate the TPH contamination found in the soil samples collected from test pits in July 2002. The boreholes were drilled within and around the large excavated area on the site which was a result of previous partial remediation of the site. The samples submitted for analysis are summarised in Table 3. A copy of the summarised laboratory results is provided in Appendix C; sample locations are shown on Figure 2.

TABLE 3						
Analysis Schedu	Analysis Schedule – August 2002					
Sample	Sample Analyte					
BH1/2.3-2.5						
BH2/2.5-2.7						
BH3/1.0-1.2						
BH3/3.0-3.2						
BH4/3.2-3.4						
BH5/1.8-2.0						
BH8/3.7-3.9						
BH9/2.0-2.2						
BH10/1.0-1.2	_					
BH11/1.0-1.2						
BH12/2.2-2.3	TPH/BTEX					
BH13/0.8-1.3						
BH14/2.0-2.2						
BH15/1.3-1.5						
BH16/1.0-1.2	_					
BH17/0.4-0.7						
BH18/3.3-3.5						
BH20/2.0-2.2						
BH21/1.8	_					
BH22/2.5-2.7	· ·					
BH23/1.3-1.5	-					
BH24/0.8-1.0						

The results of laboratory analysis indicated that the concentrations of TPH and BTEX were within the assessment criteria for all samples, except for the following:

- TPH C₁₀-C₃₆ (4134 mg/kg) in sample BH2/2.5-2.7;
- TPH C₁₀-C₃₆ (4751 mg/kg) in sample BH5/1.8-2.0;
- TPH C_{10} - C_{36} (1774 mg/kg) in sample BH14/2.0-2.2;

観

- TPH C₁₀-C₃₆ (1320 mg/kg) in sample BH15/1.3-1.5;
- TPH C_{10} - C_{36} (13,227 mg/kg) in sample BH18/3.3-3.5; and
- TPH C_{10} - C_{36} (1129 mg/kg) in sample BH20/2.0-2.2.

TPH C_{10} - C_{36} concentrations also exceeded the assessment criterion in samples DUP09 (3524 mg/kg TPH C_{10} - C_{36}) and DUP10 (5833 mg/kg TPH C_{10} - C_{36}), which were duplicate samples of BH2/2.5-2.7. Sample DUP07 (10,795 mg/kg TPH C_{10} - C_{36}), a duplicate of sample BH5/1.8-2.0, also contained TPH C_{10} - C_{36} at a concentration exceeding the assessment criterion.

The TPH (C_{10} - C_{36}) concentrations exceeding the assessment criterion were found in soil samples collected from boreholes located in the western and southern end of the pit area. Furthermore, these soil samples were collected from close to or at the depth of the water table. URS noted that the analytical results were consistent with olfactory indications of contamination noted in the field.

URS also noted that:

No observations of impact (i.e odour) were made in the boreholes completed just beyond the northern boundary of the pit (BH3, BH4, BH7 and BH8). The analytical results of the soil samples supported these observations with TPH concentrations below the laboratory detection limit. Further towards the middle and eastern end of the pit, soil samples were found to be above the laboratory limit of detection but below the guideline value of 1000mg/kg.

4.2.2 Groundwater Sampling

Boreholes BH03, BH05, BH12, and BH18 were completed as groundwater monitoring bores and identified as MW2, MW3, MW4, and MW1, respectively. Monitoring bores MW1 (BH18) and MW2 (BH03) were installed downgradient of the pit, while the other two bores were installed within the pit. Monitoring bore MW3 (BH05) was installed close to the location of test pit TP25, whilst MW4 (BH12) was installed in the eastern upgradient end of the pit. Groundwater was sampled from these boreholes and submitted for analysis for TPH and BTEX. A copy of the summarised laboratory results is provided in Appendix C. The locations of test pits completed as groundwater monitoring bores are shown on Figure 2.

The results of analysis indicated BTEX concentrations were below the limit of reporting for all samples. The TPH concentrations in the sample collected from MW2 was below the limit of reporting, whilst the TPH concentrations in the remaining bores were relatively high, as follows:

- TPH C₆-C₉ (43 μ g/L) and TPH C₁₀-C₃₆ (72,333 μ g/L) in sample MW1;
- TPH C₆-C₉ (39 μ g/L) and TPH C₁₀-C₃₆ (28,659 μ g/L) in sample MW3; and
- TPH C₆-C₉ (<20 μ g/L) and TPH C₁₀-C₃₆ (25,887 μ g/L) in sample MW4.

Furthermore sample DUP19, a duplicate sample of MW1, also contained high concentrations of TPH (132 μ g/L TPH C₆-C₉ and 115,182 μ g/L TPH C₁₀-C₃₆).

4.3 Soil Sampling, September 2002

A further fourteen test pits (identified as TP100 to TP113) were excavated in September 2002 to further delineate the TPH contamination in the vicinity of the pit. At least one sample from each of the pits was submitted for laboratory analysis for TPH. The samples submitted for analysis had generally been collected from a depth of between 3 and 4 metres, the approximate depth at

which the TPH contamination was found in the August 2002 sampling. A copy of the summarised laboratory data is included in Appendix C, whilst sampling locations are shown on Figure 2.

The results of laboratory analysis indicated that the concentrations of TPH for all samples were within the assessment criteria and less than the limit of reporting.

4.4 Discussion of Results

URS prepared summary statistics of the assessment analytical data for the site. The results are tabulated in Table 4 below.

TABLE 4 Summary of Statistics of Assessment Data (mg/Kg)							
Analyte	PQL	Set 🖓	Min.	Max.	Mean	C.V.	95% UCL*
Inorganics		·····				<u>.</u>	
Sulphate	10	11	11	1140	152	2.2	595†
Arsenic	1-5	44	<1	7	1.20	1.2	1.6
Cadmium	1-5	44	<1	<1	0.5	0.0	0.5
Copper	1-5	44	<1	228	20	2.0	71†
Nickel	1-5	44	<1	11	1.5	1.1	1.9
Lead	1-5	77	1	1880	86	2.7	290 [†]
Zinc	1-5	44	<1	433	59	1.7	484 [†]
Chromium	1-5	44	1	11	2.3	0.9	2.8
Mercury	0.1	44	<1	<1	0.074	1.0	0.1
Cyanide	1	11	<1	<u> </u>	0.50	0.0	0.5
Organics							
TPH C ₆ -C ₉ Pit Areas	2	84	1	19	4.8	0.7	5.4
TPH C ₁₀ -C ₃₆ Pit Areas	250	84	110	13200	550	2.9	492 [†]
TPH C ₆ -C ₉ Balance of Site	2	34	1	5	3.8	0.5	4.4
TPH C ₁₀ -C ₃₆ Balance of Site	250	34	110	900	155	1.1	203
РАН	0.05-0.1	66	0.6	40	6.7	1.1	8.2
Benzo(a)Pyrene	0.05	54	< 0.1	3.7	0.56	1.3	0.61
Benzene	0.2	121	<0.2	<0.2	0.28	1	0.3
Toluene	0.2	121	<0.2	< 0.2	0.28	1	0.3
Ethyl benzene	0.2	121	<0.2	< 0.2	0.28	1	0.3
Xylenes	0.2	121	<0.2	1.5	0.77	1	0.9

Notes: PQL laboratory practical quantitation limit

Set number of samples in data set

Min. minimum concentration

Max. maximum concentration

Mean arithmetic mean

t

C.V. coefficient of variation

95 per cent upper confidence limit on the mean is calculated via procedure D (NSW EPA, 1995) unless otherwise indicated.

Procedure G (NSW EPA, 1995) was used to calculate the 95 per cent upper confidence limit on the mean due to the lognormal distribution of the dataset.

URS provided the following discussion of results.

Based on all the available information from the retained historical data (1991-1998) and the 2002 data the following applies:

- All heavy metal analysis reported concentrations below the NEPM HIL D with the exception of a lead sample at TP02. The concentration however is not greater than 250% of the guideline and is therefore not identified as a contamination 'hot spot' in accordance with the NEPM guidelines.
- All samples analysed for OC/OPs, VHCs, cyanide, sulphates and phenols were either below the relevant criteria or below the level of reporting by the laboratory.
- Several samples selected for PAH analysis reported low concentrations of PAHs, however, there were no exceedences of NEPM HIL D for either total PAH or benzo(a)pyrene.
- The investigations were able to establish the general trend for the location of TPH impact within the site. Borehole samples taken from inside the main pit on site such as BH2, BH5, BH14, BH15 and BH20 and one location at the western end outside the pit, BH18, all indicated TPH concentrations in excess of the relevant guidelines. The western end of the main pit appears to be the key area of TPH impacted material.
- No observations of impact (i.e. odour) were made in the boreholes completed just beyond the northern boundary of the pit (BH3, BH4, BH7 and BH8). Similarly, the analytical results of the soil samples supported these observations in that the results were below the laboratory detection limit. As such, it is considered that the extent of any source and/or impact has been established in this area. Further towards the middle and eastern end of the pit, soil samples were found to be odorous and above the laboratory limit of detection but below the guideline value of 1000 mg/kg.
- TPH results outside of the pit at the western end reported concentrations below the limit of reporting by the laboratory. The absence of concentrations of TPH does not correspond to the presence of hydrocarbon odour during investigations. The laboratory had been requested to reanalyse a percentage of the samples to confirm the TPH results.
- Samples analysed for TPH/BTEX or only TPH indicate several samples below the relevant guidelines and/or below the level of reporting from the laboratory.

URS consider that removal of asbestos cement fragments can be undertaken by hand picking followed by inspection by a suitably qualified asbestos consultant, most likely Hibbs and Associates.

On the basis of the limited presence of asbestos fibres in SP1 (1 sample returned positive detection out of a total of 5 samples), it is considered reasonable to retain the stockpile on site as backfill. As a precautionary measure the stockpile should be placed beneath building slabs and/or pavements and that suitable control and contingency measures are taken to minimise dust generation during the relocation of the stockpile.

The groundwater results indicated TPH to be present in three wells (MW1, MW3 and MW4), however MW2 reported concentrations below the limit of reporting by the laboratory. This indicates that the TPH contaminated water is present within the main pit and flowing...in a westerly direction out of the pit. The results for MW2 and the analyses and observations made during the soil investigations, has shown no evidence of TPH contamination to the north of the pit.

4.5 Remedial Action Strategy

The remedial works proposed by URS are outlined in the following sections.

4.5.1 Remediation Objectives

The remediation objectives are to reduce the concentrations of hydrocarbon contaminants (TPH) in the soil and groundwater to levels that are acceptable for the proposed land use and are protective of human health and the environment in line with NSW EPA guidelines. This will also remove the ongoing source of groundwater contamination.

4.5.2 Remediation Area

The main area requiring remediation is the western end of the depression where TPH concentrations generally exceed 1000 mg/kg....A further small area at the eastern end of the depression (in the vicinity of BH 12) also requires remediation to remove a suspected source of groundwater contamination. The total combined area is approximately 1,000 m².

Petroleum hydrocarbon contamination at the site is expected to concentrate in the soil zone close to the water table level, which has fluctuated over a range of perhaps 0.5 m. The average thickness of material requiring remediation is estimated at approximately 3 m (including 0.5 to 1 m of contaminated material below the water table). The indicated volume of soil requiring remediation is therefore about $3,000 \text{ m}^3$.

A further area at the eastern end of the depression, may require some remediation to address odour issues and eliminate sources of groundwater contamination.

4.5.3 Proposed Soil Remediation Program

URS proposes to excavate the contaminated soils to approximately 0.5 to 1 metre below the water table and landfarm the soils on site with the addition of nutrients and regular tilling for a period of three to four months. The rate of progress of remediation will be assessed after one month, to determine whether to proceed with landfarming or dispose of the material to landfill.

4.5.4 Proposed Groundwater Remediation Program

The primary strategy for groundwater remediation will be to remove the source of the contamination via the soil remediation program. Groundwater within the remedial excavation pit will be sampled to determine the concentrations of contaminants prior to determining a suitable method of removing and disposing of the groundwater.

Should the proposed remediation technique prove to be ineffective or too slow, alternative options for remediation may be recommended. These may include removal of more additional soil, in-situ bioremediation, or in-situ chemical oxidation.

4.6 URS's Conclusions

URS concluded the following:

Based on the results from all investigations 1991-2002 the majority of the site is suitable for residential development in accordance with the exposure setting relevant to NEPM Level D. Based on the 2002 TPH results, however, a portion of the site is currently unsuitable for residential development. It will be therefore be necessary to conduct some remediation works to address soil and groundwater contamination within and surrounding the main pit area.

URS consider that the remediation works can be completed using routine methodologies and, following completion of the works, the entire site will be able to be validated for medium density residential land use.

4.6.1 Auditor's Comments

The Auditor has reviewed the report concerned and considers that the course of action proposed by URS is appropriate.

5.0 COMPLETENESS AND ADEQUACY OF INVESTIGATION

5.1 Initial Sampling Strategy

The confirmation sampling strategy adopted by URS is considered appropriate.

5.2 Sampling Procedures

The sampling procedures adopted by URS, as outlined in the assessment report, have been reviewed. These procedures are considered to comply with general industry standards and to be adequate to ensure the integrity of the data set used to assess contamination and validate remediation on this site.

5.3 Quality Assurance/Quality Control

The quality assurance and quality control procedures adopted by URS, as outlined in the assessment report, have been reviewed. These procedures are considered to comply with general industry standards and to be adequate to ensure the integrity of the data set used to assess contamination and validate remediation on this site.

A review of the laboratory QC results presented in the assessment report has also been undertaken. A copy of the QC portion of the auditing checklist has been provided in Appendix D.

The QA/QC criteria examined in this review included:

- Precision
- Accuracy
- Sensitivity
- Representativeness
- Comparability
- Completeness
- Holding times
- Blanks

It was noted that URS did not obtain equipment (rinsate) blanks. URS provided an explanation of its procedure with regard to blanks.

The Auditor considers that the overall quality of data and their presentation are of an adequate standard to support the conclusions he has reached.

5.4 Groundwater Issues

Groundwater issues remain to be addressed in a remedial action plan.

5.5 Aesthetic Issues

Aesthetic issues remain to be addressed in a remedial action plan.

The Auditor does not consider that the potential for chemical mixtures is an issue of concern on this site. $\$

5.7 Reporting Standards

Although minor inconsistencies were identified in the assessment report, the assessment report generally complies with the NSW EPA's *Guidelines for Consultants Reporting on Contaminated Sites* (1997).

6.0 CONCLUSIONS

This audit was carried out for the purpose of determining:

- (1) the nature and extend of any contamination of the land,
- (2) the nature and extent of the investigation or remediation,
- (3) what investigation or remediation remains necessary before the land is suitable for residential use with minimal opportunity for soil access.

It has been determined that parts of the site were potentially contaminated by:

- heavy metals,
- total petroleum hydrocarbons (TPH),
- benzene, toluene, ethyl benzene and xylenes (BTEX),
- polycyclic aromatic hydrocarbons (PAH), and
- asbestos-containing materials (ACM)

as a result of its previous use as a bus depot and subsequent demolition work.

Remediation and validation of the site have previously been carried out, but it has been determined that this work was not performed and documented in a manner that meets current standards.

Investigations carried out by URS in 2002 were adequate and appropriate. These investigations have established that some residual contamination (or recontamination) by petroleum hydrocarbons is present in part of the site and that both soil and groundwater are impacted, to the extent that remediation is required before the site will be suitable for residential use.

It is considered that the remedial strategy proposed by URS is appropriate for the site. It is therefore recommended that:

(1) A formal remedial action plan (RAP) be prepared.

(2) The RAP should include procedures for:

- Excavation of hydrocarbon-contaminated soil
- On-site treatment by land farming
- Validation sampling and analysis

- Groundwater monitoring
- Contingency groundwater remediation and validation.
- 3) The RAP be reviewed and approved by an Auditor. That approval should be given in letter form.
- 4) A validation report be prepared.
- 5) A further site audit be carried out and a new site audit statement issued.

Subject to successful implementation of these recommendations, it is considered that the site can be rendered suitable for residential use with minimal opportunity for soil access.

The Site Audit Statement should be used in conjunction with this Summary Site Audit Report which contains important supporting information.

REFERENCES

Australian and New Zealand Environment and Conservation Council/Agriculture and Resource Management Council of Australia and New Zealand (2000): Australian and New Zealand Guidelines for Fresh and Marine Water Quality. National Water Quality Management Strategy.

Australian and New Zealand Environmental and Conservation Council/National Health and Medical Research Council (1992): Australian and New Zealand Guidelines for the Assessment and Management of Contaminated Sites.

Imray, Paula and A. Langley (1999): *Health-Based Soil Investigation Levels*, National Environment Protection (Assessment of Site Contamination) Measure, Schedule B, Guideline 7A.

NSW Environment Protection Authority (1994): *Guidelines for Assessing Service Station Sites*, Chatswood NSW.

NSW Environment Protection Authority (1995): Sampling Design Guidelines, Chatswood NSW.

NSW Environment Protection Authority (1997): Guidelines for Consultants Reporting on Contaminated Sites, Chatswood NSW.

NSW Environment Protection Authority (1998): *Guidelines for the NSW Site Auditor Scheme*, Chatswood NSW.

Taylor, Roscoe and A. Langley (1999): *Exposure Scenarios and Exposure Settings*, National Environment Protection (Assessment of Site Contamination) Measure, Schedule B, Guideline 7B.

C.M. Jewell & Associates Pty Ltd



Water and Environmental Management ACN 056 283 295

IMPORTANT INFORMATION about your ENVIRONMENTAL SITE ASSESSMENT

These notes have been prepared by C.M. Jewell & Associates using guidelines prepared by the National Ground Water Association (NGWA) and other sources. They are offered to help you in the interpretation of your Environmental Site Assessment (ESA) reports.

REASONS FOR CONDUCTING AN ESA

ESAs are typically, though not exclusively, carried out in the following circumstances:

- as pre-acquisition assessments, on behalf of either purchaser or vendor, when a property is to be sold;
- as pre-development assessments, when a property or area of land is to be redeveloped or have its use changed – for example, from a factory to a residential subdivision – as a requirement for development approval;
- as pre-development assessments of greenfield sites, to establish "baseline" conditions and assess environmental, geological, hydrological constraints to the development of, for example, a landfill; and
- as audits of the environmental effects of an ongoing operation.

Each of these circumstances requires a specific approach to the assessment of soil and groundwater contamination. In all cases, however, the objective is to identify and if possible quantify the risks which unrecognised contamination poses to the proposed activity. Such risks may be financial (for example, clean-up costs or limitations on site use), or physical (for example, health risks to site users or the public).

THE LIMITATIONS OF AN ESA

Although the information provided by an ESA can reduce exposure to such risks, no ESA, however diligently carried out, can eliminate them. Even a rigorous professional assessment may fail to detect all contamination on a site. Contaminants may be present in areas that were not surveyed or sampled, or may migrate to areas which showed no signs of contamination when sampled.

The extent of sampling and subsequent analysis of soils is necessarily limited, and is generally targeted towards areas where contamination is considered to be most likely, based on the knowledge of the site history and visual observation. This approach maximises the probability of identifying contaminants; however, it may not identify contamination which occurs in unexpected locations or from unexpected sources.

Further, soil, rock and aquifer conditions are often variable, resulting in non-homogenous contaminant distributions across a site. Contaminant concentrations are identified at chosen sample locations; however, conditions between sample locations can only be inferred on the basis of the estimated geological and hydrogeological conditions and the nature and extent of identified contamination. Boundaries between zones of variable contamination are often indistinct, and must be interpreted based on available information and the application of professional judgement. The accuracy with which subsurface conditions can be characterised depends on the frequency and methods of sampling and the uniformity of subsurface conditions and is therefore limited by the scope of works undertaken.

ESA "FINDINGS" ARE PROFESSIONAL ESTIMATES

Site assessment identifies actual subsurface conditions only at those points where samples are taken, when they are taken. Data derived through sampling and subsequent laboratory testing are interpreted by geologists, engineers or scientists who then render an opinion about overall subsurface conditions, the nature and extent of contamination, its likely impact on the proposed development and appropriate remediation measures. Statistical tools may be used to assist in such assessment, but the validity of conclusions depends entirely on the degree to which the original data reflects site conditions. Actual conditions may differ from those inferred to exist, because no professional, no matter how qualified, and no subsurface exploration program, no matter how comprehensive, can reveal what is hidden by earth, rock and time. The actual interface between materials may be far more gradual or abrupt than a report indicates. Actual conditions in areas not sampled may differ from predictions.

Nothing can be done to prevent the unanticipated, but steps can be taken to help minimise its impact. For this reason, owners should retain the services of their consultants through the development stage, to identify variances, to conduct additional tests which may be



C.M. Jewell & Associates Pty Ltd *Water and Environmental Management* ACN 056 283 295

needed, and to recommend solutions to problems encountered on site.

SUBSURFACE CONDITIONS CAN CHANGE

Subsurface conditions are changed by natural processes and the activity of man. Because an ESA report is based on conditions which existed at the time of subsurface exploration, decisions should not be based on an ESA report whose adequacy may have been affected by time. Speak with the consultant to learn if additional tests are advisable.

ESA SERVICES ARE PERFORMED FOR SPECIFIC PURPOSES AND PERSONS

Every study and ESA report is prepared in response to a specific Brief to meet the specific needs of specific individuals. A report prepared for a consulting civil engineer may not be adequate for a construction contractor, or even some other consulting civil engineer. A report should not be used by other persons for any purpose, or by the client for a different purpose. No individual other than the client should apply a report even apparently for its intended purpose without first conferring with the consultant. No person should apply a report for any purpose other than that originally contemplated without first conferring with the consultant.

AN ESA REPORT IS BASED ON A UNIQUE SET OF PROJECT SPECIFIC FACTORS

Your environmental report should not be used:

- when the nature of the proposed development is changed - for example, if a residential development is proposed instead of a commercial one;
- when the size or configuration of the proposed development is altered;
- when the location or orientation of the proposed structure is modified;
- when there is a change of ownership; or
- for application to an adjacent site.

To help avoid costly problems, refer to your consultant to determine how any factors which have changed subsequent to the date of the report may affect its recommendations.

AN ESA REPORT IS SUBJECT TO MISINTERPRETATION

Costly problems can occur when design professionals develop their plans based on misinterpretations of an ESA. To help avoid these problems, the environmental consultant should be retained to work with appropriate design professionals to explain relevant findings and to review the adequacy of their plans and specifications relative to contamination issues.

-

1

LOGS SHOULD NOT BE SEPARATED FROM THE GEOLOGICAL REPORT

Final borehole or test pit logs are developed by environmental scientists, engineers or geologists based upon their interpretation of field logs (assembled by site personnel) and laboratory evaluation of field samples. Only final logs are customarily included in our reports. These logs should not under any circumstances be redrawn for inclusion in site remediation or other design drawings, because drafters may commit errors or omissions in the transfer process. Although photographic reproduction eliminates this problem, it does nothing to minimise the possibility of contractors misinterpreting the logs during bid preparation. When this occurs, delays, disputes and unanticipated costs are the all-too-frequent result.

To reduce the likelihood of borehole log misinterpretation, the complete report must be available to persons or organisations involved in the project, such as contractors, for their use. Those who do not provide such access may proceed under the mistaken impression that simply disclaiming responsibility for the accuracy of subsurface information always insulates them from attendant liability. Providing all the available information to persons and organisations such as contractors helps prevent costly construction problems and the adversarial attitudes which may aggravate them to disproportionate scale.

READ RESPONSIBILITY CLAUSES CLOSELY

Because an ESA is based extensively on judgement and opinion, it is necessarily less exact than design documents produced by other disciplines. This situation has resulted in wholly unwarranted claims being lodged against consultants. To help prevent this problem, model clauses have been developed for use in written transmittals. These are not exculpatory clauses designed to foist liabilities onto some other party. Rather, they are definitive clauses which identify where your consultant's responsibilities begin and end. Their use helps all parties involved recognise their individual responsibilities and take appropriate action.

Some of these definitive clauses are likely to appear in your ESA report, and you are encouraged to read them closely. Your consultant will be pleased to give full and frank answers to your questions.







· •

<u>_</u>...

• •

. . .

. .

·

.

. .

.



APPENDIX A

Contaminant Groups
Individual Species Making up Contaminant Groups

POLYCYCLIC AROMATIC HYDROCARBONS

Naphthalene Acenaphthylene Acenaphthene Fluorene Phenanthrene Anthracene Fluoranthene Pyrene Benz(a)anthracene Chrysene Benzo(b) & (k) fluoranthene Benzo(b) & (k) fluoranthene Benzo(a)pyrene Indeno(1,2,3-cd)pyrene Dibenz(a,h)anthracene Benzo(g,h,i)perylene

TOTAL PETROLEUM HYDROCARBONS

 $C_6 - C_9$ Fraction $C_{10} - C_{14}$ Fraction $C_{15} - C_{28}$ Fraction $C_{29} - C_{36}$ Fraction

MONOCYCLIC AROMATIC HYDROCARBONS

Benzene Toluene Ethylbenzene meta- & para-Xylene ortho-Xylene

HEAVY METALS

Arsenic (As) Cadmium (Cd) Chromium (Cr) Copper (Cu) Lead (Pb) Mercury (Hg) Nickel (Ni) Zinc (Zn)

> C.M. Jewell & Associates Pty Ltd Water and Environmental Management



APPENDIX B

Communications with the Auditor

. 1



C. M. Jewell & Associates Pty Ltd Water and Environmental Management A.B.N. 54 056 283 295

fax transmission

P.O. Box 10, Wentworth Falls, NSW 2782, Australia 1/13 Kalinda Road, Bullaburra, NSW 2784, Australia



 Phone (02) 4759 3251
 Email postie@cm-jewell.com.au
 Fax (02) 4759 3257

 (International +61 247 59 3251)
 (International +61 247 59 3257)

Ref: J0807.2	Date: 26 June 2002	Time: 12:02
To: URS		c.c. McLachlan Lister
Attention: Fran Mitchell		Attention: Simon Magri
Fax No. 8925 5555	From: Chris Jewell	Fax No. 9241 1898
Original to follow: No		Total pages including cover: 1

Subject: Supplementary Sampling and Analytical Plan, Lot 202 King Street, Randwick

Fran,

Further to our telephone conversation today, I confirm that I have reviewed your supplementary sampling and analytical plan and am generally happy with the approach that you have proposed.

There are two areas where some modification is appropriate. These are:

- 1) Comparability of data sets. Unless you are sure that you can demonstrate consistency of sampling and analytical methodology, allowing the original and supplementary data sets to be combined, then it would be worthwhile including some overlap sampling to allow a direct comparison.
- 2) Asbestos. I would like you to follow the approach to asbestos assessment outlined in the ACLCA Code of Practice (February 2002 draft), taking into account the comments in enHealth's review of that document. I note that you intend to sample from test pits, and concur with that approach. I would like to see your sampling program supplemented, and placed in context, by a walk-over inspection of the whole site by an asbestos specialist.

I don't think that either of these suggestions will result in significant extra costs or delays, and should help ensure that we end up with a validation data set with which we are all comfortable.

For and on behalf of <u>C. M. JEWELL & ASSOCIATES PTY LTD</u>

CHRIS JEWELL



C. M. Jewell & Associates Pty Ltd Water and Environmental Management A.B.N. 54 056 283 295

fax transmission

P.O. Box 10, Wentworth Falls, NSW 2782, Australia 1/13 Kalinda Road, Bullaburra, NSW 2784, Australia



 Phone (02) 4759 3251
 Email postie@cm-jewell.com.au
 Fax (02) 4759 3257

 (International +61 247 59 3251)
 (International +61 247 59 3257)

Ref: J0807.4	Date: 27 September, 2002	Time: 12:04
To: URS		c.c. McLachlan Lister
Attention: Fran Mitchell		Attention: Simon Magri
Fax No. 8925 5555	From: Chris Jewell	Fax No. 9241 1898
Original to follow: No		Total pages including cover: 2

Subject: Lot 202 King Street, Randwick

Fran,

I have carried out a review of your Data Assessment Report. I have some comments and have noted several issues which require clarification. These are as follows:

- 1. Could you please explain samples T3006-S1 and S3 and the VD series samples from 1998. Are these discrete samples, or area composites, or ?
- 2. Please indicate in which part of the site the 1998 samples RBD001, RBD002, and RBD003 are located.
- 3. Please identify what stockpile sample SP11-9 represents.
- 4. URS' report indicates that three samples were composited to produce a stockpile sample. Stockpile samples were analysed for BTEX, although composite samples are not suitable for analysis of volatiles. Please comment.
- 5. It appears that the results of analysis of stockpile samples have not been assessed against modified assessment criteria to take into account the effect of compositing. Please comment.
- 6. Figure 2 in your report identifies the location of a former trade waste pit. Please comment.
- 7. The TPH concentration for sample SZ-54 in Table 1 should be 700 mg/kg. This sample is located in the south-west corner of the site. Please comment on the likely source of the TPH in this sample.
- 8. The TPH concentrations for samples Z2-1 to Z2-6 in Table 1 are not consistent with the laboratory reports. The TPH concentration for these samples should be not detected.
- 9. It appears as though no wash blanks were collected during the course of the soil and groundwater sampling. Please comment.
- 10. Please provide a statistical summary of the data used to assess the site.

11. In the 1998 laboratory reports provided there are samples named SP2 and SP3. What do these samples represent? Do these samples relate to current stockpile numbers?

For and on behalf of C. M. JEWELL & ASSOCIATES PTY LTD

CHRIS JEWELL

C. M. Jewell & Associates Pty Ltd Water & Environmental Management

30.Sep	. 2002 12:54 No.0218 P. 1/5
UI	RS Fax Transmittal
To <u>:</u>	Chris Jewell
Company	CM Jewell & Associates $30.9_{-5,2}$
Facsimile:	4759 3257 DOB07 CHJ
From:	Fran Mitchell
CC:	Tim Greenaway (9241 1898)
Date:	30 September 2002
Page 1 of	5
Special nstructions	Confidential Urgent Please Reply For Your Information For Follow-up
	If you do not receive all pages or transmission is illegible, please contact the originator to re-send. Should the facsimile be sent to the wrong fax number, would receiver please destroy this copy and notify URS immediately. Thank you.
х I	
Subject:	Lot 202 King St - Response to Auditors Queries URS Project No: 51072-001 (fax 27/9/02)
Message:	Chris
· ·	Thank you for the comments from your review of the Data Assessment Report (20 September 2002) for Lot 202 King St, Randwick site. Please find the following responses:
1	1. Could you please explain samples T3006-S1 and S3 and the VD series samples from 1998. Are these discrete samples, or area composites, or?
	Samples T3006-S1, T3006-S3 and VD1, VD3, VD4, VD6 and VD RAMP were collected during 1998 as area composite samples.
	2. Please indicate in which part of the site the 1998 samples RBD001, RBD002, and RBD003 are located.
,	Samples RBD001, RBD002, RBD003 were collected beyond the southern face of the main pit excavation in grid location F5 to G5. These samples were discrete samples collected from the approximate 30m length illustrated in the attached figure.

URS Australia Pty Ltd (ABN 46 000 691 690) Level 3, 116 Miller Street North Sydney, NSW 2060 Australia Tel: 61 2 8925 5500 Fax: 61 2 8925 5555

y,



To: Chris Jewell CM Jewell & Associates 30 September 2002 Page 2

3. Please identify what stockpile sample SP11-9 represents.

Sample SP11-9 was a composite sample, composited from the individual stockpiles SP9, SP10 and SP11. Each of these stockpiles consist of sandy soil and the total volume is approximately 100 m³, less than the nominal sampling frequency of 300m³.

4. URS' report indicates that three samples were composited to produce a stockpile sample. Stockpile samples were analysed for BTEX, although composite samples are not suitable for analysis of volatiles. Please comment.

It is agreed that composite samples are typically not suitable for volatile analyses.

BTEX has been selected as a contaminant of concern for the site in general from the past site use as bus refuelling depot. BTEX results have been less than the laboratory limit of detection for the extent of residual, in-situ samples for both the historical sampling and the more recent sampling. Notwithstanding the absence of BTEX in the residual samples, it was not considered that BTEX would be a contaminant of concern for the stockpiles due to the amount of material rehandling which would have occurred in the excavation and replacement of these materials.

5. It appears that the results of analysis of stockpile samples have not been assessed against modified assessment criteria to take into account the effect of compositing. Please comment.

The composite samples collected were composited from 3-point sub-samples. URS consider that when sampling a stockpile it is not appropriate to divide the guideline value by the number of sub-samples used to prepare the composite. The purpose of collecting a composite sample from a stockpile is to ensure collection of a representative sample that provides an indication of the average concentration of the stockpile at an appropriate sampling density.

Nonetheless, URS has reviewed the data against the revised guideline value (divided by 3). The outcome is that the composite sample from SP1 exceeds the revised guideline value for TPH (C10-C36). The analytical result for SP1 was 400mg/kg in comparison with the revised guideline value of 333 mg/kg (1000 mg/kg divided by 3). This result does not exceed the revised guideline value by more than 250%.

6. Figure 2 in your report identifies the location of a former trade waste pit. Please comment.

The former trade waste pit illustrated on the drawing (Figure 2, Data Assessment Report, 20 September 2002) in the north-western corner of the site is a derelict pit associated with the former bus depot operations. The condition of the pit and its underlying surface shall be assessed during the remediation program planned for this vicinity of the site.

\$



Fax Transmittal

To: Chris Jewell CM Jewell & Associates 30 September 2002 Page 3

7. The TPH concentrations for sample SZ-54 in Table 1 should be 700 mg/kg. This sample is located in the south-west corner of the site. Please comment on the likely source of the TPH in this sample.

Sample SZ-54 was collected as a discrete sample from a remedial excavation (known as Zone 5) undertaken in 1995. The analytical result is 700 mg/kg (Amdel laboratory certificate No. 9502353) and the non-detect result reported in Table 1 (Data Assessment Report, 20 September 2002) is an error in the data compilation.

The excavation was completed following a site investigation program completed in 1994 where an isolated area of TPH impact was identified. The source of this TPH impact was considered to be associated with waste fuels and/or oils from a former workshop building. The 'Zone 5' excavation was undertaken beyond the western end of the footprint of a former workshop building. The building footprint was located in grid H1 through to E1, aligned in a east-west direction, approximately 60m long by 20m wide.

During the course of the excavation in 1995, observations of impact were limited. The excavation continued beyond the extent of any visual impact observations and was extended in depth to bedrock at a depth of 3.5m.

8. The TPH concentrations for samples Z2-1 to Z2-6 in Table 1 are not consistent with the laboratory reports. The TPH concentration for these samples should be not detected.

Agreed. This is a data entry error from the data compilation. The correct analytical results for these samples are presented in Amdel laboratory certificate No. 9505659.

9. It appears as though no wash blanks were collected during the course of the soil and groundwater sampling. Please comment.

Rinsate blanks have not been collected as part of the recent groundwater sampling as URS used single use disposable bailers for purging and sampling. Rinsate blanks have not been collected for recent soil sampling. Soil rinsate blanks can provide an indication of the potential for cross contamination of soil samples from sampling equipment. However, the data collected cannot be interpreted in any meaningful way an as such URS do not normally collect rinsate blanks for soil. Should cross contamination occur, then this could result in a false positive being reported and therefore it would, if it occurred, overestimate not underestimate the extent of contamination. However, if it did occur the rinsate blanks do not provide any information that could be used to estimate that magnitude or extent of cross contamination. It is for this reason no rinsate blanks for soils were collected.

TIRS

ł

Fax Transmittal

To: Chris Jewell CM Jewell & Associates 30 September 2002 Page 4

10. Please provide a statistical summary of the data used to assess the site.

This statistical summary will follow this fascimile later today.

Should you require any further information to assist with your review of the report, please do not hesitate to contact either of the undersigned.

Regards, URS Australia Pty Ltd

Fran Mitchell Project Manager

Dr Martin Howell Project Director



____N_____

		· 2002 16:01 No.0221 P. 1
	U	RS Fax Transmittal
	1	n an
ć		Chris Jewell 8700 30.9.02 J0807 JOB07 CMJ 4759 3257 TP
	. To :	Chris Jewell 30.9.02
	Company.	CM Jewell & Associates
	Facsimile:	4759 3257
,	From	Fran Mitchell
`	CC:	Tim Greenaway (9241 1898)
·	Date:	30 September 2002
	Page 1 of	6
	Special Instructions:	Confidential Urgent Please Reply For Your Information For Follow-up
	× •	If you do not receive all pages or transmission is illegible, please contact the originator to re-send. Should the facsimile be sent to the wrong fax number, would receiver please destroy this copy and notify URS immediately. Thank you.
· · · · · · · · · · · · · · · · · · ·	Subject:	Lot 202 King St - Response to Auditors Queries URS Project No: 51072-001 (fax 27/9/02)
	Message:	Chris
	, , ,	Further to our fax of earlier today, please find following the response to your Question No. 11:
		11. In the 1998 laboratory reports provided there are samples named SP2 and SP3. What do these samples represent? Do these samples relate to current stockpile numbers.
	, , ,	These sample numbers do not relate to the current stockpile numbering system. Site records from 1998 suggest that the material from former SP2 and SP3 was disposed off-site. However, this could not be confirmed for validation purposes and as such, a decision was made to sample and analyse the stockpiles located presently on-site.
: : :		Should you require any further information to assist with your review of the report, please do not hesitate to contact either of the undersigned.
		Regards, URS Australia Pty Ltd Molecule Fran Mitchell Project Manager Dr Martin Howell Project Director
· ·		N URS Australia Pty Ltd (ABN 46 000 691 690) Level 3, 116 Miller Street North Sydney, NSW 2060 Australia Tel: 61 2 8925 5500 Fax: 61 2 8925 5655 NSYD1/PROJECTS/PROJECTS/S1072/001/CORRESPONDENCE/F020_AUDITOR RESPONSE.DOC

30.Sep. 2002 15:25 No.0219 P. 1 Fax Transmittal Ľ To: Chris Jewell Company: CM Jewell & Associates Facsimile: 4759 3257 From: Fran Mitchell CC. Tim Greenaway (9241 1898) Date: 30 September 2002 Page 1 of 6 Special Confidential Urgent Please Reply ☐ For Your Information For Follow-up istructions: If you do not receive all pages or transmission is illegible, please contact the originator to re-send. Should the facsimile be sent to the wrong fax number, would receiver please destroy this copy and notify URS immediately. Thank you. Subject: Lot 202 King St - Response to Auditors Queries URS Project No: 51072-001 (fax 27/9/02) Message: Chris Further to our fax of earlier today, please find following a summary of the statistical assessment of the data compiled as part of the characterisation of Lot 202 King St, Randwick.

10. Please provide a statistical summary of the data used to assess the site.

Statistical analyses have been completed for the residual data sets for the site. Statistics for PAHs and inorganics have been completed across the site as a whole. Statistics for TPH have been divided into those sampling locations associated with the main pit area and those sampling locations associated with the balance of the site. The two separate data sets for TPH are presented in the attached tables.

The attached summary table provides a summary of the statistical analysis. The assessment illustrates that a number of sample results exceed more than 250% of the guideline value for TPH (C9-C36). The area represented by these samples falls within the extent of the footprint proposed for remediation, as referred to in Section 6 of the Data Assessment Report.

URS Australia Pty Ltd (ABN 46 000 691 690) Level 3, 116 Miller Street North Sydney, NSW 2060 Australia Tel: 61 2 8925 5500 Fax: 61 2 8925 5555 URS

Ş,

Fax Transmittal

To: Chris Jewell CM Jewell & Associates 30 September 2002 Page 2

Should you require any further information to assist with your review of the report, please do not hesitate to contact either of the undersigned.

Regards, URS Australia Pty Ltd

Fran Mitchell Project Manager

SJE

Dr Martin Howell Project Director

TPH Statistical Summary	Residual Data Set	Lot 202 King St Randwick
--------------------------------	-------------------	--------------------------

Analyde					chinese isonitario	_			OldBuco	Statistical Mualysis		
	Mean	Maximum Concentration for Single Semple ²	PQL ³	Number Analysed	Maximum	Maimum	Distribution Type ⁴	Mean	Median	Ş	Std Deviation	1201%58
Inorganics												
Sulfate - Soluble	2000	5000	¢	1	1140	5	Log-Nomal	152,455	8	22	525	501.6
Arsenic - Total	400	1000	-÷	4	2	v	Normat	1.205	0.5	12	1.4	16
Cadmium - Total	80	200	1-5	4	v	⊽	Normal	0.500	⊽	00	0.0	02
Copper - Total	4000	10000	1-5	4	228	⊽	Log-Normal	20.273	5.5	2.0	40.3	712
Nickel - Total	2400	6000	1-5	4	۲	⊽	Normat	1.511	-	Ð	17	19
	1200	3000	1-5	22	1880	-	Log-Normal	85.571	8	2.7	231.2	288.7
Zinc - Total 24	28000	70000	1-5	44	433	⊽	Log-Normal	59.045	EZ	17	100.5	484.4
Total Chromium	400	1000	1.5	44	Ŧ	1	Normal	2.284	2	0.9	2.0	28
	60	150	0.1	44	V	v	Normal	0.074	v	1.0	0.1	10
Total Cyankte 1	1000	2500	•	+	-	⊽	Normal	0.500	0.5	0.0	0.0	5.0
Grganica												
TPH C ₆ -C ₉ . Pit Area	65	162.5	2	8	19	-	Normal	4.8	5	0.7	3.4	5.4
	1000	2500	250	84	13227	110	Log-Normat	551.9	125	2.9	1580.4	492.2
	65	t62.5	2	34	5	-	Nomal .	3.8	цэ I	0.5	1.8	4.4
iss. Balance of Site	1000	2500	250	평	300	110	Normal	155	110	1.1	165.7	203.2
PAH(total)	ิส	50	0.05-0.1	8	40.05	8:0	Nomal	6.717	*	÷	7.4	82
Benzo(a)pyrene	2	5	6.05	54	3.70	. ₹	Log-Normal	0.556	0.25	E.1	0.7	ĝ.
Benzene		2.5	0.2	121	40.2 40.2	40.2	Normal	0.275	40,2	-	0	0.3
Toluene	1.4	3.5	0.2	121	<0.2	8.8 2	Normal	0.275	<0.2	-	0	0.3
zene	3.1	7.8	0.2	121	<0.2	2.9	Normai	0.275	<0.2	-	0	0.3
Xylenes	14	35	0.2	121	1.5	<0.2	Nonnai	0.769	0.2	-	-	50

1: NEPM HIL - Level D and NSW EPA Service Station Guidelines

.

2: Maximum allowable concentration for a single sample

3: Laboratory Practical Quantification Limit (or Limit of Reporting)

Raf : Sumpfing Design Guidelines, 1995

.

4: Normel - 95% upper confidence limit (UCL) of the arithmetic average concentration. Log Normal - 95% UCL of the average concentration for a lognormal distribution

Page 1 of 1

S:projects\51072\001\Data Sets\Stat Summary_LW

, . . . |

"|

• •

. 1

. .

¢

`

ł

2

Ň

TPH Analytical Data (mg/kg) Main Pit Area

Units SAMPLE BH8_3.7-3.9 BH17_0.4-0.7 BH1_2.3-2.5	TPH (C6-C9) mg/kg 10	TPH (C10-36) mg/kg
SAMPLE BH8_3.7-3.9 BH17_0.4-0.7		
BH8_3.7-3.9 BH17_0.4-0.7	10	
BH17_0.4-0.7		125
	10	125
	10	125
BH3 1.0-1.2	10	125
BH3_3.0-3.2	10	125
BH20_2.0-2.2	10	
BH10 1.0-1.2	10	816
BH11 1.0-1.2	10	707
BH12 2.2-2.3	10	465
BH21_1.7	10	125
BH22_2.5-2.7	10	642
BH13_0.8-1.3	10	130
BH9_2.0-2.2	10	445
BH2_2.5-2.7	10	4034 - mail
BH14_2.0-2.2	10	
BH15_1.3-1.5	10	1350
BH16 1.0-1.2	10	795
BH5 1.8-2.0	10	
BH18 3.3-3.5	13	
BH4 3.2-3,4	10	125
BH23 1.3-1.5	· · · · · · · · · · · · · · · · · · ·	125
	·······	
BH24_0.8-1.0	1	606
TP100 2.0-2.2	1	125
TP101 3.0-3.2	1	125
TP102 1.0-1.2	1	125
TP103 3.5-3.7	1	125
TP104 1.0-1.2	. 1	125
TP104 3.5-3.7	1	125
TP105 3.0-3.2	1	125
TP106 3.0-3.2	1	125
TP107 3.5-3.7	1	125
TP108 3.5-3.7	1	125
TP109 3.0-3.2	1	125
TP110 3.0-3.2	1	125
TP111 2.0-2.2	1	125
TP112 3.0-3.2	1	125
TP113 2.0-2.2	1	125
TP113 3.0-3.2	1	125
TP22_0.3-0.5	1	125
TP22_0.3-0.3	1	125
TP25_1.3-1.5	1	AND THE REPORT OF A DESCRIPTION OF A DES
TP26_1.3-1.5	1	839
TP14_1.3-1.5	1	643
TP25_0.5-0.7	<u>1</u>	448
	··· ·· ··	· · · · · · · · · · · · · · · · · · ·
TP26_0.3-0.5	i j	169
TP26_2.9-3.0	1	953
TP14_0.3-0.5	1	125
TP14_2.4-2.6	1	770
Z2-1	5	110

Ref: S:\51072\001\datasets\TPH stats_pit area.xls DLANK_ASTM

URS

;

j

: 1

5 J

. .

х i

ı.

1.1

τ

.

TPH Analytical Data (mg/kg) Main Pit Area

ł.

	TPH (C6-C9)	TPH (C10-36)
Units	mg/kg	mg/kg
SAMPLE		
Z2-2	5	110
Z2-3	5	110
Z2-4	5	110
Z2-5	5	110
Z2-6	5	110
Z2-7	5	110
Z2-12	5	110
Z2-13	5	110
Z2-14	5	110
Z2-15	5	110
Z2-16	5	110
Z2-17	5	110
Z2-18	5	110
Z2-19	5	110
Z2-20	5	110
Z2-21	5	110
Z2-22	5	510
Z2-23	5	110
Z2-24	5	110
Z2-25	5	110
Z2-26	5	110
Z2-27	5	110
Z3-1	5	110
Z3-2	5	110
Z3-3	5	• 110
Z3-4	5	110
Z3-8	5	1300
Z3-9	5	400
Z3-10	5	360
Z3-11	5	110
Z3-12	** *** ******	110
Z3-13	5	520
Z3-13	5	110
Z3-14 Z3-15	5	110
Z3-16	ວ 5	110
20-10	⁵	110

ł

TPH Analytical Data (mg/kg) Balance of Site

	TPH (C6-C9)	TPH (C10-36)
Units	mg/kg	mg/kg
SAMPLE		
WINNING	5	
TW S2	5	110
W-S3	5	110
Z8-1 SM	5	
Z6-2	5 11 2 1 2	110 1010
28-3		110
William 78-4		110
Z8-5	5	BIO
Z8-6		
29-7	5 16 19	E994233
Z8-8 W	5	110
SZ 61		110
57-62 57-63		
SZ-03	5 PC	1/10 31. 10 10
SZ 52		A NOT NOT
57252		110
SZ 54		907 H
F/884	9	110
	Bannes Stellinger	
H/483	ан на стану и стану и стану и стану Стану и стану и стану и стану и стану и стану Стану и стану и с	
HAS4 MUNIC		110
HA85	5 Sector Se	no.
TP15_0.3-0.5	1	125
TP15_1.3-1.5	1	125
TP24_0.3-0.5		125
TP23_1.3-1.5 TP21 0.3-0.5		125
TP01 1.3-1.5	······································	125 125
TP03 0.3-0.5		125
TP09 1.3-1.5	······	125
TP6 1.3-1.5		125
TP8 1.3-1.5	1	125
	. ' J	16.0



APPENDIX C

Information Relied Upon by the Auditor

. 1

/ 1 | |

<u>ر</u>...

٢Ŋ

:;

TABLE 1 - 1995 Historical Data

э. ,

										.,							_										. ,		,			<u>.</u>	<u> </u>												_
VHCs	mg/kg		1	1		1	1	1	1	1	1	1	1	1	1	1	1	I	I		I	1	1	ı	1	1	1	I	;	ı	g	Ð	2	g	1	1	1	1	1			1	1	1	1
PCBs	mg/kg		1	1		1	1	1	1	1	ι	1	1	:	I	1	ı	I	1		1	I	1	I	•	1	g	9	Q	Q	1	1	I	1	I	1	1	t	1	1		I	1	ı	1
OC/OPs	mg/kg	-	1	۱		ſ	1	,	1	1	1	1	,	1	1	1	I	1	1		I	I	1	,	I	1	ΟN	QN	QN	QN	L	1	ı	1	I	ι	1	1	1	l		1	ŧ	Ι	:
Benzo(a)	mg/kg	4	<lor< th=""><th>0.5</th><th></th><th>1</th><th>1</th><th>ł</th><th>I</th><th>1</th><th>1</th><th> </th><th> 1</th><th>1</th><th>:</th><th>,</th><th>1</th><th>I</th><th>1</th><th></th><th>1</th><th>1</th><th>I</th><th>J</th><th>1</th><th>1</th><th>t</th><th>1</th><th>1</th><th>1</th><th>I</th><th>I</th><th>ι</th><th>1</th><th>1</th><th>I</th><th>1</th><th>I</th><th>1</th><th>I</th><th></th><th>1</th><th>1</th><th>J</th><th>I</th></lor<>	0.5		1	1	ł	I	1	1		1	1	:	,	1	I	1		1	1	I	J	1	1	t	1	1	1	I	I	ι	1	1	I	1	I	1	I		1	1	J	I
PAH	mg/kg	8	<lor< th=""><th>6.3</th><th></th><th>1</th><th> </th><th></th><th>1</th><th>1</th><th>t</th><th></th><th> </th><th>1</th><th> </th><th>1</th><th>l</th><th>1</th><th>1</th><th></th><th>1</th><th>I</th><th>I</th><th>ł</th><th>1</th><th>1</th><th>1</th><th>1</th><th>1</th><th>1</th><th>1</th><th>I</th><th>I</th><th>ι</th><th>1</th><th>I</th><th>1</th><th>t</th><th>1</th><th>1</th><th></th><th>1</th><th>1</th><th>1</th><th>I</th></lor<>	6.3		1			1	1	t			1		1	l	1	1		1	I	I	ł	1	1	1	1	1	1	1	I	I	ι	1	I	1	t	1	1		1	1	1	I
BTEX	mg/kg		1	1		1	1	t	:	,			1	1	1		t	t	ı		ı	1	1	1	1	1	1		1		QN	QN	Q	Q	Q	ND	Q	g	QN	Q		Ð	Q	I	1
ТРН	толка	1000	1	1	Ì	1	1	1	1	t	1	:		1	1		1	ı	۱		1	1	1	1	1	1	1		1	1	Q	GN	QN	QN	Q	QN	Q	Q	Q	Q		ND	DN	q	006
PP	тдуд	1200	11	120		ŝ	4	4	4	6	14	<lor< th=""><th>83</th><th><lor< th=""><th><lor< th=""><th>40R</th><th>9</th><th><1.0R</th><th>9</th><th></th><th><lor< th=""><th>co</th><th>15</th><th>4.0R</th><th>4LOR</th><th>14</th><th>1</th><th>1</th><th>1</th><th>1</th><th>ł</th><th>t</th><th>1</th><th>:</th><th>I</th><th>1</th><th>1</th><th>1</th><th>1</th><th>1</th><th></th><th>1</th><th>ι</th><th>1</th><th>I</th></lor<></th></lor<></th></lor<></th></lor<>	83	<lor< th=""><th><lor< th=""><th>40R</th><th>9</th><th><1.0R</th><th>9</th><th></th><th><lor< th=""><th>co</th><th>15</th><th>4.0R</th><th>4LOR</th><th>14</th><th>1</th><th>1</th><th>1</th><th>1</th><th>ł</th><th>t</th><th>1</th><th>:</th><th>I</th><th>1</th><th>1</th><th>1</th><th>1</th><th>1</th><th></th><th>1</th><th>ι</th><th>1</th><th>I</th></lor<></th></lor<></th></lor<>	<lor< th=""><th>40R</th><th>9</th><th><1.0R</th><th>9</th><th></th><th><lor< th=""><th>co</th><th>15</th><th>4.0R</th><th>4LOR</th><th>14</th><th>1</th><th>1</th><th>1</th><th>1</th><th>ł</th><th>t</th><th>1</th><th>:</th><th>I</th><th>1</th><th>1</th><th>1</th><th>1</th><th>1</th><th></th><th>1</th><th>ι</th><th>1</th><th>I</th></lor<></th></lor<>	40R	9	<1.0R	9		<lor< th=""><th>co</th><th>15</th><th>4.0R</th><th>4LOR</th><th>14</th><th>1</th><th>1</th><th>1</th><th>1</th><th>ł</th><th>t</th><th>1</th><th>:</th><th>I</th><th>1</th><th>1</th><th>1</th><th>1</th><th>1</th><th></th><th>1</th><th>ι</th><th>1</th><th>I</th></lor<>	co	15	4.0R	4LOR	14	1	1	1	1	ł	t	1	:	I	1	1	1	1	1		1	ι	1	I
Date of	Sampling	Guideline/HIL 'D'	1995	1995		1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995		1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995		1995	1995	1995	1995
Batch		Guic	N95/035203	N95/035204	9606041	9505153	9505153	9505196	9506018	9506018	950601B	9506025	9506018	9505153	9505153	9505153	9505153	9505153	9505153	N95/031688	9505153	9505153	9506025	9506025	9506025	9506025	9505849	9505849	9505849	9505849	9505849	9505849	9505849	9505849	9505802	9505802	9505802	9505802	9505815	9505802	N95/032612	9505815	9505815	9502353	9502353
Laboratory			AGAL	AGAL	Enviromet	Enviromet	Enviromet	Enviromet	Enviromet	Enviromet	Environet	Environmet	Enviromet	Environmet	Foviromet	Frimmet	Fuviromet	Enviromet	Enviromet	AGAL	Enviromet	Enviromet	Enviromet	Enviromet	Enviromet	Enviromet	Amdel	Amdel	Amdel	Amdel	Amdel	Amdel	Amdel	Amdel	Amdel	Amdel	Amdel	Amdel	Amdel	Amdel	AGAL	Amdel	Amdet	Amdel	Amdel
Sample	•		0-0.15	0-0.15		0-0.15	0-0.15	0-0.15	0-0 15	0-0 15	1100	0-0-15	121.00		0-0.5	0-0.5	50-0	0-0.5	0-0.15		0-0.15	0-0.15	0.15-0.3	0.15-0.3	0.15-0.3	0.15-0.3	0-0 15	0-0-15	0-0.15	0-0.15	0-1.5	0-1.5	0-1.5	0-1.5	2.7-2.9	27-29	27-29	66-26	55-57	27-29		2.7-2.9	2.7-2.9	1.0-1.2	0.5-0.7
COCID			V3-1	V3-2		Z2-M1	Z2-M2	Z2-M6	7M-67	72-MB	20 MO	23-M11	NE M1	77.4	6-77	14	7.4	77-5	7.6		1-12	8-12	PS-12	PS-13	PS-14	PS-15	Peet-1	Pest-3	Pest-4	Pest-5	TW-S1	TW-S2	TW-S3	TW-S4	Z8-1	78-2	78-3	78.4	78-5	28-6		Z8-7	Z8-8	SZ-61	SZ-62

Prepared By RSM Checked By Dod 2

Page 1 of 3

TABLE 1 - 1995 Historical Data

л ,

1

.....

		_															,										,								;														,
VHCs	mg/kg		1	ı	1	1	1	1	ł	1	1		1	1	۱	1	ł	1	t	I		1	1	l	1	ł	1	1	t	ł	1		I		1	1	1	1	1	1	I		I	I	1	t		t	1
PCBs	Bylöw		I	1	1	1	ı	l	I	ı	1		1	1	,	1	1	L	1	1	1	ł	I	1	1	1	I	1	1	1	1		1	1	1	1	I	1	1	1	1		1	I	1	1	I	1	1
OC/OPs	mg/kg		:	1	1	1	I	I	1	1	1	.	1	•	I		1	I	1	I	ı	1	1	ı	1	1	1	I	3	t			I	1	1	I	1	1	1	I	I		I	I	1			-	•
Benzo(a)	бурбш	4	1	1	1	I	1	1	1	1	1		t	•	1	1	1	1	1	1	I	1	:	1	ſ	1	1	1	1	I	1		1	1	1	1	1	ι	1	ı	1		1	۱	I	I	1	1	
PAH	тg/kg	80	1	I	1	I	I	1	1	ı	1		1	1	:		1	I	:	1	1	I	ı	1	1	1	I	1	1	1	1		1	1	I	1	I	1	I	I	۲		1	I	I	1	1	I	
втех	Byßw		j	1	I	1	і 0	QN	QN	QN				2	2	QN	QN	QN	g	QN	g	QN	2	QN	GN	Q	g	Q	QN	g	Q		QN	Q	QN	DN	ND	QN	ND	QN	QN		g	QN	QN	QN	QN	g	QN
HAT	mg/kg	1000	QN	QN	QN	Q	pl_QN	QN XQ	A NO	UN X	1.4	22.4	AN R	ank	9	DN	Q	Q	g	QN	Q	Ð	R	QN	g	510	g	g	QN	R	QN		g	9	QN	QN	1300	400	360	GN	QN		520	Q	QN	QN	QN	g	QN
đ	mg/kg	1200	ı	۱	1	I	1	1		1		I	1	ı	'	1	1	1	1	1	1	ı	1	1	I	1	I	1	1	F	1		1	l		:	1	I	5	1	Ι		T	1	I	L	1	1	I
Date of	Sampling	Guideline/HIL 'D'	1995	1995	1995	1995	1995	1995	1995	1995	1005	1001	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995		1995	1995	1995	1995	1995	1995	1995	1995	1995		1995	1995	1995	1995			
Batch		Guic	9502353	9502353	9502353	9502353	9502353	9505659	GENERED			REGENCE	9505659	9505659	9505727	9505668	9505668	9505668	9505668	9505668	9505668	9505727	9505727	0505777	9505777	9505777	9505777	9505777	9505777	9505777	9505815	N95/032613	9503656	9503656	9503656	9503656	9505802	9505802	9505802	9505802	9505849	N95/032831	9505802	9505802	9505802	9505802	9503299	9503299	9503299
Laboratory			Amdel	Amdel	Amdel	Amdel	Amdel	Amdel	Amdal	Amdel		Amdel	Amdel	Andel	AGAL	Amdel	Amdel	Amdel	Amdel	Amdel	Amdel	Amdel	Amdel	Amdel	AGAL	Amdel																							
Sample Depth			0.5-0.7	1.5-1.7	3.0-3.2	1.5-1.7	1.5-1.7	22.24	1000	+ 7 4 4 4	+	2.2-2.4	2.2-2.4	4.5-4.7	2.2-2.4	2.2-2.4	2.2-2.4	2 2-2.4	2 2-2 4	2 2-2 4	2 2-2 4	2 2-2 4	1000	1517	4547	45.47	4547	20-00	2 2 - 2 4	20.00	2 2.2 4		15-17	1.5-1.7	1.5-1.7	1.5-1.7	3.0-3.2	3.0-3.2	3.0-3.2	3.0-3.2	3.0-3.2	3.0-3.2	3.0-3.2	1.5-1.7	3.0-3.2	1.5-1.7	0.8-1.0	0.8-1.0	0.8-1.0
COCID			SZ-63	SZ-51	SZ-52	SZ-53	S7-54	73-1	1.5	72.7	3	22-4	Z2-5	Z2-6	Z2-7	22-12	Z2-13	72-14	72.15	72-16	72-17	73-18	73 10	06 62	79-21	77-77	77-73	AC 07	77.75	77.26	72-27		73-1	73-2	Z3-3	Z3.4	Z3-8	Z3-9	73-10	Z3-11	Z3-12		Z3-13	Z3-14	Z3-15		HA81	HAB2	HA83

Prepared By RSM Checked By Dvd 2

Page 2 of 3

Data
Historical
1995
- -
TABLE

1

								-			
							1				-0114
Laboratory		Batch	Date of	q	H	BTEX	PAH	Benzo(a)	OC/UPS	PUBS	VINCE
			Sampling	теле	mg/kg	тg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg
	•	B	Guideline/HIL 'D'	1200	1000		80	4			**
Δmdel		9503388			QN	Q	1	1	1	I	1
Amdel		9503299		:	Q	QN	1	1	1	I	1
5010								ĺ			

NOTES (1) ND denotes not detected (2) ¹² denotes not analysed (3) <LOR denotes less than laboratory limit of reporting

,**х** .

Prepared By RSM Checked By Dod 2

Page 3 of 3

s:projects\51072\001\Reports\Tables\Historical1995_1998.xts\Table 1\19-09-02

TABLE 2 - 1998 Historical Data

34. 1

	,	. <u> </u>		ł	<u>,</u>	<u> </u>						-				7		٦
Phenols	mg/kg		Q		2		Q		N	Q			Q	CZ			CN	
Sulphate	mg/kg	2000	1140	110	211	205	32	2	17	11	24	17	32	27	5	44	<u>т</u>	2
Cyanide	mg/kg	1000	QZ			DN	QN		NN	QN		N	Q			QN	CN	
VHCs	mg/kg				1	1	1		1	1		ł	I		!	ł		I
PCBs	mg/kg		1			ł			1	1		1	1		1	I		
oc/oPs	mg/kg				1				l			l			I	l		l
Benzo(a)	mg/kg	4			1.3	3.3			<lor< td=""><td><1 OR</td><td></td><td>0.5</td><td>1.3</td><td></td><td><lok< td=""><td><lor< td=""><td></td><td>SLUK</td></lor<></td></lok<></td></lor<>	<1 OR		0.5	1.3		<lok< td=""><td><lor< td=""><td></td><td>SLUK</td></lor<></td></lok<>	<lor< td=""><td></td><td>SLUK</td></lor<>		SLUK
PAH	ma/ka	80	12 85	2	16.45	40.05			7.5	aO IV		8.85	212		⊲LOR	6.3	1	<pre><fok< pre=""></fok<></pre>
BTEX	ma/ka	0			1				1		1	1	,		1			1
HdT	ma/ka	1000		1	1	1		1	1		I	1			;	1		1
4 d	malkn	Eugn	202	5	105	175		84	140		3	159	106	2021	26	232	1	4
Date of	Semuling	Rundingo		Z-00-20	2-06-98	2-06-9R	22.00-7	30-06-98	30-06-98		02-10-CI	15-07-98	15 07 00	06-10-01	15-07-98	8 07 08		15-07-98
Date:		- ilobin C	Contract of the second	ESTUB9/ 2-00-30	ES10697	ES10607		ES11195 30-06-98	ES11195		ES1135/ 15-0/-38	ES11357 15-07-98			ES11357	EC11204	501101	ES11357 15-07-98
1	Depth Laboratory		:	ALS	ALS	ST S		ALS	U IA	2	ALS	N N		ALO	ALS		ALO	ALS
Sample	nebru			0-0.15	0-0.15		<u>.</u>	0-0.15	1		0-0.15	0.0.15	2.0	0-0.15	0-0 15		0-0-10	0-0.15
				RBD001	RRD002		KBDUU3	T3006-S1	TOODE CO	22-0002	ZDI	1002		VD4	ND6		KBDUU/	VD-Ramp

NOTES

(1) ND denotes not detected
(2) '-' denotes not analysed
(3) <LOR denotes less than laboratory limit of reporting

s:projects\51072\001\Reports\Tables\Historical1995_1998.xls\Table 2\19-09-02

Prepared By RSM Checked By Dod 2

TABLE 3 - 2002 Residual Inorganic Anarytical Data (concentrations in mg/kg)

		1		Cloned	roor L	1 2.1 51 TP26 1 3-1 51 TP14 1 3-1.5	TP14 13-15	DU-U1	1 PZ5 0.5-0.7	1220 U.S-0.71 1720 U.S-U.S1 171 0.S-U.S1 171 0.S-U.S	10.0-0.0 41 JI 1	01 L L 0 01 01 01 01 01	201 02
	Lean	Health Investigation Levels	CVCIS			CC34602 0	EC3/682.0	ES34682-0	FS34735-0	ES34735-0	ES34735-0	ES34735-0	ES34736-0
				Batch Number	2 2	ESOHUG-U							
	VE NH MOUN	3. TH WEAN OF THE WARN 'N' THE WARN	JIH MAEN	LOR									0
Ì			000		-		2	-	-	-	_ AQ1	<lor< td=""><td><luk< td=""></luk<></td></lor<>	<luk< td=""></luk<>
Arsenic	100	400	7007	-				2017		A OR	< OR	< OR	<10R
Cadmium	20	80	40		ALUR A								,
	100	007	000	*	~	en	2	~		Z	7	c	7
Chromium	201	201	7/1				c	C	03	σ	ų		<
	1000	4000	2000	•	<u>75</u>	71	0	D	50				
			000		6	T	F	~	2	2		7	Z
Nickel	600	2400	000	-	v				, cr		00	20	•
	0000	1200	600	•	113	164	26	6 0	27	10	60	50	3
Lead	200	202	33			C		20	36	÷	15		ŝ
Zinc	7000	28000	14000		33	20	Q	20	3				
		-				100					aC V	A C ∆	∆ 0R
Mercurv	15	09	30	0.1	0.1		4LUK	SLUR			100-		

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) N/A denotes not analysed

s:projects\51072\001\Reports\Tables\Residual data-a.xls\Table 3\19-09-02

TAP

Prepared By TAP Checked by Tods

TABLE 3 - 2002 Residual Inorganic Analytical Data (concentrations in mg/kg)

Autombure E334736-0 ES34736-0 ES34756-0 ES34736-0 E334736-0 E334736-0 <t< th=""><th></th><th>L la solication</th><th>0,00</th><th>Comple D</th><th>TP15 03-05</th><th></th><th>PUPU4</th><th>1 PZU U.3-U.31</th><th>10.0-0.0 2271</th><th>N.1-0-1 22-11</th><th>1720 U.S-U. B.S. 1. C.J. 2271 C.J. 2271 C.J. 2271 C.J. 201</th><th>2010</th><th>5 - 5 - 1</th></t<>		L la solication	0,00	Comple D	TP15 03-05		PUPU4	1 PZU U.3-U.31	10.0-0.0 2271	N.1-0-1 22-11	1720 U.S-U. B.S. 1. C.J. 2271 C.J. 2271 C.J. 2271 C.J. 201	2010	5 - 5 - 1
Batch Number Example	Lean	n investigation	Tevela			0 3047001	1	EC34736_0	FS34736-0	ES34736-0	ES34736-0	EB47571-0	ES34741-0
L M NEPM HIL TO NEPM HIL TO LOR 1 CLOR 200 1 CLOR 200 1 CLOR CLOR CLOR CLOR 1 CLOR 1 CLOR 1 CLOR 1 CLOR 200 1 CLOR CLOR CLOR CLOR CLOR 1 CLOR CLOR CLOR 1 CLOR CLOR CLOR CLOR CLOR CLOR CLOR CLOR				Batch Number	D-05/4501	0.00/#000							
1 <1OR 2 <1OR <1OR<	A' IIH MG:	CI, IIH MdHN	NEPM HIL'E									1	
400 200 1 < < < < < < < < < < < < < < < < < < < < < < < < < < < < < < < < < < < < < < < <					80 IV	6	<1 OR	<lor< td=""><td>ALOR</td><td>LOR</td><td>COR</td><td>~</td><td>7</td></lor<>	ALOR	LOR	COR	~	7
80 40 1 <luk< th=""> <luk< <="" td=""><td>100</td><td>400</td><td>2 NDZ</td><td>-</td><td></td><td></td><td></td><td></td><td></td><td></td><td>aC ⊽</td><td>∆ NOR</td><td><10R</td></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<></luk<>	100	400	2 NDZ	-							aC ⊽	∆ NOR	<10R
400 200 1 1 1 4 1 1 1 2 400 2000 1 5 <1OR	20	BD	40		LOR	¥0,↓	<pre>CUK</pre>						
400 200 1 5 <1OR 4 <1OR 3 <1OR 10 1 2400 500 1 <	3		000		-	4	- -	¢LOR	2	4 -2	-	2	n
4000 2000 1 3 400 200 1 30 2 400 2 400 2 400 2 400 2 400 2 400 1 30 2 1 400 2 400 2 400 1 2 400 2 400 2 400 2 400 2 400 2 1 2 2 400 2 1 2 2 400 4 </td <td>DDL</td> <td>400</td> <td>700</td> <td></td> <td>- 1</td> <td></td> <td></td> <td></td> <td>e</td> <td><10R</td> <td><10R</td> <td></td> <td>85</td>	DDL	400	700		- 1				e	<10R	<10R		85
2400 600 1 <lock< th=""> 2 1 <lock< th=""> 1 <lock< th=""> 1 2 <lock< th=""> 1 2 100 2 100 2 100 2 100 2 1 2 2 2 2 2 1 2 1 2 2 2 2 2 2 1 2 1 2 2 1 2 2 1 2 2 1 2 2 1 2 2 1 2 2 1 2 2 1 2 2 1 2 2 1 2 2 1 2 2 1 2 2 1 2 2 1 2 2 1 1 2 2 1 2 2 1 1 2 2 1 4 1 2 1 4 1 4 1 4 <th1< th=""> 4 <th1< th=""> 4</th1<></th1<></lock<></lock<></lock<></lock<>	1000	4000	2000	-	0		*		,			6	6
-2400 600 1 30 5 28 2 50 2 1 2 1200 600 1 23 1 16 <0R	000	00%6	800		A NO I∧	2		<lor< td=""><td>_</td><td>7</td><td>YLUTA</td><td>4</td><td></td></lor<>	_	7	YLUTA	4	
1200 600 1 30 3 1 16 <10R 15 <10R <10R 4 28000 14000 1 23 1 16 <10R	000	24UU	000			4	28	6	50	2	4	2	88
28000 14000 1 23 1 16 4.0K 15 4.0K 200 7 200 200 200 200 200 200 200 200 2	000	1200	600	-	20	2	24						218
	0002	00000	11000	-	23	* *	10	<pre>LOR</pre>	15	<lur< td=""><td>FOT.</td><td>+</td><td>717</td></lur<>	FOT.	+	717
20 20 20 20 20 20 20 20 20 20 20 20 20 2	1000	2000											
					_								
	ų,	80	30	0.1	≤LOR	<10R	<lor< td=""><td><lor< td=""><td><lor< td=""><td><lok< td=""><td><pre>%</pre></td><td>4LUR</td><td></td></lok<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><lok< td=""><td><pre>%</pre></td><td>4LUR</td><td></td></lok<></td></lor<></td></lor<>	<lor< td=""><td><lok< td=""><td><pre>%</pre></td><td>4LUR</td><td></td></lok<></td></lor<>	<lok< td=""><td><pre>%</pre></td><td>4LUR</td><td></td></lok<>	<pre>%</pre>	4LUR	

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) N/A denotes not analysed

s:projects\51072\001\Reports\Tables\Residual data-a.xts\Table 3\19-09-02

Prepared By <u>TAP</u> Checked by <u>Tod 2</u>

TABLE 3 - 2002 Residual Inorganic Analytical Data (concentrations in mg/kg)

1

	- 1		Cl olumes	TDU2 0 2-0 4	TP04 0 3-0 5	0 2-0 4 TP04 0 3-0 5 TP04 1 3-1 5	DUP06	TP5 1.3-1.5	DUP10	TP12_0.3-0.5 TP8_0.3-0.5		1.3-1.5
nean	Health Investigation Levels	revels					I.	0.000	LC21714 0	D FATAGO	EC3/761_0	EC34761_0
			Batch Number	r ES34741-0	ES34741-0	ES34741-0	ES34741-0	ES34/41-0	1-1+1+001	C004/41-0	D-10/#000	2-12-12-21
,⊽, IIH I	A. IIH WEEN A. IIH WEEN A. IIH WEEN	HIH MdHN	LOR									
				6	I OR	<10R	<1 OR	<lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
001	400	700		4					40 I 2	<1 OR	A OR	<lor< td=""></lor<>
20	80	40		ALOK	227							
		200	-	<u> </u>		AOR	<10R	ALOR	<lor< th=""></lor<>	1	< LOR	<10K
	000				4	<10R	<10R	⊲LOR	<lor< td=""><td>7</td><td>-</td><td><lor< td=""></lor<></td></lor<>	7	-	<lor< td=""></lor<>
nnn	4000							aC IV	A OR	<1 OR	<lor <<="" td=""><td><10R</td></lor>	<10R
600	2400	600		N	7						4	
	1200	800	-	第二日80回来		≜LOR	<10R		2	44	3	< LOK
200	00000					2012	I OR	<0R	<10R	45	9	<lor< td=""></lor<>
000	28000	14000	-	222	8							
							_					
	00	UC.	6	AL DR	I OR	<0R	<lor< td=""><td><lor< td=""><td><lor< td=""><td>LOR</td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td>LOR</td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td>LOR</td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	LOR	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
2		3										

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) N/A denotes not analysed

s:projects\51072\001\Reports\Tables\Residual data-a.xls\Table 3\19-09-02

Prepared By TAP Checked by Dad2

s:projects\51072\001\Reports\Tables\Residual data-a.xls\Table 3\19-09-02

TABLE 3 - 2002 Residual Inorganic Analytical Data (concentrations in mg/kg)

×	1 Health	Health Investigation Levels	evels	Sample ID	DUP12	DUP13	DUP14
				Batch Number	ES34761-0	ES34761-0	ES34761-0
	NEPM HIL 'A'	1	JIH WEEN J. D. NEPM HIL	LOR			
Arsenic	100		200	-	<lor< td=""><td><lor< td=""><td>⊲LOR</td></lor<></td></lor<>	<lor< td=""><td>⊲LOR</td></lor<>	⊲LOR
Cadminm	20	80	40	-	<lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
Chromitim	101	400	200	-	<lor< td=""><td>⊲LOR</td><td>⊲LOR</td></lor<>	⊲LOR	⊲LOR
Conner	1000	4000	2000	1	<1.0R	<lor< td=""><td>⊲LOR</td></lor<>	⊲LOR
Nickel	POD -	2400	600	-	<lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
	300	1200	600	-	2	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
7inc	7000	28000	14000	+	4	1	<lor< td=""></lor<>
Mercuin	15	60	30	0.1	<lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
							Ĩ

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) N/A denotes not analysed

Prepared By <u>TAP</u> Checked by <u>Pvd 2</u>

TP18 0.3-0.5	ES34736-1		AN		AN	NA	2170	AN I	aC ⊳			NIA		NIA		AN	AN		NA	N/A		
TP26 2 9-3.0 TP14 0.3-0.5 TP14 2.4-2.6 TP18 0.3-0.5	FS34735-1		<10R		159	611			770				202	A OR		<luk< td=""><td><10R</td><td></td><td><pre><pre>COK</pre></pre></td><td></td><td></td><td></td></luk<>	<10R		<pre><pre>COK</pre></pre>			
TP14 0.3-0.5	ES34735-1		A OR		<lor< td=""><td>A OR</td><td></td><td>V CK</td><td></td><td></td><td></td><td></td><td></td><td>¢ OR</td><td></td><td><lor< td=""><td>IOR</td><td></td><td>LOR</td><td></td><td></td><td></td></lor<></td></lor<>	A OR		V CK						¢ OR		<lor< td=""><td>IOR</td><td></td><td>LOR</td><td></td><td></td><td></td></lor<>	IOR		LOR			
TP26 2.9-3.0	EC34735.1		AC IV		259	604		<10R	053	202			404	<1 OR		<lor< td=""><td>21 OR</td><td>107</td><td>4.0R</td><td></td><td>21012</td><td></td></lor<>	21 OR	107	4.0R		21012	
TP26 0.3-0.5	EC21725.1			ίΩ,	LOR	160	3	AOA	005	601			_ AOR		107	<10R	100	1077	LOR		<pre><pok< pre=""></pok<></pre>	
TP25 0.5-0.7	LO0170E 1	E034/30-1			^LOR	210	210	132		448			ALOR			<1.0R			<10R			
DI 101	1	E034082-1		< LUK	116	602	200	134		833			<lor< td=""><td></td><td></td><td>< OR</td><td></td><td>PLOX</td><td><10R</td><td></td><td><lor< td=""><td></td></lor<></td></lor<>			< OR		PLOX	<10R		<lor< td=""><td></td></lor<>	
TP14 1 3-1 5		ES34682-1		- Ct	147	1.1	100	<1 OR	107	643			< OR		227	<1 OR		¥01>	<1 OR		⊲LOR	
TONE 1 2 4 6) TONE 1 2 4 5 TD14 1 2 4 5		ES34682-1		<lor <</lor 	113	2	504	162	102	839			< OR		z2_∧	40 i>		<10R			<lor< td=""><td></td></lor<>	
TOPE 1 2 1 E	11-6-1 07-1-0	ES34682-1		4.0R	954	501	1110	263	200	就能91213的X					~ √ √	001		<pre>>LOR</pre>			<lor< td=""><td></td></lor<>	
	Sample IU	Batch Number	LUK	ر	1	nc	100	007	DUL					7"П	C U		7.N	00		7.0	0.2	
	Guideline			65	3					1000	200											
						C10 - C14 Fraction	C15 C38 Eraction		C29 - C36 Fraction	101018 Tabl				Benzene	Tolvoro		Chlorobenzene	Tethulbootopo		Imeta- & para-Xvlene	ortho-Yvlone	

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) N/A denotes not analysed

s:projects\51072\001\Reports\Tables\Residual data-a.xls\Table 4\19-09-02



•

DUP03	EB47571-1				LOR	A OR		< LOK	ALOR				101	<pre> 4 0 R </pre>	00		▲LOR			<pre>~LOR</pre>	
TP21_0.3-0.5	ES34736-1			¥Ŋ7	ALOR	A OR	1	<10R		107		100	207	^LOR			_ LOR			<u>∧</u> 08	
TP24_0.3-0.5 TP22_0.3-0.5 TP22_1.3-1.5 TP23_1.3-1.5 TP21_0.3-0.5	ES34736-1				<lor< td=""><td>a012</td><td></td><td>AOR</td><td></td><td>407</td><td></td><td></td><td>SLUK</td><td><10R</td><td></td><td>4LUR</td><td>^_OR</td><td></td><td></td><td>a V V</td><td></td></lor<>	a012		AOR		407			SLUK	<10R		4LUR	^_OR			a V V	
TP22_1.3-1.5	ES34736-1			Å Å	<lor< td=""><td></td><td>107</td><td><lor< td=""></lor<></td><td>207</td><td>20.12</td><td>-</td><td></td><td><pre>FOR</pre></td><td>A OR</td><td></td><td><pre>FUR</pre></td><td>I OR</td><td></td><td>¥07∨</td><td>aC V</td><td></td></lor<>		107	<lor< td=""></lor<>	207	20.12	-		<pre>FOR</pre>	A OR		<pre>FUR</pre>	I OR		¥07∨	aC V	
TP22_0.3-0.5	ES34736-1			A QR	<1 OR			A OR	C C	40K			<10K	AO IN		A OR	NOR NOR		V COX		
TP24_0.3-0.5	FS34736-1			_ AOR	aC V		PCCR	<1 OR		<lor< td=""></lor<>			LOR			<10R			<lor< td=""></lor<>		2077
DUP04	ES34736-1			A OR	207		¥07	A OB		<pre> 4 0 R </pre>			<10R					577	<10R		PLOR
5 0.3-0.5 TP15 1.3-1.5	E-SAT26-1			<1 OR			▲LOR		107	<pre> COR </pre>			A OR		2012	A OR		LCT LCT	I OR		<luk< td=""></luk<>
TP15 0.3-0.5	L 001706.1	1-00/+000					ALOR NO.1	001		A OR			aC IV		ALOR	<1 OR		YOT V	AC IV	10-1	<lor< td=""></lor<>
CU IPUS	10001	E034/30-1				SLUK	A OR			A DR				-FOI	AOA			VLO1		57	<lor< td=""></lor<>
Samila ID		Batch Number	LOR	c	7	50	100	001	100					0.2	0.2	¢ U	7'N	0.2	6.0	7-0	0.2
O. dollar					3					1000	0001							_			
	3				C6 - C9 Fraction	C10 - C14 Fraction		C12-C28 FIACIUL	IC79 - C36 Fraction	040 000 T-1-1	C10-C30 10131			Benzene	Tohrene		Chlorobenzene	Ethylhenzene		meta- & para-Xylene	artho-Xviene

NOTES (1) <LOR denotes less than the taboratory limit of reporting (2) N/A denotes not analysed





BH8 3.7-3.9	ES35296-0			20	<pre>LOR</pre>	40 IV		SLUR	aC IV			4OH	an DR		<10R	A OR		< LOR			
DUP14	ES34761-1			<lor< td=""><td>_ LOR</td><td></td><td></td><td>4LOR</td><td></td><td></td><td></td><td><lor< td=""><td>aC V</td><td></td><td><lor< td=""><td>A OR</td><td></td><td>_ ∧LOR</td><td></td><td></td><td></td></lor<></td></lor<></td></lor<>	_ LOR			4LOR				<lor< td=""><td>aC V</td><td></td><td><lor< td=""><td>A OR</td><td></td><td>_ ∧LOR</td><td></td><td></td><td></td></lor<></td></lor<>	aC V		<lor< td=""><td>A OR</td><td></td><td>_ ∧LOR</td><td></td><td></td><td></td></lor<>	A OR		_ ∧LOR			
* DUP13	ES34761-1				< OR			AOR				<lor< td=""><td>aur</td><td></td><td><lor< td=""><td></td><td></td><td> ≜CR</td><td></td><td></td><td>1</td></lor<></td></lor<>	aur		<lor< td=""><td></td><td></td><td> ≜CR</td><td></td><td></td><td>1</td></lor<>			 ≜CR			1
TP8 1.3-1.5	ES34761-1			<lor< td=""><td>< OR</td><td></td><td>PLOTA</td><td>A OR</td><td></td><td></td><td></td><td>A.OR</td><td></td><td>YTOY.</td><td><!--<br-->COR</td><td></td><td></td><td>^LOR</td><td>001</td><td>4LUK</td><td></td></lor<>	< OR		PLOTA	A OR				A.OR		YTOY.	<br COR			^LOR	001	4LUK	
TP6 1.3-1.5	FS34741-1			4LOR	AO A		<luk< td=""><td>^LOR</td><td></td><td>FUR</td><td></td><td>AOR</td><td>100</td><td>207</td><td><lor< td=""><td>100</td><td>ALCA</td><td>< OR</td><td></td><td>SLUK</td><td></td></lor<></td></luk<>	^LOR		FUR		AOR	100	207	<lor< td=""><td>100</td><td>ALCA</td><td>< OR</td><td></td><td>SLUK</td><td></td></lor<>	100	ALCA	< OR		SLUK	
DUP07	EC34741-1			ALOR	AO IV		<lor< td=""><td><lor< td=""><td></td><td><10K</td><td></td><td>4.0R</td><td></td><td>FLOR</td><td><lor< td=""><td></td><td>SLUK</td><td><1 OR</td><td></td><td>< LOK</td><td></td></lor<></td></lor<></td></lor<>	<lor< td=""><td></td><td><10K</td><td></td><td>4.0R</td><td></td><td>FLOR</td><td><lor< td=""><td></td><td>SLUK</td><td><1 OR</td><td></td><td>< LOK</td><td></td></lor<></td></lor<>		<10K		4.0R		FLOR	<lor< td=""><td></td><td>SLUK</td><td><1 OR</td><td></td><td>< LOK</td><td></td></lor<>		SLUK	<1 OR		< LOK	
DUP05	EC24741-1			<1 OR			<10R	A OR		€UK		<lor< td=""><td></td><td>- Ct</td><td>≤ OR</td><td></td><td>ALUK</td><td>20 IV</td><td></td><td><10R</td><td></td></lor<>		- Ct	≤ OR		ALUK	20 IV		<10R	
TP00 13-15	EC24741-1			< OR		2017	^ LOR	I OR		£Q4		<10R		₹ CCX	<1 OR	107	ALOK	aO I>		^ LOR	
TEN1 1 3 1 E TEN3 0 3 0 51	2 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	1-1+1+000		<1 OR			<lor< td=""></lor<>			<lor< td=""></lor<>		<1 OR		<lor< td=""><td>2012</td><td>57</td><td></td><td>100</td><td></td><td><1.0R</td><td></td></lor<>	2012	57		100		<1.0R	
TD01 1215		ーーナンカクロ		00 IZ		<luk< td=""><td>A SQA</td><td>2</td><td></td><td>LOR</td><td></td><td>AC I></td><td>101</td><td>A OR</td><td>100</td><td>57</td><td>A OR</td><td></td><td></td><td><10R</td><td></td></luk<>	A SQA	2		LOR		AC I>	101	A OR	100	57	A OR			<10R	
Cimila ID		Batch Number	1 OR		7	50	100		100			60	7.0	0.2	100	7-7	C U		7-N	0.2	1
	Culdeline				3					1000									_		
				:	Co - Ca Fraction	C10 - C14 Fraction	C15_C28 Eraction		C29 - C36 Fraction	C10.C36 Total			Benzene	Toluono		Chlorobenzene	Ethickhon 200		Imeta- & para-Xylene	ortho Virlano	oluio-Ayieric

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) N/A denotes not analysed

s:projects\51072\001\Reports\Tables\Residual data-a.xls\Table 4\19-09-02



			7 7 7 7 7 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	1 000 Ma	BU2 10-10	RH3 30-32	DUP06	BH20 2.0-2.2 BH10 1.0-1.2	BH10 1.0-1.2	🕷 DUP12	BH11_1.0-1.2	BH12_2.2-2.3
	Guideline			E					Looroor	LCOCOUD 0	ECSESOE D	EC35306-0
		Ratch Number	ES35296-0	ES35296-0	ES35296-0	ES35296-0	ES35296-0	ES35296-0	E030290-U	E333280-U	D-DAJCOCI	
										-		
		22							çç	000	001	06~
CB - CO Emortion	65	~	<20	27	₹	8	<20	<20	NZ2	220		3
	3			<10R	<0 b	4 OR	≤LOR	305	196	193	125	13/
C10 - C14 Fraction							AI OR	824	620	596	582	328
C15 - C28 Fraction		100		107						100	9012	AC A
C36 Eraction		100	<10R	LOR	A No1∧	VLOK	<pre>COX</pre>	ALUR				107
	1001		2012	I OR	4 OR	<lor< td=""><td><lor< td=""><td>状态于1129年的</td><td>816</td><td>789</td><td>707</td><td>400</td></lor<></td></lor<>	<lor< td=""><td>状态于1129年的</td><td>816</td><td>789</td><td>707</td><td>400</td></lor<>	状态于1129年的	816	789	707	400
C10-C30 10(3)	1000											
										a0 1	AL DR	A OR
Borrene		60	<lor< td=""><td><pre> COR </pre></td><td><pre> COR </pre></td><td>¥07₹</td><td>SLUK</td><td></td><td>207</td><td></td><td></td><td></td></lor<>	<pre> COR </pre>	<pre> COR </pre>	¥07₹	SLUK		207			
		100		<10P	A OR	<10R	≤LOR	<lor< td=""><td>A OR</td><td><lor< td=""><td><10R</td><td><10R</td></lor<></td></lor<>	A OR	<lor< td=""><td><10R</td><td><10R</td></lor<>	<10R	<10R
loluene		7.0					2012	A OR	<1 OR	4.0R	<lor< td=""><td></td></lor<>	
Chlorobenzene		0.2		207								
Cthuthon 200		0.0	A OR	<10R	<lor< td=""><td><lor< td=""><td><2 CK</td><td>¥07</td><td></td><td>2077</td><td></td><td></td></lor<></td></lor<>	<lor< td=""><td><2 CK</td><td>¥07</td><td></td><td>2077</td><td></td><td></td></lor<>	<2 CK	¥07		2077		
	-	100			41 OR	<ior< td=""><td><1.0R</td><td><lor< td=""><td><lor< td=""><td><lor< td=""><td>4.0R</td><td>4OR</td></lor<></td></lor<></td></lor<></td></ior<>	<1.0R	<lor< td=""><td><lor< td=""><td><lor< td=""><td>4.0R</td><td>4OR</td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td>4.0R</td><td>4OR</td></lor<></td></lor<>	<lor< td=""><td>4.0R</td><td>4OR</td></lor<>	4.0R	4OR
Imeta- & para-Xylene		7.0		107							a O V	AC A
ntho-Xvlene		0.2	<lor <!--</td--><td><10R</td><td><10K</td><td><luk< td=""><td><pre><pre>LUK</pre></pre></td><td></td><td></td><td></td><td></td><td></td></luk<></td></lor 	<10R	<10K	<luk< td=""><td><pre><pre>LUK</pre></pre></td><td></td><td></td><td></td><td></td><td></td></luk<>	<pre><pre>LUK</pre></pre>					

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) N/A denotes not analysed

s:projects\51072\001\Reports\Tables\Residual data-a.xls\Table 4\19-09-02



BH16_1.U-1.2	ES35296-0		740	<10R	578	217	795		<lor< th=""><th><lor< th=""><th>⊲LOR</th><th><lor< th=""><th><lor< th=""><th><!-- OR</th--><th></th></th></lor<></th></lor<></th></lor<></th></lor<>	<lor< th=""><th>⊲LOR</th><th><lor< th=""><th><lor< th=""><th><!-- OR</th--><th></th></th></lor<></th></lor<></th></lor<>	⊲LOR	<lor< th=""><th><lor< th=""><th><!-- OR</th--><th></th></th></lor<></th></lor<>	<lor< th=""><th><!-- OR</th--><th></th></th></lor<>	OR</th <th></th>	
BH14_2.0-2.2 BH15_1.3-1.5 BH16_1.0-1.2	ES35296-0		2	110	906	304	《王文1320 》 。		<lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""><td>aC IV</td><td>-</td></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""><td>aC IV</td><td>-</td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><lor< td=""><td>aC IV</td><td>-</td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td>aC IV</td><td>-</td></lor<></td></lor<>	<lor< td=""><td>aC IV</td><td>-</td></lor<>	aC IV	-
BH14_2.0-2.2	ES35296-0	00	220	414	1360	לΩR	JZ661774-52	1	<lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""><td></td><td></td></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""><td></td><td></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><lor< td=""><td></td><td></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td></td><td></td></lor<></td></lor<>	<lor< td=""><td></td><td></td></lor<>		
DUP10	ES35296-0		072	1800	3820	213	STOREPROPERTY STORE	ているがいわりためを見てい	<lor< td=""><td><lor< td=""><td><lor< td=""><td><1.0R</td><td><lor< td=""><td></td><td></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><1.0R</td><td><lor< td=""><td></td><td></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><1.0R</td><td><lor< td=""><td></td><td></td></lor<></td></lor<>	<1.0R	<lor< td=""><td></td><td></td></lor<>		
DUP09	ES35296-0		2Z0	930	2460	134	Statistics FOA Statistics		AOA	<lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""><td></td><td></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><lor< td=""><td></td><td></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td></td><td></td></lor<></td></lor<>	<lor< td=""><td></td><td></td></lor<>		
BH2 2.5-2.7			<20	1170	2810	154	の実施化している場合		<lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""><td>< A OR</td><td></td><td></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><lor< td=""><td>< A OR</td><td></td><td></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td>< A OR</td><td></td><td></td></lor<></td></lor<>	<lor< td=""><td>< A OR</td><td></td><td></td></lor<>	< A OR		
BH9 2.0-2.2			7 30	95	350	<10R	AAE	ł	<lor< td=""><td><lor< td=""><td><10R</td><td>≤ 0R</td><td>I OR</td><td></td><td></td></lor<></td></lor<>	<lor< td=""><td><10R</td><td>≤ 0R</td><td>I OR</td><td></td><td></td></lor<>	<10R	≤ 0R	I OR		
BH13 0.8-1.3	FS35296-0		² 3	<lor< td=""><td>130</td><td>< 0R</td><td>064</td><td>net</td><td><lor< td=""><td><10R</td><td><1 OR</td><td>< 0R</td><td><10R</td><td></td><td>-LOK</td></lor<></td></lor<>	130	< 0R	064	net	<lor< td=""><td><10R</td><td><1 OR</td><td>< 0R</td><td><10R</td><td></td><td>-LOK</td></lor<>	<10R	<1 OR	< 0R	<10R		-LOK
RH21 1 7 BH22 2.5-2.7 BH13 0.8-1.3	FS35296-0		~ 30	140	502	<1 OR		240	<lor< td=""><td>I OR</td><td><1 OR</td><td><1 OR</td><td></td><td></td><td><pre></pre></td></lor<>	I OR	<1 OR	<1 OR			<pre></pre>
BH21 1.7	EC35206-0		52	<1.0R	< OR	4 OR		4 UK	A O.R	N DR					<lor< td=""></lor<>
Samula ID	Batch Number	LUK	2	50	191	100	201		0.2	60	0.0	2.0	7.0	7.0	0.2
Guidalino			65					1000							
	A		IC6 - C9 Fraction	C10 - C14 Fraction				C10-C36 Total		Tohono	Chlorohomono			Inela- & para-Ayene	ortho-Xylene

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) N/A denotes not analysed

Prepared By TM Checked by Dods

BH4 3234 I BH23 1.3-1.5I BH24 0.8-1.0 TP100 2.0-2.21 TP101 3.0-3.2 I P102 1.0-1.2 TP103 3.5-3.7	ES35360-0				ALOR	ALOR	-ALOR			N/A	P10	AN N	NA	N/A		NIA	N/A	1171	
TP102 1.0-1.2	ES35360-0				<lor< td=""><td><lor< td=""></lor<></td><td>⊲LOR</td><td></td><td></td><td>N/A</td><td></td><td>N/A</td><td>N/A</td><td>N/A</td><td></td><td>NA</td><td>N/A</td><td></td><td></td></lor<>	<lor< td=""></lor<>	⊲LOR			N/A		N/A	N/A	N/A		NA	N/A		
TP101 3.0-3.2	ES35360-0				<lor< td=""><td><10R</td><td><lor td="" <=""><td>001</td><td>VEUR</td><td>NIA</td><td></td><td>N/A</td><td>N/A</td><td>N/A</td><td></td><td>N/A</td><td>NIA</td><td></td><td></td></lor></td></lor<>	<10R	<lor td="" <=""><td>001</td><td>VEUR</td><td>NIA</td><td></td><td>N/A</td><td>N/A</td><td>N/A</td><td></td><td>N/A</td><td>NIA</td><td></td><td></td></lor>	001	VEUR	NIA		N/A	N/A	N/A		N/A	NIA		
TP100 2.0-2.2	ES35360-0			<pre>LUK</pre>	_ LOR	⊲LOR	<1.0R		4CTA	 N/A		NA	NIA	NIA	274	N/A	NUA	Y MI	
BH24 0.8-1.0	FS35293-0			<10K	94	512	I OR		909	< OR		A.OR	<lor< td=""><td></td><td>107</td><td><lor< td=""><td></td><td>2012</td><td></td></lor<></td></lor<>		107	<lor< td=""><td></td><td>2012</td><td></td></lor<>		2012	
BH23 1.3-1.5	FS35293-0			4 0R	4_OR	134	<1 OR		134	a0 12	107	<lor< td=""><td><10R</td><td></td><td></td><td><lor< td=""><td></td><td>4LUK</td><td></td></lor<></td></lor<>	<10R			<lor< td=""><td></td><td>4LUK</td><td></td></lor<>		4LUK	
BH4 3.2-3.4	ES5208-0			<10R	<10R	A OR			<lor< td=""><td></td><td>222</td><td><lor< td=""><td><1 OR</td><td></td><td>-CCK</td><td><1 OR</td><td></td><td><luk< td=""><td></td></luk<></td></lor<></td></lor<>		222	<lor< td=""><td><1 OR</td><td></td><td>-CCK</td><td><1 OR</td><td></td><td><luk< td=""><td></td></luk<></td></lor<>	<1 OR		-CCK	<1 OR		<luk< td=""><td></td></luk<>	
BH18 3.3-3.5	ECEDOR-D			5	4240	REOD	284	107	研究13227 並設		SLUR.	OR</td <td>A OR</td> <td></td> <td>- Story</td> <td><i or<="" td=""><td></td><td><lor< td=""><td></td></lor<></td></i></td>	A OR		- Story	<i or<="" td=""><td></td><td><lor< td=""><td></td></lor<></td></i>		<lor< td=""><td></td></lor<>	
DI IP07	EC36706.0	2000000		<20 ≺20	3550	6010	225		差碳10795%%		SLUR	<i or<="" td=""><td></td><td></td><td>¥07</td><td>ao iv</td><td></td><td><lor< td=""><td></td></lor<></td></i>			¥07	ao iv		<lor< td=""><td></td></lor<>	
RH5 1 8-2 0	LOSEDOE D	L0002000		<20	1550	2050	154	101	年間。4751萬世		¥07₹		aor	107	<10R		10-11-11-11-11-11-11-11-11-11-11-11-11-1	<lor< td=""><td></td></lor<>	
Comple ID	Odilipie ID	Batch Number	LOR	6	#U		100	100			0.2	0.2		7"0	0.2		7-0	0.2	
C. Malian	פתומפוונופ			65	3				1000										
	*			C6 - C0 Erantion			C15 - C28 Fraction	C29 - C36 Fraction	C10-C36 Total		Benzene		1 Olueite	Chiorobenzene	Ethylhanzana		meta- & para-Xylene	artho-Xvlene	

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) N/A denotes not analysed s:projects\51072\001\Reports\Tables\Residual data-a.xls\Table 4\19-09-02

Prepared By **F** AP Checked by **Dod 2**

6

4

		4	TTOADA 4 D 4 D		2 E 2 71 TE 10 E 3 0-3 21	DI IP103	TP106 3 0-3.2 TP107 3.5-3.7 TP108 3.5-3.7 TP109 3.0-3.2	FP107 3.5-3.7	TP108 3.5-3.7	TP109 3.0-3.2	DUP107	201-102
	Guideline	Sample IU	11/104 1.0-1.2 11/104		70 00 01 11			C COCOCUL	ECSESSO 0	EC25561_0	ES35561-0	FS35561-0
		Batch Number	ES35360-0	ES35360-0	ES35360-0	ES35360-0	E233300-0	ESSOUTU	0-5000000	E-COULD		
									-			
		LUK				401		100	<10R	< OR	<lor< td=""><td><pre>LOR</pre></td></lor<>	<pre>LOR</pre>
C6 - C0 Eraction	65	2	</td <td>₹ CCX</td> <td>ALOR</td> <td>ALUR.</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	₹ CCX	ALOR	ALUR.						
	3		100	N 20 10 B	<1 OR	^LOR	AΩR	<pre>COR</pre>	<10K	V CK		
C10 - C14 Fraction		00					A OR	< OR	<lor< td=""><td><lor </lor </td><td><pre>LOR</pre></td><td><0R</td></lor<>	<lor </lor 	<pre>LOR</pre>	<0R
C15 - C28 Fraction		100	<ru><td>SLUR</td><td>101</td><td>101</td><td></td><td></td><td></td><td><1 OR</td><td>≤ 0R</td><td>A OR</td></ru>	SLUR	101	101				<1 OR	≤ 0R	A OR
COO COC Lastion		100	v OR	ALOR ALOR	<pre>LOR</pre>	2012	FOY	207	157			
UZ3 - U30 FIACIUI		201			<1 OR	<1 OR	<10R	^ LOR	<lor< td=""><td>4.0R</td><td><10R</td><td>¥91</td></lor<>	4.0R	<10R	¥91
C10-C36 Total	1000		4LOK									
-												
		4	A LLA	NIN	NIA	N/A	NIA	N/A	AN	NIA	NA	AN
Benzene		0.2	AN N					NICA	NIV	NIA	N/A	N/A
		60	N/N	N/N	N/A	A/A	NA	AN	YN Y			
		4 0		NIA	NIA	N/A	NIA	NN	NA	NIA	N/A	NA
Chlorobenzene		0.2	AW				A1/A	NIA	NIA	NIA	N/A	AN
Cthurbon con		0.2	A/N	A/A	N/A	N/A	A/N		1.2			
Clinkingiccile			NITA	NIV	N/A	N/A	NIA	N/A	NA	NA	AN	¥2
meta- & para-Xylene		7.0				A17A	NIA	NIA	N/A	N/A	N/A	- NN
orthorXylene		0.2	NN	NIA	N/A	N/A	YN Y					

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) N/A denotes not analysed

s:projects\51072\001\Reports\Tables\Residual data-a.xls\Table 4\19-09-02

Prepared By TAP Checked by Dad2

÷,

1

> 1 1 1

.

.

	Guideline	Sample ID	TP110 3.0-3.2	TP111 2.0-2.2	TP112 3.0-3.2	TP110 3.0-3.2 TP111 2.0-2.2 TP112 3.0-3.2 TP113 2.0-2.2	_
		Batch Number	ES35561-0	ES35561-0	ES35561-0	ES35561-0	ES35561-0
		B					
C6 - C0 Eraction	65	2	_ LOR	<10R	<lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
C10 - C14 Fraction	8	20	<lor< td=""><td><1.0R</td><td><lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<>	<1.0R	<lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
C15 C38 Fraction		100	<10R	⊲LOR	<lor< td=""><td><lor< td=""><td>⊲.0R</td></lor<></td></lor<>	<lor< td=""><td>⊲.0R</td></lor<>	⊲.0R
		100	4.0R	<lor< td=""><td><lor< td=""><td><lor< td=""><td>₹LOR</td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td>₹LOR</td></lor<></td></lor<>	<lor< td=""><td>₹LOR</td></lor<>	₹LOR
C10-C36 Total	1000		40R	<lor< td=""><td><lor< td=""><td><10R</td><td><1.0R</td></lor<></td></lor<>	<lor< td=""><td><10R</td><td><1.0R</td></lor<>	<10R	<1.0R
Dentend		0.2	N/A	N/A	N/A	NIA	N/A
		0.5	N/A	N/A	NIA	NIA	NA
Chlorobenzene		4.0	NA	N/A	N/A	NIA	N/A
Cillor ouch izeria		0.2	N/A	NVA	NIA	NA	N/A
meta_ & nara-Yviene		0.2	AVA	N/A	N/A	NA	N/A
ndho.Yvlene		0.2	N/A	N/A	NIA	N/A	N/A

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) N/A denotes not analysed Prepared By TAP Checked by Dodz

rP15_1.3-1.5	ES34736-2			40 IV	5		< COX	ALOR	⊲LOR	<lor< th=""><th><lor <</lor </th><th></th><th></th><th>ALUK</th><th><lor< th=""><th><lor< th=""><th><10R</th><th></th><th></th><th></th><th><10R</th><th><lor< th=""><th><10R</th><th></th></lor<></th></lor<></th></lor<></th></lor<>	<lor <</lor 			ALUK	<lor< th=""><th><lor< th=""><th><10R</th><th></th><th></th><th></th><th><10R</th><th><lor< th=""><th><10R</th><th></th></lor<></th></lor<></th></lor<>	<lor< th=""><th><10R</th><th></th><th></th><th></th><th><10R</th><th><lor< th=""><th><10R</th><th></th></lor<></th></lor<>	<10R				<10R	<lor< th=""><th><10R</th><th></th></lor<>	<10R	
	ES34736-2					4LQK	<lor< td=""><td><10R</td><td><lor< td=""><td><lor< td=""><td>< LOR</td><td>aC IV</td><td></td><td>¥S</td><td><lor< td=""><td>^LOR</td><td>ALOR</td><td></td><td></td><td>¥07</td><td><pre> COR </pre></td><td>ALOR</td><td>AL IN</td><td></td></lor<></td></lor<></td></lor<></td></lor<>	<10R	<lor< td=""><td><lor< td=""><td>< LOR</td><td>aC IV</td><td></td><td>¥S</td><td><lor< td=""><td>^LOR</td><td>ALOR</td><td></td><td></td><td>¥07</td><td><pre> COR </pre></td><td>ALOR</td><td>AL IN</td><td></td></lor<></td></lor<></td></lor<>	<lor< td=""><td>< LOR</td><td>aC IV</td><td></td><td>¥S</td><td><lor< td=""><td>^LOR</td><td>ALOR</td><td></td><td></td><td>¥07</td><td><pre> COR </pre></td><td>ALOR</td><td>AL IN</td><td></td></lor<></td></lor<>	< LOR	aC IV		¥S	<lor< td=""><td>^LOR</td><td>ALOR</td><td></td><td></td><td>¥07</td><td><pre> COR </pre></td><td>ALOR</td><td>AL IN</td><td></td></lor<>	^LOR	ALOR			¥07	<pre> COR </pre>	ALOR	AL IN	
TP18_0.3-0.5	ES34735-2					<lor< td=""><td><lor< td=""><td><lor< td=""><td><1.0R</td><td><lor< td=""><td><_lOR</td><td>10</td><td></td><td>≤LOR</td><td><lor< td=""><td><lor< td=""><td>AL OR</td><td></td><td></td><td><lor< td=""><td><lor< td=""><td>ALOR</td><td></td><td>11/1</td></lor<></td></lor<></td></lor<></td></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><1.0R</td><td><lor< td=""><td><_lOR</td><td>10</td><td></td><td>≤LOR</td><td><lor< td=""><td><lor< td=""><td>AL OR</td><td></td><td></td><td><lor< td=""><td><lor< td=""><td>ALOR</td><td></td><td>11/1</td></lor<></td></lor<></td></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><1.0R</td><td><lor< td=""><td><_lOR</td><td>10</td><td></td><td>≤LOR</td><td><lor< td=""><td><lor< td=""><td>AL OR</td><td></td><td></td><td><lor< td=""><td><lor< td=""><td>ALOR</td><td></td><td>11/1</td></lor<></td></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	<1.0R	<lor< td=""><td><_lOR</td><td>10</td><td></td><td>≤LOR</td><td><lor< td=""><td><lor< td=""><td>AL OR</td><td></td><td></td><td><lor< td=""><td><lor< td=""><td>ALOR</td><td></td><td>11/1</td></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	<_lOR	10		≤LOR	<lor< td=""><td><lor< td=""><td>AL OR</td><td></td><td></td><td><lor< td=""><td><lor< td=""><td>ALOR</td><td></td><td>11/1</td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td>AL OR</td><td></td><td></td><td><lor< td=""><td><lor< td=""><td>ALOR</td><td></td><td>11/1</td></lor<></td></lor<></td></lor<>	AL OR			<lor< td=""><td><lor< td=""><td>ALOR</td><td></td><td>11/1</td></lor<></td></lor<>	<lor< td=""><td>ALOR</td><td></td><td>11/1</td></lor<>	ALOR		11/1
TP14_0.3-0.5	ES34735-2				POY -	<lor< td=""><td>ALOR</td><td><lor< td=""><td><lor< td=""><td>≤LOR</td><td><0R</td><td></td><td>5</td><td>ALOR R</td><td><lor< td=""><td>A OR</td><td></td><td></td><td></td><td>4_OR</td><td><lor< td=""><td><10R</td><td></td><td>177</td></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	ALOR	<lor< td=""><td><lor< td=""><td>≤LOR</td><td><0R</td><td></td><td>5</td><td>ALOR R</td><td><lor< td=""><td>A OR</td><td></td><td></td><td></td><td>4_OR</td><td><lor< td=""><td><10R</td><td></td><td>177</td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td>≤LOR</td><td><0R</td><td></td><td>5</td><td>ALOR R</td><td><lor< td=""><td>A OR</td><td></td><td></td><td></td><td>4_OR</td><td><lor< td=""><td><10R</td><td></td><td>177</td></lor<></td></lor<></td></lor<>	≤LOR	<0R		5	ALOR R	<lor< td=""><td>A OR</td><td></td><td></td><td></td><td>4_OR</td><td><lor< td=""><td><10R</td><td></td><td>177</td></lor<></td></lor<>	A OR				4_OR	<lor< td=""><td><10R</td><td></td><td>177</td></lor<>	<10R		177
IP26_0.3-0.5	ES34735-2			00.	4LOK	<10R	< LOR	<lor< td=""><td><lor< td=""><td><10R</td><td> N B B B B B B B B B B B B B B B B B B</td><td></td><td>YOY</td><td>4LOR</td><td><lor< td=""><td>AO I≥</td><td></td><td></td><td>¥ð</td><td><lor< td=""><td><lor< td=""><td>I OR</td><td></td><td>SLUK</td></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><10R</td><td> N B B B B B B B B B B B B B B B B B B</td><td></td><td>YOY</td><td>4LOR</td><td><lor< td=""><td>AO I≥</td><td></td><td></td><td>¥ð</td><td><lor< td=""><td><lor< td=""><td>I OR</td><td></td><td>SLUK</td></lor<></td></lor<></td></lor<></td></lor<>	<10R	 N B B B B B B B B B B B B B B B B B B		YOY	4LOR	<lor< td=""><td>AO I≥</td><td></td><td></td><td>¥ð</td><td><lor< td=""><td><lor< td=""><td>I OR</td><td></td><td>SLUK</td></lor<></td></lor<></td></lor<>	AO I≥			¥ð	<lor< td=""><td><lor< td=""><td>I OR</td><td></td><td>SLUK</td></lor<></td></lor<>	<lor< td=""><td>I OR</td><td></td><td>SLUK</td></lor<>	I OR		SLUK
TP25_0.5-0.7	ES34735-2				<pre> </pre>	40R	<10R	<10R	<10R	A DR				<lor< td=""></lor<>	40R				<lor< td=""><td>ALOR</td><td><10R</td><td></td><td></td><td>SLUK</td></lor<>	ALOR	<10R			SLUK
DUP01	ES34682-2	+-			⊲LOR	<lor< td=""><td><10R</td><td>A OR</td><td>A OR</td><td></td><td></td><td></td><td>¥07</td><td><lor< td=""><td>I OR</td><td></td><td></td><td>PLOT</td><td><lor< td=""><td><lor< td=""><td>I OR</td><td></td><td></td><td><10K</td></lor<></td></lor<></td></lor<></td></lor<>	<10R	A OR	A OR				¥07	<lor< td=""><td>I OR</td><td></td><td></td><td>PLOT</td><td><lor< td=""><td><lor< td=""><td>I OR</td><td></td><td></td><td><10K</td></lor<></td></lor<></td></lor<>	I OR			PLOT	<lor< td=""><td><lor< td=""><td>I OR</td><td></td><td></td><td><10K</td></lor<></td></lor<>	<lor< td=""><td>I OR</td><td></td><td></td><td><10K</td></lor<>	I OR			<10K
rP25 1 3-1 ftP26 1.3-1.ftP14 1.3-1.5	FS34682-2				LOR	<lor< td=""><td><lor< td=""><td>NO I∧</td><td></td><td></td><td></td><td></td><td><10R</td><td>4 OR</td><td><1 OR</td><td></td><td></td><td>× V</td><td>ALOR</td><td><10R</td><td>aor</td><td></td><td>2077</td><td><10R</td></lor<></td></lor<>	<lor< td=""><td>NO I∧</td><td></td><td></td><td></td><td></td><td><10R</td><td>4 OR</td><td><1 OR</td><td></td><td></td><td>× V</td><td>ALOR</td><td><10R</td><td>aor</td><td></td><td>2077</td><td><10R</td></lor<>	NO I∧					<10R	4 OR	<1 OR			× V	ALOR	<10R	aor		2077	<10R
fTP26 1.3-1.	FS34682-2				A ROA	4.0R	A OR					PLOK	0.0	A OR				<10K	<br COR	A OR				0.6
TP25 1 3-1		_			4.0R	IOR						¥07⊽	AOR	NO N			PLOK	<pre> 4 0 R </pre>	<10R	a0 Iv			<pre>VK</pre>	<lor< td=""></lor<>
Camula ID	Botch Numba			LOR	0.5	0.5				C'1	0,1	0.5	0.5	0.5	200	0.0	0.5	0.5	0.5				0.5	40
- Louole			NEPM																					
in the stand			NEPM	HL'Ú																				0 80
	Health		NEPM	HIL'A'																	0	<u> </u>		20
	,				Monthholomo		Acenaphunylene	Acenaphthene	Fluorene	Phenanthrene	Anthracene	Fluoranthene		ryrene	Benz(a)anthracene	Chrysene	Benzo(b)fluoranthene	Banzo(k)filoranthene		Benzo(a)pyrene	Indeno(1.2.3.cd)pyrene	Dibenz(a.h)anthracene	Benzh(n h iherviene	Total

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) N/A denotes not analysed

Prepared By <u></u> Checked by <u></u>

ES34741-2 ≜ NOR A A A ROA ROA <LOR Å R <1.0R ALOR ALOR NO 401 ES34741-2 0.6 0.6 0.6 0.6 <u>^</u>LOR 1.3 1.3 10.7 **≜**LOR <LOR</td> 0.6 0.6 4.9 DUP04 [IP20_0.3-0.4[IP23_1.3-1.4[IP21_0.3-0.4[P16_0.2-0.4[P02_0.2-0.4[P04_0.3-0.4[P04_1.3-1.4]] ES34736-2 ES34736-2 ES34736-2 ES34736-2 ES34741-2 ES34741-2 ES34741-2 ES34741-2 ÅÅÅÅ ALOR. ₹OR ALOR ALOR ALOR Å E <LOR **≜LOR** ALOR . الم 0.7 4LOR 0.8 2.8 2.8 2.8 Å R 6,2 3.7 2 ო 2 4 OR A A ROA A OR ▲ OR <1.0R ^LOR <LOR ^ LOR AOR AOR AOR AOR AOR AOR A A A A ROA A ROA A ROA A CR <LOR <LOR</td> 4LOR A AOR A ROA ROA A A A ROA ROA 4.0R <10R ⊲LOR 4.0R <LOR <LOR <LOR <LOR A Ro Ro Ro Lo R R R R R R R R R R R R R R R R R 1.2 × LOR 0.5 0.7 Batch Numbe Sample ID 40 HIF E Investigation Levels 80 NEPM HIL 'D' 23 Health NEPM HIL'A' Benzo(a) pyrene Indeno(1.2.3.cd) pyrene Dibenz(a.h) anthracene Benzo(b)fluoranthene Benzo(k)fluoranthene Pyrene Benz(a)anthracene 3enzo(g.h.i)perylene Acenaphthylene Acenaphthene henanthrene uoranthene Japhthalene nthracene luorene Chrysene otal

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) N/A denotes not analysed

Prepared By TAP Checked by Dod 2
TABLE 5 - 2002 Residual PAH Analytical Data (concentrations in mg/kg)

<u>с</u>

	I talen	in the first section		I Samila ID		TP5 13-1517P6 1.3-1.5	DUP10	DUP09	DUP09 [1P12 0.3-0.41P10 0.3-0.41P8 0.3-0.5	TP10_0.3-0.4	TP8_0.3-0.5	DUP12
			FEACIN	Botch Number ES4741-9	EC3/741-7	ES34741-7	FS34741-2	ES34741-2	ES34741-2	ES34761-2	ES34761-2	ES34761-2
				Dallal Nullion	-1-1-1-CO							
	NEPM	NEPM	NEPM									
	HIL 'A'	HIL U		LOR C				00 IV	<1 OR	<1 OR	< OR	<lor td="" <=""></lor>
Naphthalene				c.0	< LUK	104						
Aronanhthulana				0.5	<_ LOR	<lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""><td>SLUK</td><td>Ň</td><td></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><lor< td=""><td>SLUK</td><td>Ň</td><td></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td>SLUK</td><td>Ň</td><td></td></lor<></td></lor<>	<lor< td=""><td>SLUK</td><td>Ň</td><td></td></lor<>	SLUK	Ň	
According to the				0.5	≤LOR	<10R	<lor< td=""><td>∕LOR</td><td><lor< td=""><td><lor< td=""><td><lor< td=""><td><lok< td=""></lok<></td></lor<></td></lor<></td></lor<></td></lor<>	∕LOR	<lor< td=""><td><lor< td=""><td><lor< td=""><td><lok< td=""></lok<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><lok< td=""></lok<></td></lor<></td></lor<>	<lor< td=""><td><lok< td=""></lok<></td></lor<>	<lok< td=""></lok<>
				0.5	<1.0R	40R	<lor< td=""><td><1.0R</td><td><10R</td><td><10R</td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	<1.0R	<10R	<10R	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
Fluolene				0.5	I OR	<10R	<lor< td=""><td>0.7</td><td><10R</td><td><lor ■</lor </td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	0.7	<10R	<lor ■</lor 	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
Prenantmene				0.5	<1 OR	_ 10R	<lor< td=""><td><lor< td=""><td>⊲LOR</td><td><lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td>⊲LOR</td><td><lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<>	⊲LOR	<lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
Anthracene				200		<10R	<10R	1.3	⊲LOR	<lor< td=""><td><lor< td=""><td><10R</td></lor<></td></lor<>	<lor< td=""><td><10R</td></lor<>	<10R
Fluorantnene				200		<10R	<10R	1.6	<1.0R	<lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
Pyrene				2.0		<10R	<10R	0.5	<lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
Benz(a)anuracene				0.0	10R	<10R	<lor< td=""><td>0.8</td><td><lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	0.8	<lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
Curysene				0.5	< OR	<10R	<lor< td=""><td>0.9</td><td><lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	0.9	<lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
Benzo(b)nuoranmene				0.0	<10R	<10R	<lor< td=""><td>0.5</td><td><lor< td=""><td><lor< td=""><td>SLOR</td><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<>	0.5	<lor< td=""><td><lor< td=""><td>SLOR</td><td><lor< td=""></lor<></td></lor<></td></lor<>	<lor< td=""><td>SLOR</td><td><lor< td=""></lor<></td></lor<>	SLOR	<lor< td=""></lor<>
Benzo(k)IIU0Ialiuterie	•			0.5	 A OR 	4.0R	<lor< td=""><td>0.8</td><td>≤LOR</td><td><lor< td=""><td>⊲LOR</td><td>₹LOR</td></lor<></td></lor<>	0.8	≤LOR	<lor< td=""><td>⊲LOR</td><td>₹LOR</td></lor<>	⊲LOR	₹LOR
Henzo(a)pyrene	-				A OR	<lor< td=""><td>⊲LOR</td><td><lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""><td>AOR</td></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	⊲LOR	<lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""><td>AOR</td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><lor< td=""><td>AOR</td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td>AOR</td></lor<></td></lor<>	<lor< td=""><td>AOR</td></lor<>	AOR
				0.5	A OR	<lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""><td>4OR</td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><lor< td=""><td>4OR</td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td>4OR</td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td>4OR</td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	4OR	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
				0.5	<lor< td=""><td><lor< td=""><td><lor< td=""><td>0.0</td><td><lor< td=""><td><lor< td=""><td><lor< td=""><td>LOR</td></lor<></td></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td>0.0</td><td><lor< td=""><td><lor< td=""><td><lor< td=""><td>LOR</td></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td>0.0</td><td><lor< td=""><td><lor< td=""><td><lor< td=""><td>LOR</td></lor<></td></lor<></td></lor<></td></lor<>	0.0	<lor< td=""><td><lor< td=""><td><lor< td=""><td>LOR</td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td>LOR</td></lor<></td></lor<>	<lor< td=""><td>LOR</td></lor<>	LOR
Total	20	80		40	⊲LOR	<lor< td=""><td><lor< td=""><td>7.7</td><td><lor< td=""><td>⊲LOR</td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td>7.7</td><td><lor< td=""><td>⊲LOR</td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<>	7.7	<lor< td=""><td>⊲LOR</td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	⊲LOR	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
10101												

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) N/A denotes not analysed

s.projects\51072\001\Reports\Tables\Residual data-a.xls\Table 5\19-09-02

Prepared By TAP Checked by Dod2

TABLE 6 - Residual Asbestos Analytical Data

÷.,			TP17 03-051	TP20 0.3-0.5	TP22 0.3-0.5		TP23 0.3-0.5 TP21 0.3-0.5	TP21_0.3-0.5
	11-18 0.3-0.3		2027021	EC24736	ES34736	FS34736	ES34736	ES34736
	ES34736	ES34/30	וני			Within or cond	Mixture of sand	Mixture of sand.
Mixture of sand, Mixture Mixture of sand, Mixture sand, Mixture of sand, Mixture sand, Mixture <th< td=""><td>Mixture of sand fragments of plaster and debris</td><td>Mixture of sand, Mixture of sand, Mixture fragments of fragments of stones, plaster and debris plaster and debris of pla</td><td>Mixture of sand, stones, fragments of plaster and bitumin and debris</td><td>Mixture of sand, i stones, fragments of plaster and debris</td><td>Mixture of sand, sofnes and debris</td><td>of sand Mixture of stones and debris fragments of stones and debris stones, fragments of stones and debris ster and plaster and debris plaster and debris debris and debris a</td><td>stones, fragments of plaster and debris</td><td>stones and debris</td></th<>	Mixture of sand fragments of plaster and debris	Mixture of sand, Mixture of sand, Mixture fragments of fragments of stones, plaster and debris plaster and debris of pla	Mixture of sand, stones, fragments of plaster and bitumin and debris	Mixture of sand, i stones, fragments of plaster and debris	Mixture of sand, sofnes and debris	of sand Mixture of stones and debris fragments of stones and debris stones, fragments of stones and debris ster and plaster and debris plaster and debris debris and debris a	stones, fragments of plaster and debris	stones and debris
							-	
Asbestos Result	Not Detected	Not Detected	Not Detected Not Detected Not Detected Not Detected	Not Detected	Not Detected	Not Detected	Not Detected	Not Letected

s:projects\51072\001\Reports\Tables\Residual data-a.xls\Table 6\19-09-02 Page 18 of 20

Prepared By <u>TAP</u> Checked by <u>Dod 2</u> TABLE 6 - Residual Asbestos Analytical Data

÷.,

			TD01 0 3-0 5	TP03 0 3-0 5	<u>Thes As AFI TEMI A3 AFI TEM3 A3 A51 TE09 0.3-0.5 TP09 0.3-0.5 TP13 0.2-0.4</u>	TP09 0.3-0.5	TP13_0.2-0.4
	20702	0.0-0.0 0Z-1					TC24744
	EBA7571	FS34741	ES34741	ES34741	ES34/41	E034/41	F004/41
		- 1		titi titi te	Alistino of condy coil	Mixture of sand	Mixture of sandy soil.
	Soil	Mixture of sand,	Mixture of sandy sou,	อี		dence framonic of	sand, Mixture of sandy sou, Mixture of satisfy hixture of satisfy hixture of satisfy hixture of satisfy hixture of
		stones, fragments	stones. fragments stones, fragments of soil,		stones, plant matter,	stones, namenus un	stones, stones, plant matter, stones, natifietus of stones, ingriteria of
Samule Description		of plaster and	plaster and bitumin	fragments of plaster	and plaster and bitumin fragments of plaster firagments of plaster and plaster and debrs. [plaster and bitumin	plaster and debris	plaster and bitumin and
		- 2	and debris	and debris	bitumin and debris		debris
Asbestos Result	Not Detected Not Detec	Not Detected	Not Detected	Not Detected	Not Detected	Not Detected	Not Detected

s:projects\51072\001\Reports\Tables\Residual data-a.xis\Table 6\19-09-02 Page 19 of 20

Dudz

Prepared By TAP Checked by Da

TABLE 6 - Residual Asbestos Analytical Data

1

Sample Description and debris	ES34741 Mix	ES34741 Aixture of sand, M tones, fragments st	ES34741 ES34741 ES34761 ES34761 store of sand, Mixture of sand, Mixture of sand, Mixture of sand, stores, tragments stores, plant matter, soil, stores, stores and debris	ES34741 Mixture of sandy	41 ES34761 sandy Mixture of sand, stones, stones and debris
4 I ndy soil, Soil nents of bitumin		ture of sand, here, fragments s	Aixture of sandy soil, stones, plant matter,	Mixture of sandy	Mixture of sand, stones and debris
Mixture of sandy soil, Soil stones, fragments of plaster and bitumin and debris	Mix	kture of sand, here, fragments s	Aixture of sandy soil, stones, plant matter, stones, of clother hride	Mixture of sandy	Mixture of sand, stones and debris
Sample Description stones, fragments of plaster and bitumin and debris	sto	nes, fragments s	tones, plant matter,		stones and debris
			memory of plactor brick		
	of	bitumin and f	laginerits or presid in citraling	and fragments of plaster, brick fragments of plaster	
	det	debris a	and bitumin and debris	and debris	
Asbestos Result Not Detected Not Di	Not Detected N	Not Detected	Not Detected	Not Detected	Not Detected Not Detected

s:projects\51072\001\Reports\Tables\Residual data-a.xls\Table 6\19-09-02 Page 20 of 20

off I

Prepared By____ Checked by ____ TABLE 7 - 2002 Stockpile Inorganic Analytical Data (concentrations in mg/kg)

.

. .

÷

16	BA7654-D			Ċ		_	0	5	3	m		117	56							
DUP16	Ĺ	Ì		-		7	0	ç	3			14	5	*	_					
DUP11	10	-1			,	5										į	7			
DUP17	10011001	E004/07-				AOR	m	1	בי	~	+	218	6	2			U.1			
DUP15		ES34/62-0		-		_ ADR	6		77		-	7		511		00	7'N			
SP10	2	ES34762-U		a V V		<10R	ſ		- -	Ţ	-	24		R	-		¥07			
CD11_0		ES34762-0		00 17	57	4 OR	ſ	7	¢		-	ΑF	2	8						
502	20	FS34762-0				a V		*	6		N	007	97	2			5			
222	SPG	EC34762-0		6	N	100		n	ç	1	•		121	44			A DR			
	SP5	CC24760.0	COUNCE		2		57	(m)	40	2	ç		141	67	3					
	SP4	le	-		a C V		ΥNY V	m		4	•	-	37	6	3			2077		
	SP3	0 001	E839/62-0		<i>.</i>	4	₹UK	0		87	ſ	1	8		171	-	0	U.Z		
	CD2		ES34762-0		207	5	4COR	~		8		7	749		282			<lor.< td=""><td></td><td></td></lor.<>		
	š	5	ES34762-0		٢	_	₹LOR	44		229		-	978	1	272			0.2		
		Calific In	Batch	LOR		-	-	-	-	÷				-	-			5		
		eveis		J. TH MEAN		20	4	000		2000	2224	609	000	- 000	14000			8		
		Health Investigation Levels		NEPM HIL 'D'		Ş		200	400	ADDO	2	7400		IZUN	28000	22224		- Ug	3	
		Health		NEDU HIL Y		90 10	6	3	100	1000	32	600		00m	2000		-	4	2	
		X				Arsenic			Chromitem			Mindeal	IIIVEI	ead			-	10000	Mercury	İ

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) N/A denotes not analysed

s:projects\51072\001\Reports\Tables\Stockpiles-a.xls\Table 7\19-09-02

Prepared By <u>Pod z</u>

TArdLE 8 - 2002 Stockpile TPH/BTEX Analyייכal Data (concentrations in mg/kg)

r

· · ·

,

ļ

1

	;			600	603	103	CDR	9dS	SP7	SP11-9	SP10	DUP15	DUP17	DUP08	DUP16
	Criteria	Sample IU	ž	arz	200	5	3			1 002 002	* C211601	PC04760 1	CC2/762_1	ERA7661-1	ERA7654.1
3		Batch	ES34762-1	ES34762-1	ES34762-1	ES34762-1	ES34762-1	ESG4/62-1	E004/07-1	1-20/501	1-20/1001	1-70 14000	1-30 1-00-1		
							_								
	20					40 M	OR</td <td>< 0R</td> <td><10R</td> <td><lor< td=""><td><lor< td=""><td><lor< td=""><td><10R</td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<></td></lor<></td>	< 0R	<10R	<lor< td=""><td><lor< td=""><td><lor< td=""><td><10R</td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><10R</td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><10R</td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	<10R	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
	8	4								100	001	100	a0 12	a U V	A OR
C10 - C14 Fraction		22	A COR	<10R	<10R	<luk< td=""><td>×rox ▼</td><td></td><td>¥07</td><td>57</td><td></td><td></td><td></td><td></td><td></td></luk<>	×rox ▼		¥07	57					
C1E C28 Eraction		100	232	<1.0R	102	<10R	<lor< td=""><td>134</td><td>40R</td><td><10R</td><td>⊲LOR</td><td><_CCR</td><td><pre></pre></td><td>¥07</td><td>501</td></lor<>	134	40R	<10R	⊲LOR	<_CCR	<pre></pre>	¥07	501
			168	4012	<1.0R	<10.8	<10R	<10R	⊲LOR	AoR	ALOR	^LOR	<lor< td=""><td><1.0R</td><td><10R</td></lor<>	<1.0R	<10R
								134	A 08	<10R	<10R	< LOR	<lor< td=""><td><1.0R</td><td>201</td></lor<>	<1.0R	201
C10-C36 Total	1001		400	FUT.	701										
				-											
			007	40 I2	a0 12	40k	<10R	< OR	<1.0R	40R	<lor< th=""><th>LOR</th><th><lor< th=""><th><0.2</th><th>40.2</th></lor<></th></lor<>	LOR	<lor< th=""><th><0.2</th><th>40.2</th></lor<>	<0.2	40.2
Denzene		70				0			a	A 08	<1 OR	< OR	<10R	<u>6.2</u>	07 9
{Totuene		0.2	ALOK	SEUK.	¥07₹	YDT?	207	107		, in the second se				ç	ç
		0.0	OR</th <th>< 0R</th> <th><10R</th> <th><10R</th> <th>4LOR</th> <th><pre>LOR</pre></th> <th><!--</th--><th><1.0R</th><th><lor< th=""><th><lor< th=""><th><10K</th><th><0.2</th><th></th></lor<></th></lor<></th></th>	< 0R	<10R	<10R	4LOR	<pre>LOR</pre>	</th <th><1.0R</th> <th><lor< th=""><th><lor< th=""><th><10K</th><th><0.2</th><th></th></lor<></th></lor<></th>	<1.0R	<lor< th=""><th><lor< th=""><th><10K</th><th><0.2</th><th></th></lor<></th></lor<>	<lor< th=""><th><10K</th><th><0.2</th><th></th></lor<>	<10K	<0.2	
				2012	A DR	<10R	≙OR	<lor< td=""><td>⊲LOR</td><td><lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""><td>0.2</td><td><0.2</td></lor<></td></lor<></td></lor<></td></lor<></td></lor<>	⊲LOR	<lor< td=""><td><lor< td=""><td><lor< td=""><td><lor< td=""><td>0.2</td><td><0.2</td></lor<></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td><lor< td=""><td>0.2</td><td><0.2</td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td>0.2</td><td><0.2</td></lor<></td></lor<>	<lor< td=""><td>0.2</td><td><0.2</td></lor<>	0 .2	<0.2
EINVIDELIZENS		75								100	AO IN	a O №	4 OR	<u>6</u> 2	40 ^{.2}
meta- & para-Xylene		0.2	¥0,₹	UK</td <td>SLUK</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>00</td> <td>ç</td> <td>5</td>	SLUK								00	ç	5
artho-Xviene		0.2	4.0R	4.0R	<lor< td=""><td><lor< td=""><td><lor< td=""><td>4.0R</td><td>4.0R</td><td>₹0K</td><td>PLOK</td><td>¥0¥</td><td>4LUK</td><td>7.77</td><td></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td>4.0R</td><td>4.0R</td><td>₹0K</td><td>PLOK</td><td>¥0¥</td><td>4LUK</td><td>7.77</td><td></td></lor<></td></lor<>	<lor< td=""><td>4.0R</td><td>4.0R</td><td>₹0K</td><td>PLOK</td><td>¥0¥</td><td>4LUK</td><td>7.77</td><td></td></lor<>	4.0R	4.0R	₹0K	PLOK	¥0¥	4LUK	7.77	
DTEV Total									_						

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) NIA denotes not analysed

s:projects\51072\001\Reports\Tables\Stockpiles-a.xfs\Table 8\19-09-02

Prepared By TAP Checked By Dod2

TABLE 9 - 2002 Stockpile PAH Analytical Data (concentrations in mg/kg)

SP11-9	ES34762-2				A.OR	<lor< th=""><th>A OR</th><th></th><th>404</th><th><lor< th=""><th><10R</th><th></th><th></th><th><pre>LOR</pre></th><th><lor< th=""><th></th><th><10K</th><th><lor< th=""><th><!--OR</th--><th>NR</th><th></th><th>ALOR</th><th><lor< td=""></lor<></th><th>21.08</th><th></th><th></th></th></lor<></th></lor<></th></lor<></th></lor<>	A OR		404	<lor< th=""><th><10R</th><th></th><th></th><th><pre>LOR</pre></th><th><lor< th=""><th></th><th><10K</th><th><lor< th=""><th><!--OR</th--><th>NR</th><th></th><th>ALOR</th><th><lor< td=""></lor<></th><th>21.08</th><th></th><th></th></th></lor<></th></lor<></th></lor<>	<10R			<pre>LOR</pre>	<lor< th=""><th></th><th><10K</th><th><lor< th=""><th><!--OR</th--><th>NR</th><th></th><th>ALOR</th><th><lor< td=""></lor<></th><th>21.08</th><th></th><th></th></th></lor<></th></lor<>		<10K	<lor< th=""><th><!--OR</th--><th>NR</th><th></th><th>ALOR</th><th><lor< td=""></lor<></th><th>21.08</th><th></th><th></th></th></lor<>	OR</th <th>NR</th> <th></th> <th>ALOR</th> <th><lor< td=""></lor<></th> <th>21.08</th> <th></th> <th></th>	NR		ALOR	<lor< td=""></lor<>	21.08		
SP7	ES34762-2				<lor< td=""><td><10R</td><td></td><td></td><td><lor< td=""><td><lor ■</lor </td><td>≤ OR</td><td></td><td><1.0K</td><td>LOR</td><td>< OR</td><td></td><td><.0R</td><td><lor< td=""><td><10R</td><td></td><td>107</td><td><lor< td=""><td><10R</td><td></td><td></td><td></td></lor<></td></lor<></td></lor<></td></lor<>	<10R			<lor< td=""><td><lor ■</lor </td><td>≤ OR</td><td></td><td><1.0K</td><td>LOR</td><td>< OR</td><td></td><td><.0R</td><td><lor< td=""><td><10R</td><td></td><td>107</td><td><lor< td=""><td><10R</td><td></td><td></td><td></td></lor<></td></lor<></td></lor<>	<lor ■</lor 	≤ OR		<1.0K	LOR	< OR		<.0R	<lor< td=""><td><10R</td><td></td><td>107</td><td><lor< td=""><td><10R</td><td></td><td></td><td></td></lor<></td></lor<>	<10R		107	<lor< td=""><td><10R</td><td></td><td></td><td></td></lor<>	<10R			
SPG	FS34762-2			< LOR	<lor< td=""><td><10R</td><td></td><td>PLUK</td><td>4.9</td><td>1.6</td><td>24</td><td></td><td>3.2</td><td>1.5</td><td>9</td><td></td><td>1.7</td><td>0.6</td><td> </td><td></td><td></td><td><lor< td=""><td><10R</td><td></td><td>12.0</td><td></td></lor<></td></lor<>	<10R		PLUK	4.9	1.6	24		3.2	1.5	9		1.7	0.6	 			<lor< td=""><td><10R</td><td></td><td>12.0</td><td></td></lor<>	<10R		12.0	
SP5	ES34762-2	4 40 1000		4 OR	<lor< td=""><td>A OR</td><td></td><td><lok< td=""><td><lor< td=""><td>< OR</td><td></td><td>2.2</td><td>-</td><td>0.5</td><td>90</td><td>2</td><td>0.8</td><td>I ∩R</td><td>0.6</td><td></td><td>4UK</td><td><10R</td><td>∆ OR</td><td></td><td>4.4</td><td></td></lor<></td></lok<></td></lor<>	A OR		<lok< td=""><td><lor< td=""><td>< OR</td><td></td><td>2.2</td><td>-</td><td>0.5</td><td>90</td><td>2</td><td>0.8</td><td>I ∩R</td><td>0.6</td><td></td><td>4UK</td><td><10R</td><td>∆ OR</td><td></td><td>4.4</td><td></td></lor<></td></lok<>	<lor< td=""><td>< OR</td><td></td><td>2.2</td><td>-</td><td>0.5</td><td>90</td><td>2</td><td>0.8</td><td>I ∩R</td><td>0.6</td><td></td><td>4UK</td><td><10R</td><td>∆ OR</td><td></td><td>4.4</td><td></td></lor<>	< OR		2.2	-	0.5	90	2	0.8	I ∩R	0.6		4UK	<10R	∆ OR		4.4	
SP4	EC34769-9	7-70/1000		<lor< td=""><td>≤LOR</td><td>2012</td><td>107</td><td><lor< td=""><td><lor< td=""><td><1 OR</td><td></td><td>4LUK</td><td><10R</td><td>a ⊂ Iv</td><td></td><td></td><td>I OR</td><td>AO I></td><td></td><td></td><td>401∧</td><td><10R</td><td>A OR</td><td></td><td><lor< td=""><td></td></lor<></td></lor<></td></lor<></td></lor<>	≤LOR	2012	107	<lor< td=""><td><lor< td=""><td><1 OR</td><td></td><td>4LUK</td><td><10R</td><td>a ⊂ Iv</td><td></td><td></td><td>I OR</td><td>AO I></td><td></td><td></td><td>401∧</td><td><10R</td><td>A OR</td><td></td><td><lor< td=""><td></td></lor<></td></lor<></td></lor<>	<lor< td=""><td><1 OR</td><td></td><td>4LUK</td><td><10R</td><td>a ⊂ Iv</td><td></td><td></td><td>I OR</td><td>AO I></td><td></td><td></td><td>401∧</td><td><10R</td><td>A OR</td><td></td><td><lor< td=""><td></td></lor<></td></lor<>	<1 OR		4LUK	<10R	a ⊂ Iv			I OR	AO I>			401∧	<10R	A OR		<lor< td=""><td></td></lor<>	
5P3	C C3720 3	E334102-2		<lor< td=""><td><1 OR</td><td></td><td></td><td><lor< td=""><td>0.5</td><td>40 IV</td><td></td><td>1.2</td><td>14</td><td>90</td><td></td><td>0.0</td><td></td><td>aut</td><td></td><td>0.3</td><td>0.6</td><td><1 OR</td><td>10</td><td>0.0</td><td>7.6</td><td></td></lor<></td></lor<>	<1 OR			<lor< td=""><td>0.5</td><td>40 IV</td><td></td><td>1.2</td><td>14</td><td>90</td><td></td><td>0.0</td><td></td><td>aut</td><td></td><td>0.3</td><td>0.6</td><td><1 OR</td><td>10</td><td>0.0</td><td>7.6</td><td></td></lor<>	0.5	40 IV		1.2	14	90		0.0		aut		0.3	0.6	<1 OR	10	0.0	7.6	
CD3	10011001	ES34/02-2		A DR			4 CK	ACOR	<1 OR		57	₹LOR				LOX V					^LOR	aOP		PLOR	<lor< td=""><td></td></lor<>	
100	011	ES34762-2		aC IV			AUK	<10R			0.0	3.4	9			2	 	0.1	-	2	+ +			1.2	医峰-213美国	
		Batch	LOR	0.5		C'N	0.5	0.5		0.0	c .0	0.5		C.D	c .0	05	2	C'N	c .0	0.5	0.5	2.5	C-N	0.5		
	evels		NEPM HILL'E																	2					40	
	Health Investigation Levels		NEPM HI '																	4					80	
	Health		E H WEDN I.U. HIH WEDN I.V. IIH WEDN																	-	•				20	24
	;				Naphthalene	Acenaphthylene	A second theory	Acenaprillelle	Fluorene	Phenanthrene	Anthracene		Fluoranthene	Pyrene	Renz(a)anthracene		Chrysene	Benzo(b)fluoranthene	Renzo(k)finoranthene			Indeno(1.2.3.cd)pyrene	Dibenz(a.h)anthracene		Benzo(g.n.i)perviene	IOTAI

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) N/A denotes not analysed

s:projects\51072\001\Reports\Tables\Stockpiles-a.xls\Table 9\19-09-02

#Ap

Prepared By TM

TABLE 9 - 2002 Stockpile PAH Analytical Data (concentrations in mg/kg)

1

	d the other	tionlik Investigation avale	attala	Sample ID	SP10	DUP15	DUP17	DUP11	01-10
	I ROUT			Batch	ES34762-2	ES34762-2	ES34762-2	EB47653-2	EB47654-2
	ALCONALUT 'A'	NEDM HIL 'D'	NEPM HILF	1 OR					
	_ L			50	AC IN	∆ OR	<10R	<1.0R	<10R
Naphthalene				220	aC 12	<1 OR	<lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
Acenaphthylene				220	<10R	<10R	<lor< td=""><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
vcenaphinene				0.5	<10R	<lor< td=""><td><1.0R</td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	<1.0R	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
Fluorene				0.5	A OR	<lor< td=""><td><lor< td=""><td>< LOR</td><td>3.4</td></lor<></td></lor<>	<lor< td=""><td>< LOR</td><td>3.4</td></lor<>	< LOR	3.4
henanmene				0.5	<1 OR	<lor< td=""><td><lor< td=""><td><lor< td=""><td>0.7</td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td>0.7</td></lor<></td></lor<>	<lor< td=""><td>0.7</td></lor<>	0.7
Anthracene				0.5	<10R	0.7	<lor< td=""><td><lor< td=""><td>3</td></lor<></td></lor<>	<lor< td=""><td>3</td></lor<>	3
luoranthene				0.5	< OR	0.8	<lor< td=""><td><lor< td=""><td>3.7</td></lor<></td></lor<>	<lor< td=""><td>3.7</td></lor<>	3.7
Pyrene				0.5	<10R	<lor< td=""><td><lor< td=""><td>⊲LOR</td><td>1.5</td></lor<></td></lor<>	<lor< td=""><td>⊲LOR</td><td>1.5</td></lor<>	⊲LOR	1.5
Benz(a)anuracene				0.5	<10R	<lor< td=""><td>⊲LOR</td><td>⊲LOR</td><td>1.6</td></lor<>	⊲LOR	⊲LOR	1.6
Unysene				0.5	<<	0.5	40R	<lor< td=""><td>3</td></lor<>	3
				0.5	< 0R	<lor< td=""><td><1.0R</td><td><lor< td=""><td>3</td></lor<></td></lor<>	<1.0R	<lor< td=""><td>3</td></lor<>	3
Benzo(k)iluoranmene	Ŧ		6	0.5	4OR	<pre></pre>	<lor< td=""><td><10R</td><td>1.8</td></lor<>	<10R	1.8
senzo(a)pyrene	-	+	1	0.5	<10R	<10R	<lor< td=""><td><10R</td><td>1</td></lor<>	<10R	1
Indeno(1.2.3.cd)pyrene				0.5	<10R	<lor< td=""><td><10R</td><td><lor< td=""><td><lor< td=""></lor<></td></lor<></td></lor<>	<10R	<lor< td=""><td><lor< td=""></lor<></td></lor<>	<lor< td=""></lor<>
Jipenz(a.n)aniniacene				0.5	⊲LOR	<lor< td=""><td><lor< td=""><td><lor< td=""><td>1.2</td></lor<></td></lor<></td></lor<>	<lor< td=""><td><lor< td=""><td>1.2</td></lor<></td></lor<>	<lor< td=""><td>1.2</td></lor<>	1.2
Benzol <u>g.n.jp</u> erylerie	06	80	40		<lor< td=""><td>2</td><td><lor< td=""><td><lor< td=""><td>23.9</td></lor<></td></lor<></td></lor<>	2	<lor< td=""><td><lor< td=""><td>23.9</td></lor<></td></lor<>	<lor< td=""><td>23.9</td></lor<>	23.9

NOTES (1) <LOR denotes less than the laboratory limit of reporting (2) N/A denotes not analysed

s:projects\51072\001\Reports\Tables\Stockpiles-a.xls\Table 9\19-09-02

Prepared By <u>TAP</u> Checked By <u>Dodz</u>

TABLE 10 - 2002 Stockpile Asbestos Analytical Data

-

Ì

	N1044	501	ςD2	EdS	SP4	SP5	SP6	SP7	SP11-9
			1 2	2					
		EC31767	ES34762	FS34762	ES34762	ES34762	ES34762	E534/62	E034/02
		7012001	1222						
A choose		Chrysofila	CN	Q	QN	QN	UN	N	ND
PSDESICS									

4

NOTES (1) NDdenotes not detected

s:projects\51072\001\Reports\Tables\Stockpiles-a.xls\Table 10\19-09-02

Prepared By ±AP Checked By <u>Dod2</u>

TABLE 10 - 2002 Stockpile Asbestos Analytical Data

									(
3 .	SP10	DUP15	DUP17	DUP16	FD01	SP1A	SP1B	SP1C	SP1D
	0022	COLLOOL	LC21767	CDATERA	EC3/078	EC3/078	ES34078	FS34978	FS34978
	20/4003	E004102	E004/02	100/100	L-004310		201001		1
Achactoc	ND		CN	QN	QN	Q	Q	QN	Q
CONCOUCT OF	2								

NOTES (1) NDdenc

s:projects\51072\001\Reports\Tables\Stockpiles-a.xls\Table 10\19-09-02

Prepared By <u>IAP</u> Checked By <u>Dodt</u>

TABLE 11 - Groundwater Monitoring Summary

<u>ب</u>ر ۱

	Max	MW1	MW2	MW3	MW4	Dup19
TPH						
C6 - C9 Fraction	132	43	<20	39	20	132
C10 - C14 Fraction	26800	17000	<50	5830	4250	26800
C15 - C28 Fraction	84800	53200	<100	21200	20700	84800
C29 - C36 Fraction	3450	2090	. <50	1590	937	3450
C10-C36 Total	115182	72333	0	28659	25887	115182
BTEX						
Benzene	0	٢	4	4	Ŷ	⊽
Totuene	0	₽	₽	2	8	2
Chlorobenzene	0	4	\$	<2	2	8
Ethylpenzene	0	2	8	5	Q	2
meta- & para-Xvlene	0	₽	8	\$	5	8
ortho-Xvlene		8	8	\$	\$	\$
BTEX Total	0	0	0	0	0	0
Water Parameters						
Date		9-10-02	9-10-02	9-10-02	9-10-02	1
Ha	6.92	6.69	6.06	6.71	6.92	•
Temperature (deg. Cel	17.1	15.5	17.1	14.1	13.6	'
EC (uS/cm)	830	248	240	465	830	1
Redox Er (mV)	201	-53	201	30	-71	1
Eh (Er+230)	431	177	431	260	159	1
TDS (ma/L)	410	118.9	112	222	410	'
Dissolved Oxygen (mg/	5.77	2.13	5.77	4.99	2.54	4
Mater Lougle						
SMI (mRGS)	3.32	3.26	3.32	1.94	2.15	1
SWL (mAHD)	36.56	35.48	35.4	35.79	36.56	1
TOC (mAHD)	38.74	38.74	38.72	37.73	38.71	•

Prepared By RSM Checked By Dod 2

5.0 Inspection Findings

The following section details the site inspection findings of the site.

Table 1: Asbestos Inspection Findings of the Site – Lot 202, Former RandwickBus Depot

Lot 202 of the former Randwick Bus Depot was divided into the eight (8) areas. The site plan contained in Appendix 2 shows the location of these areas.

AREA	DESCRIPTION OF AREA	ASBESTOS INSPECTION FINDINGS
Area 1	SE Section of the site – the eastern section of this area has long sparse to thick grasses. There is a large stockpile located in the SE corner, which comprises mostly large concrete pieces.	Many small AC fragments were noted on the ground surface along the eastern side of this area (i.e. between coordinates A3 and A5) with light building rubble. Refer to
	The ground surface of the site access road to the south is predominately crushed building rubble material. The site access road to the north is covered with a concrete slab surface.	Sample No. S2807-01 in Appendix 1.
	Long, sparse to thick grasses restricted visual access to the area located near the mid western side.	
Area 2	Mid Southern Section of the site – the area is predominately covered with a concrete ground slab. There are several small stockpiles, which appear to comprise mostly sandy soil and have a light grass cover and light vegetation.	No asbestos or AC fragments were noted on the ground surface and the surface of the stockpiles.
	There is a small building located on the mid western side and appears to have been constructed post 1980.	
Area 3	NE Section of the site – there is a building, which occupies a large portion of this area.	No asbestos or AC fragments were noted on the ground surface.
	The ground surface of the area north of the building is predominately bitumen. Grass areas to the east and west of the building restricted visual access.	
	The building in this area appears to have been constructed post 1980.	e
Area 4	Mid Northern Section of the site – there is a building, which occupies a large portion of this area.	noted on the concrete stairs and landing to the three southern
	Long, thick grasses restricted visual access surrounding the building to the north, west and east. The ground surface of the area south of the building is predominately bitumen.	entrances and also immediately south of this building (which has a bitumen ground surface). Refer to Sample No. S2807-04 in Appendix 1.
,		Several small AC fragments were noted on the timber flooring in a localised area (i.e. NE corner) inside

)

HIBBS & Associates PTY. LTD.

	AREA	DESCRIPTION OF AREA	ASBESTOS INSPECTION FINDINGS
-			the building. Refer to Sample No. S2807-05 in Appendix 1. Note: These AC fragments were removed by Hibbs & Associates.
untan (17			2 AC downpipe sections (with a total of 5 metres) were noted on the grassed area immediately NW of the building. Refer to Sample No. S2807-06 In Appendix 1.
	-		Note: The roof and eaves lining on the building in this area has been removed and is suspected to have been AC sheeting.
	Area 5	NW Section of the site – concrete ground surface. A large stockpile located in the eastern section of this area comprises mostly bricks and large concrete pieces. A small stockpile in the far SW corner appears to be mostly sandy soil with a grass cover and light	At least 10 small AC fragments were noted on the surface of the small far SW stockpile (fringe SW corner of small stockpile only). Refer to Sample No. S2807-07 in Appendix 1. 3 small AC fragments were noted on
		vegetation. There is a building located on the eastern side of this area.	the concrete ground immediately west of the building. Refer to Sample No. S2807-07 in Appendix 1.
		Long, thick grasses and low-height vegetation restricted visual access on the northern embankment.	Note: The roof and eaves lining on the building in this area has been removed and is suspected to have been AC sheeting.
	Area 6	Mid Western Section of the site – mostly earth ground surface with very light, sparse grass cover. The northern section is covered with a concrete ground slab. Part of a large stockpile, which is also present in Area 7 comprises mostly large concrete pieces, some large brick sections and sandy soil / earth material.	No asbestos or AC fragments were noted on the ground surface.
	Area 7	SW Section of the site – several large stockpiles mostly comprising large concrete pieces, large brick sections, sandy soil / earth material and other construction materials (eg; rail lines sections, metal pipes).	No asbestos or AC fragments were noted on the ground surface and the surface of the stockpiles.
		Concrete ground surface noted in the SW and SE corners of this area.	· · ·
		Long, thick grasses and low-height vegetation restricted visual access particularly on the top portion of most stockpiles and the eastern embankment.	
ļ	Area 8	Near the centre of the site - large excavated area.	No asbestos or AC fragments were noted on the ground surface.
		Areas of long, thick grasses, reeds and low- height vegetation restricted visual access.	





APPENDIX D QA/QC Information

.

9.0	QUALITY ASSURANCE/QUALITY CONTROL		
		Yes	No ^V (Comment below)
9.1	Is a QA/QC narrative included in the report?		
9.2	Sample Handling		
	e COC forms provided, dated and appropriately signed?	V	V X
	ere the samples in proper custody between the field and the laboratory? documentation of sample preservation provided?		Not swe
	documentation of sample condition on receipt provided? ere the samples properly and adequately preserved? This includes keeping		
the	ere the samples received by the laboratory in good condition?	 	Don't know
7) We	ere the sample holding times met?	V	Don'tknor Don'tknor Don'tknor
	as sample handling adequate? Comment as necessary.	L	Don't Know
	MENTS:	•	
*	= exceptions include COC's missing for And	el 95	-03656 an
	Environet 9506018, 9505196, 9506	04.	
4	No information provided separating conc		
	U J	coocc_	of accorder
	upon receipt.	<u>.</u>	. <u> </u>
· · · · · ·			<u></u>
•		:	
		·	•

2

i.

Site Audit Checklist

QA-025

-far

J080

1995

Data

i.

10.0 LABORATORY

- 1. Is the laboratory identified?
- 2. Was a NATA registered laboratory used?
- 3. Did the laboratory perform the requested tests?
- 4. Are laboratory methods identified?
- 5. Were all the laboratory methods adopted NATA endorsed?
- 6. Are any non-standard methods adequately justified and validated?
- 7. Were the appropriate test procedures followed?
- 8. Are PQLs and MDLs for each analyte/matrix combination given?
- 9. Were the reporting limits satisfactory?
- 10. Was the NATA Seal on the reports?

7W-SI,

erceptions

9505802

This is

=) sloppy rypeting.

- 11. Were the reports signed by an authorised person?
- 12. Were laboratory reports satisfactory? Comment as required.

Unsure about whether VHC's were carried out for

we Andel reports 9505659 and

both had incereetly placed

52, -53, -64, as pyle

the moisture data into the TPHCB-C36 total

report which mismapested the co-C36 totals

into the CG-Cg field when CG-Cg was in fact ND.

Also, AGAL root 08078-133-366 dated 1/6/95 does not specify the units - assumed to be mg/kg.

particularly significant

missing from Andel report 9505849

here

COMMENTS:

NoN (Comment Yes below) 1 × 1 V ズ

6

Page 9

Document QA-O25 Rev 2/08/2002 CMJ

he 9505802

FIELD QA/QC 11.0

1) Number of samples collected

Soil:_	150	(includes	duplicat	$\frac{1}{5}$
Water:	~		- 	
Soil:_	15			
	.7			

2) Number of days of sampling:

3) Number and type of QA/QC samples collected:

, i i i i i i i i i i i i i i i i i i i		SOIL			WATER	
	No.	Frequency	Criterion	No.	Frequency	Criterion
Field Duplicates	? \$	re below .		0		
Trip Blanks	\mathcal{O}			0		
Wash Blanks	28	a helow,		O	•	
Other (Field Blanks, Spiked Trip	\bigcirc			\bigcirc		
Blanks, etc.)		<u> </u>		\mathcal{O}		

Field Duplicates 4)

- a. Were an adequate number of field duplicates collected?
- Were RPDs within control limits? b.
 - Organics
 - Metals/Inorganics

Contro	l Limits	Yes	No (Comment Below)
			Ser below
Min	Max		
		\checkmark	

COMMENTS: Difficult to compile without haven orifinal reports. Intra lab duplicates appear submitted to AGAL ave been TEX - repeatability was good 774 B an URS notes that no field duplicates Howeve ratory as the submitted within the same we Overall, the number of field duplicates years to be low, i.e. less than 1 in 10.

5) Trip Blanks	r	DI-
	Yes	No (Comment Below)
 a. Were an adequate number of trip blanks collected? b. Were the trip blanks free of contaminants? (If no, comment whether the contaminants present are also detected in the samples and whether they are common laboratory chemicals.) 		
COMMENTS:		
Protably not - but not likely to be a	in is	Lue
Probably not - but not likely to be a as heavy compounds are the issue of 	n nite	-, not
		···· · · · · · · · · · · · · · · · · ·
6) Wash Blanks		,
	Yes	No (Comment Below)
 a. Were an adequate number of wash blanks collected? b. Were the wash blanks free of contaminants? (If no, comment whether the contaminants present are also detected in the samples and whether they are common laboratory chemicals.) 7) Overview: Was field QA/QC satisfactory? Comment as necessary. 		
<u>COMMENTS:</u>		
	00. 0	nal
Don't Kink so - (but don't have	<u>oup</u>	
Den 't kink po - (but den 't have report to go for) -	<u>cup</u>	
	oup	
	oup	
· ,	<u>.</u>	
· ,	·	

12.0 LABORATORY INTERNAL QUALITY CONTROL PROCEDURES

1) Type and Number of QA/QC Samples

						Sec.
		SOIL			WATER	
	No.	Frequency	Criterion	No.	Frequency	Criterion
Method Blanks/Reagent Blanks		1				1
Matrix Spikes/Matrix Spike Duplicates					· · · · · · · · · · · · · · · · · · ·	
Standard/Certified Reference Material Analysis		- del	at	tau	hed	
Laboratory Duplicates						
Surrogates						

- 2) Were the method blanks/reagent blanks free of contamination?
- 3) Were the spike recoveries within control limits? Organics
 - Metals/Inorganic
- 4) Were the RPDs of the laboratory duplicates within control limits?
- 5) Were the surrogate recoveries within control limits?
- 6) Were the origin and batch number of certified reference material stated?
- 7) Are all QC results provided?
- 8) Was the overall standard of Laboratory QA/QC adequate? Comment below.

Control 1	Limits	Yes	No
Min	Max		(Comment
<u> </u>		·······	Below)
		\checkmark	
		V	
		. /	
		\checkmark	
			NA

COMMENTS:

Amdel QA/QC good - though sometimes too few duplicates Enviromet QA/QC quality O.K. but no blacks canied out. ø AGAL QA/QC quality O.K. but a bit too few

Document QA-O25 Rev 2/08/2002 CMJ

t. T			15 700		747		、 、			· · · · · · · · · · · · · · · · · · ·	-	1100
)		enter de la composition de la										
Lab	Report		Soil water # blank	Connert	At spilles	Comment	# CRM	Comment	# lab.	Connert #	- surrelater	Comment
Andel		44 6Tex	\$		2	¥ •••••				<u>v</u> Š	0	
Andel	9292802	116/10/1/	S	clear, but	- N	ـــــــــــــــــــــــــــــــــــــ		·····		- tere		
1] -	4505815			194 4 4 14 14 14 4 4 14 14 14 14 4 5 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14	2	لاد ٥	- 0 -	(- 🕞	مبر <i>ب</i> ر ا		
·		7944/17	1		1 2	i _j)	<u> </u>)	
-	9505849	H C C C C C	S	· · · · · · · · · · · · · · · · · · ·		<u> </u>	- 4 (
		10 20 20 20 20 20 20 20 20 20 20 20 20 20	· · · · · · · ·	X	N-0-	×])	
	· · · · · · · · ·	8 794 252			- 7	ن لا 0						
ų	9502353	12 At	S	بلا 	2	د د	Q	<u> </u>	·····	لار	-0	
1 I	9503299	9 10 F					- ()	•	· · · · · · · · · · · · · · · · · · ·	. Q		
~	9503388	1 TONY	<u> </u>	<u>بر</u>	N	à			. ()]	0	
anna Nast	9505656	10 744/ BTEX	S	بد 0	7	2	_0	1		0. K		
7	9205659	77 H/ 876X	S	بد. 0	<u> </u>	لر. 	0	. <u> </u>		لر. 	0	
V	892505.6	6 14H/ 61EX		ند ک	2	<u>لد</u>	0	1		j o	-0	1
4	45052	0	(S	- <u>-</u>	2	در - م-	Q	-1-1	0		Ð	
	· · · · · · · ·	- - 		<u>ار ار ا</u>		°, k	0	<u> </u>	-0		0	

50.007	1 - Zer - Aid	1	6	J- Hol) J		- · · · · · · · · · · · · · · · · · · ·	· · · · ·	· · · · ·		
907	FT H Waday	# sample	Sortuta	# Blank	Commet	A. Contres	Comment.	#CRM	Comment	# lab.	$\mathbf{+}$	Hundrates	Comment.
Emiranet	9505153		s S metals							· · · · · · · · · · · · · · · · · · ·	لد_ 6	0	v. ungala kala sua la Ribado I.
1	9,506018	-b	metals		1	0)	10 ⁷		اد۔ ن	Q	
L L	9505196	-7-	N P	0			}			· · · · · · · · · · · · · · · · · · ·	j.	Q	alace VII and The Manager
5	9506025	- Ft	NH							·	لد_ م	0	
n	1409056	e.	mutals	0	1	0		- 🔘				0	u
REAL	0/6/62		Metalst PHT				0	- 10		- Q ()		0 2	
Fam.	69	- 4	S - Tht	O			0. k		د د				
AGAL	1 e l a 5	N Z Z	S 1 Part				1) ->						بر لات
									Star of Low A. E. Sakaray and Additional University Providential and Additional University Providential Additional University of the Additional University of				

13.0 DATA USABILITY

Are the field and laboratory analytical data provided of adequate quality for the purpose of this audit? Comment below as necessary.

Reasonable - except for Calle of confirmation transit, preservation famps 00lab. CON

14.0 ASSESSMENT CRITERIA

Assessment criteria discussed Assessment criteria appropriate Assessment criteria tabulated in text Assessment criteria provided as appendix Assessment criteria appropriately referenced Limitations of criteria stated and discussed

14.1 Evaluating land-use suitability

Has the consultant followed the decision process for assessing urban redevelopment sites (pp 28-29 of the Auditor guidelines) when assessing the suitability for a particular land use?

14.2 Soil Investigation Levels (SILs)

Has the consultant used SILs? Have SILs been used appropriately and competently by the consultant? If SILs have not been used, has the consultant undertaken a site-specific risk assessment?

14.3 Petroleum hydrocarbons

Has the consultant used the threshold values published in the *Guidelines for Assessing* Service Station Sites 1994?

If the Service Station Guideline criteria have not been used, has the consultant undertaken a site specific risk assessment?





Document QA-O25 Rev 2/08/2002 CMJ

7

9.0 QUALITY ASSURANCE/QUALITY CONTROL

9.1 Is a QA/QC narrative included in the report?

9.2 Sample Handling

- 1) Are COC forms provided, dated and appropriately signed?
- 2) Were the samples in proper custody between the field and the laboratory?
- 3) Is documentation of sample preservation provided?
- 4) Is documentation of sample condition on receipt provided?
- 5) Were the samples properly and adequately preserved? This includes keeping the samples chilled, where applicable.
- 6) Were the samples received by the laboratory in good condition?
- 7) Were the sample holding times met?
- 8) Was sample handling adequate? Comment as necessary.

COMMENTS:

QA-025

	No
Yes	(Comment
	below)

1	
V	
N	
V	
\checkmark	
V,	

JO807 1978 Nata Site Audit Checklist

10.0 LABORATORY

- 1. Is the laboratory identified?
- 2. Was a NATA registered laboratory used?
- 3. Did the laboratory perform the requested tests?
- 4. Are laboratory methods identified?
- 5. Were all the laboratory methods adopted NATA endorsed?
- 6. Are any non-standard methods adequately justified and validated?
- 7. Were the appropriate test procedures followed?
- 8. Are PQLs and MDLs for each analyte/matrix combination given?
- 9. Were the reporting limits satisfactory?
- 10. Was the NATA Seal on the reports?
- 11. Were the reports signed by an authorised person?
- 12. Were laboratory reports satisfactory? Comment as required.

COMMENTS:

Yes	No , (Comment below)
V	
V	
NA	
V	
·/	
1/	
V	
V	
V .	

 	·····	
<u></u>		
<u> </u>		

Document QA-O25 Rev 2/08/2002 CMJ

.

11.0 FIELD QA/QC

2)

1)	Number	of samp	les col	lected
-/				

Number of days of sampling:

Soil: 25	······
Water: 0	
Soil: <u>5</u>	
Water:	

3) Number and type of QA/QC samples collected:

	SOIL			WATER		
	No.	Frequency	Criterion	No.	Frequency	Criterion
Field Duplicates	0?					
Trip Blanks	·				I/A	Γ
Wash Blanks	07				V/	1
Other (Field Blanks, Spiked Trip	A 2				F	
Blanks, etc.)	0:					ŀ

4) Field Duplicates

a. Were an <u>adequate number</u> of field duplicates collected?

- b. Were RPDs within control limits?
 - Organics
 - Metals/Inorganics

COMMENTS:

Contro	l Limits	Yes	No (Comment Below)
Min	Max		N ®

DNo field duplicates appear to have been collected-however this information is based on URS's the Anditas has not writed assessment these data. the pretto painer

QA-025 Site Audit Checklist

Trip Blanks

5)

a,	Were an	adequate	number	of trip	blanks	collected?

No Yes (Comment Belðw) V

b. Were the trip blanks free of contaminants? (If no, comment whether the contaminants present are also detected in the samples and whether they are common laboratory chemicals.)

COMMENTS:

 None collected - heref not particula	rly	
relevant on this sile due absence		
volatiles.	V	

Wash Blanks 6)

a. Were an adequate number of wash blanks co	collected?
--	------------

b. Were the wash blanks free of contaminants? (If no, comment whether the contaminants present are also detected in the samples and whether they are common laboratory chemicals.) Overview: Was field QA/QC satisfactory? Comment as 7)

Yes	No (Comment Below)
· ····································	/

COMMENTS;

necessary.

12.0 LABORATORY INTERNAL QUALITY CONTROL PROCEDURES

1) Type and Number of QA/QC Samples

	SOIL			WATER		
	No.	Frequency	Criterion	No.	Frequency	Criterion
Method Blanks/Reagent Blanks						
Matrix Spikes/Matrix Spike Duplicates					. /A	
Standard/Certified Reference Material (See	attaci	ud)		N	[
Laboratory Duplicates						
Surrogates						

- 2) Were the method blanks/reagent blanks free of contamination?
- 3) Were the spike recoveries within control limits? Organics
 - Metals/Inorganic
- 4) Were the RPDs of the laboratory duplicates within control limits?
- 5) Were the surrogate recoveries within control limits?
- 6) Were the origin and batch number of certified reference material stated?
- 7) Are all QC results provided?
- 8) Was the overall standard of Laboratory QA/QC adequate? Comment below.

Control	Limits	Yes	No
Min	Max		(Comment Below)
		V	
		\checkmark	
		V	
			/
		2	1
		· /	

COMMENTS:

of ALS'S GATOC was good - thou low - particularly = no a bit blank BTEV TPH. PAH 40

×.

Document QA-O25 Rev 2/08/2002 CMJ

T. F. USVC	I Levallo	•	1998 Data		10 7.74	24/6/						, -, ., .	
707	1+ 12/2011	# Jup/c	Sal/ antes #	# Harles	Comment	H.M. Ke	comment	#CRM	Connent	H/ab	Comment #	# sumpetes	Comment
¥ -2	ES 10697	8	5- refats cyanide salphate Brex/191. J.		z (sz	0(لا لا ن-ف				o. K.		} 0.k . 0.k .
- X - S-	111 111 112 112 112 112 112 112 112 112	or le	mark the)	t active) === 00 =		5	· · ·	3	10	2 <u>7</u> 4 2 4 2 4 2 2 4 2 2 4 2 2 4 2 2 4 2 2 4 2 2 4 2 2 4 2 2 2 2 2 2 2 2 2 2 2 2 2	0 . r.
A S S	ES/1304		101 pterols metals cyamick			0		0.00	· · · · · · · · · · · · · · · · · · ·	5- N-A-6			C.L.
S 7 4	ES/1357	5	BTEX/TALA BTEX/TALA PART/PLENDS Marten Cymerick	00	Co.L	00)-0-0		Ι- Φ·Θ	Contraction of the sector	2000-4	C.k.
		2	Julyhett BTEX/TPH DMY/phyds TCLP metal	_ 00	C (1)		i 	9-9-9- <u>9-</u> 9		090	(i) vir vir vir vir vir vir vir vir	- to - to - c	
									· · · · · · · · · · · · · · · · · · ·				
1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 -				and a second									

13.0 DATA USABILITY

Are the field and laboratory analytical data provided of adequate quality for the purpose of this audit? Comment below as necessary.

WOH dirocle pears re L blanks O_ 0 re larol be conclusively establish succes campt ina

14.0 ASSESSMENT CRITERIA

Assessment criteria discussed Assessment criteria appropriate Assessment criteria tabulated in text Assessment criteria provided as appendix Assessment criteria appropriately referenced Limitations of criteria stated and

discussed

14.1 **Evaluating land-use suitability**

Has the consultant followed the decision process for assessing urban redevelopment sites (pp 28-29 of the Auditor guidelines) when assessing the suitability for a particular land use?

Soil Investigation Levels (SILs) 14.2

Has the consultant used SILs? Have SILs been used appropriately and competently by the consultant? If SILs have not been used, has the consultant undertaken a site-specific risk assessment?

14.3 **Petroleum hydrocarbons**

Has the consultant used the threshold values published in the Guidelines for Assessing Service Station Sites 1994?

If the Service Station Guideline criteria have not been used, has the consultant undertaken a site specific risk assessment?

Required?	Present?	Adequate?	Comments
			· · · · · · · · · · · · · · · · · · ·

Yes	No	Comments	
		· ·	
	-	1	

'		
	· •	

1		
	 	······
1		
1		
1		

Document QA-O25 Rev 2/08/2002 CMJ

Site Audit Checklist

9.0 QUALITY ASSURANCE/QUALITY CONTROL

J0807 - 2002 Date.

9.1 Is a QA/QC narrative included in the report?

9.2 Sample Handling

- 1) Are COC forms provided, dated and appropriately signed?
- 2) Were the samples in proper custody between the field and the laboratory?
- 3) Is documentation of sample preservation provided?
- 4) Is documentation of sample condition on receipt provided?
- 5) Were the samples properly and adequately preserved? *This includes keeping the samples chilled, where applicable.*
- 6) Were the samples received by the laboratory in good condition?
- 7) Were the sample holding times met?
- 8) Was sample handling adequate? Comment as necessary.

COMMENTS:

	No
Yes	(Comment
	below)

1	
1	
1	
1	
/	
1	
~	
~	

Document Qa-O25 Rev 5/99:CMJ

x*

Site Audit Checklist

10.0 LABORATORY

- 1. Is the laboratory identified?
- 2. Was a NATA registered laboratory used?
- 3. Did the laboratory perform the requested tests?
- 4. Are laboratory methods identified?
- 5. Were all the laboratory methods adopted NATA endorsed?
- 6. Are any non-standard methods adequately justified and validated?
- 7. Were the appropriate test procedures followed?
- 8. Are PQLs and MDLs for each analyte/matrix combination given?
- 9. Were the reporting limits satisfactory?
- 10. Was the NATA Seal on the reports?
- 11. Were the reports signed by an authorised person?
- 12. Were laboratory reports satisfactory? Comment as required.

COMMENTS:

. _____<u>____</u>

Document Qa-O25 Rev 5/99:CMJ

Site Audit Checklist

11.0 FIELD QA/QC

2)

1)	Number	of	samples	collected
----	--------	----	---------	-----------

Number of days of sampling:

Soil:	SEF ATTACHED
Water:	SUMMARY SHEET.
Soil:	4
Water	1

3) Number and type of QA/QC samples collected:

		SOIL			WATER	
	No.	Frequency	Criterion	No.	Frequency	Criterion
Field Duplicates	Ì	710%	10%	<u>.</u>	7107.	10%
Trip Blanks	0					
Wash Blanks	0					
Other (Field Blanks,						
Spiked Trip Blanks, etc.)						

4) Field Duplicates

• Organics

a. Were an <u>adequate number</u> of field duplicates collected?

b. Were RPDs within control limits?

• Metals/Inorganics

Contro	l Limits	Yes	No (Comment Below)
		\checkmark	
Min	Max		
0	35%	\checkmark	

COMMENTS:

4.5.	RPD'S	geneistly	withi	acceptable	Junets	
		0 ,		,		
<u> </u>						
					, , , maanda waa da	
	·					
<u></u>			<u> </u>	· · · · · · ·		

Document Qa-O25 Rev 5/99:CMJ

1.2

5) Trip Blanks

1.

a.	Were an adequate	e number of	f trip blanks	collected?
----	------------------	-------------	---------------	------------

b. Were the trip blanks free of contaminants? (If no, comment whether the contaminants present are also detected in the samples and whether they are common laboratory chemicals.)



COMMENTS:

6) Wash Blanks

a. Were an adequate number of wash blanks collected?

b. Were the wash blanks free of contaminants? (If no, comment whether the contaminants present are also detected in the samples and whether they are common laboratory chemicals.)

7) Overview: Was field QA/QC satisfactory? Comment as necessary.



COMMENTS:

Document Qa-O25 Rev 5/99:CMJ

12.0 LABORATORY INTERNAL QUALITY CONTROL PROCEDURES

1) Type and Number of QA/QC Samples

		SOIL			2	
For 2002 szysting.	No.	Frequency	Criterion	No.	Frequency	Criterion
Method Blanks/Reagent Blanks	Î	1 batch	Ibatch		1 batch	1 botch
Matrix Spikes/Matrix Spike Duplicates		water Aruhed		D		
Standard/Certified Reference Material Analysis	fall	e stanes	1	2	332	52
Laboratory Duplicates				0		
Surrogates					100%	100%

	Control Limits		Yes	No
	Min	Max		(Comment Below)
2) Were the method blanks/reagent blanks free of contamination?			\checkmark	
3) Were the spike recoveries within control limits?	70%	130%		
Organics Metals/Inorganic	70%	130%	~	
4) Were the RPDs of the laboratory duplicates within control limits?	-30b	+ 307	\checkmark *	
5) Were the surrogate recoveries within control limits?	70%	130%	/	
6) Were the origin and batch number of certified reference material stated?				1
7) Are all QC results provided?			\checkmark	
8) Was the overall standard of Laboratory QA/QC adequate? Comment below.			/	

COMMENTS:

* Except where noted on attached sheet.

Document Qa-O25 Rev 5/99:CMJ

^hr.

Ň,

13.0 DATA USABILITY

Are the field and laboratory analytical data provided of adequate quality for the purpose of this audit? Comment below as necessary.



14.0 ASSESSMENT CRITERIA

Assessment criteria discussed Assessment criteria appropriate Assessment criteria tabulated in text Assessment criteria provided as appendix

Assessment criteria appropriately referenced

Limitations of criteria stated and discussed

14.1 Evaluating land-use suitability

Has the consultant followed the decision process for assessing urban redevelopment sites (pp 28-29 of the Auditor guidelines) when assessing the suitability for a particular land use?

Required?	Present?	Adequate?	Comments
/	1	\checkmark	· · · · · · · · · · · · · · · · · · ·
 ✓ 	. 🗸		
	\mathcal{X}	X	Tablated in summary tables.
X			
\checkmark	~	/	



LOT 202 KING ST RANDWICK 2002 DATA

SOILS

() 1 PAH waysound 10w MS (61%) recovery () RPD of 44 % for DUPID & Isbam for lead.

¢ [1		· ·······			Q RPD .	of 44% for DUP Pb<22	17 4 126 a	Sp for	-S-	
ATCH	Anslyte	No. of	Method	Lebur	ster	LCS	scs	MSD		2	800	101
- ·	¢	Samples	Blank	No.	RPD's		DCS	MSD		3	Sug	Field
ES34682	Metals	4	1 /	0		1 -		1 ~		/		1
week 22	ТРН	4		Q			2 ~	2 ,		/	~	1
6,200 23. V	BTEX	4	1 /	0			2 ~	2 /		/		1
	PAM	4		0			21	2 🗸		/		1
ES34735	Metals	4	<u> </u>	0		15		1 ~		\leq		0
wheel .	ТРИ	5	🗸	0			2 /	2 /	ļ	<u> </u>	/	C
1.07/ 23	BTEX	5	1 1	0			2 -	2 /		<u> </u>	<u> </u>	0
<i>d</i> .	РАН	4	<u> </u>				2 -	2 /		<	~	c
ES 34736	Metals	8	<	0		11		1 1		<u> </u>		2
- Jed	TPN	9	1 /	2			2~	21		/		2
i pleded 127.02 j	BTEX	9	1 /	2	1		2 /	21		1	~	2
pp.	PAM	6	1 🗸	2	1		2%	2 🗸		Z		1
5 34.762	Metali	11	1 ~	2	V.O	1/		2 <		/		2
Sidepule Samples.	TPH	ļ	/	2	1		2,	2 ,		1		2
	BTEX	()	1 1	2	1		2 '	2 1		1		2
1 Julia 25-202/	PAM		99999 9 9997 8 9 9 - 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	2	1		21	2. ~		1		2
534741	Metals	8	I 🗸	0		1-				1		2
. 1	TPN	6	1,	0			2,	2 ,		1	~	2
toted r	BTEX	6	1 1	0			21	21		/	~	2
	PAM	1		Ũ			2.	2 1		/		
b\$34761	Metals	5	1 🗸	0		11		1 /			:	3
weeked or	ТРЧ	3		0			21	21			-	2
25.	BTEX	3	⁄	0	· Workersteit - Handelsen		21	2 /			1	2
	PAH	3		0			2 /	2 - 0	~		~	1
ES 35293	TPИ	2		0			2 /	2 /	1			0
We deal . 02	BTEX	2	/	0	· · · · · · · · · · · · · · · · · · ·	, <u></u>	21	2 /	-			0
<u> </u>								· · · · · · · · · · · · · · · · · · ·				~
	1	5	· <u> </u>				· · · · · · · · · · · · · · · · · · ·	**************************************		·}		
· · · · · · · · · · · · · · · · · · ·	******	· ····································				·····			-			
·		· · · · · · · · · · · · · · · · · · ·	1976ad-1996994-144-0000-1-000-1-0010-000-1-000						<u> </u>			
										<u> </u>		

LOT 202 KING ST RANDWICK

2002 DATA

K histysös requert for TP113 3.0-3.2 & DUP110 No analysis reauth for DUP110 suppliest although results for TP113 2.0-2.2 provided.

	002 DATA	- 4	Avislysös regu No suslysis m far	tiviti for TP113	2-0-2-2	110 supple 2 provide	iel 1th ded .	reigh results			4	philitte
= GROUND	WATER.						(<u>*</u>	+	न्की-।	
BATCH	Analyte	No. of	Method	Dupi	icstel	LCS	scs	MS	-	3		Field
· · · · · · · · · · · · · · · · · · ·		Samples	Blank	No .	RPD'S		DCS	MSD		2	N N	1
ES 35294	TPH	6	1 ~	0)	2 .			1		1
ES35294 w ¹ acol #? p ⁰ / acol?	BTEX	6	1 /	0			2 1					1
SOIL				1							Ŀ	
E 35296	ТРЧ	24	2 ~	4			4-	4 -			/	6
Went 18. 8. 22/	BTEX	24	2 1	4			41	4 /			/	6
E 35,298	ТРИ	2		Ð			2/	2 /				D
E 35298 Wester 27.8.02/	BREX	2 10	1/	О			21	2 1				0
1 35560	TPH wheelest	2 11	1/	2			21	21				
£835561	TPY	8 🐇		0			21	2 🗸		Ŷ		2
										1		[
TOTALS		-				-	-			-		
		1	. A L	252	<i>i</i>	6152	0	7 17%		_		710%
SOILS	Metals	40	<u> </u>	10	i.		282			_		2101
1	TPH	85	least one		2		24	24				18 21-1
· ·	BTEX	66	per	8127	ī	-	20	20				4- 707
i 	PAH	39	batch.	4			12	2				5
								ļ	,		<u> </u>	
V ATER	Т.РИ	6	1	0			2352					
· · · · · · · · · · · · · · · · · · ·	BTEX	6	ļ	0			2 3372					
:												
, _, _, _, _, <u>, , , , , , , , , ,</u>						-						T __
		-	_		-		-				-	
			-						_	_	-	+
												-
3						_		-				-
											-	
<u> </u>				_							-	
······································												
Never 16 - 1											_	
	-											
			,				1 k a		And and a second s		1	/
1 2222								_			\mathbf{T}	
											[ı
		1										